RELIABILITY ANALYSIS AND SERVICE LIFE PREDICTION OF PIPELINES

MOJTABA MAHMOODIAN

A thesis submitted in partial fulfilment of the requirement of the University of Greenwich for the Degree of Doctor of Philosophy

June 2013

DECLARATION

I certify that this work has not been accepted in substance for any degree, and is not concurrently being submitted for any degree other than that of Doctor of Philosophy being studied at the University of Greenwich. I also declare that this work is the result of my own investigations except where otherwise identified by references and that I have not plagiarised the work of others.

Student: Mojtaba Mahmoodian

Supervisor: Professor Amir Alani

To my Mother and my Father

ACKNOWLEDGEMENTS

I would like to express my deepest appreciation to my supervisor who provided me the possibility to complete this research. A special gratitude I give to my supervisor, Professor Amir Alani, whose supervision, guidance and advices were great help for me both in the research and in obtaining professional skills in academia.

I also acknowledge the help from the following academics and industrial experts during the last three years for their technical advice and support for improving the quality of the research:

Dr. Kong Fah Tee, University of Greenwich

- Dr. Paul Davis, Intelligent Water Networks, CSIRO Land and Water
- Mr. David Hanson, Yorkshire Water

Dr. Hafiz Elhaq, CPSA

- Dr. Maarten-Jan Kallen, Consultant Risk Analysis and Safety
- Mr. Ken Kienow, BCF Engineers
- Mr. Karl Kienow, Fellow member ASCE
- Dr. Ouahid Harireche, University of Greenwich

Special thanks to my family for all their support during these years. In addition to the technical results of this research, experiencing this PhD period was a great opportunity for me to begin a professional academic career.

ABSTRACT

Pipelines are extensively used engineering structures for conveying of fluid from one place to another. Most of the time, pipelines are placed underground, surcharged by soil weight and traffic loads. Corrosion of pipe material is the most common form of pipeline deterioration and should be considered in both the strength and serviceability analysis of pipes.

The study in this research focuses on two different types of buried pipes including concrete pipes in sewage systems (concrete sewers) and cast iron water pipes used in water distribution systems. This research firstly investigates how to involve the effect of corrosion as a time dependent process of deterioration in the structural and failure analysis of these two types of pipes. Then two probabilistic time dependent reliability analysis methods including first passage probability theory and the gamma distributed degradation model are developed and applied for service life prediction of the pipes. The obtained results are verified by using Monte Carlo simulation technique. Sensitivity analysis is also performed to identify the most important parameters that affect pipe failure.

For each type of the pipelines both individual failure mode and multi failure mode assessment are considered. The factors that affect and control the process of deterioration and their effects on the remaining service life are studied in a quantitative manner.

The reliability analysis methods which have been developed in this research, contribute as rational tools for decision makers with regard to strengthening and rehabilitation of existing pipelines. The results can be used to obtain a cost-effective strategy for the management of the pipeline system.

The output of this research is a methodology that will help infrastructure managers and design professionals to predict service life of pipeline systems and to optimize materials selection and design parameters for designing pipelines with longer service life.

CONTENTS

D	ECL		ATION	2
A	CKN	NON	/LEDGEMENTS	4
A	BST	'RA(CT	5
C	ONI	ΓΕΝ	TS	6
T.	ABL	LES.		9
F	GU	RES		10
Li	ist of	f Syr	nbols	13
1	IN	NTR	ODUCTION	15
	1.1	Ba	ckground and research significance	15
	1.2	Str	ructure of thesis	20
2	S	COP	E OF THE RESEARCH	22
	2.1	Re	search Aim and Objectives	22
	2.2	Re	search Methodology	23
3	L	ITE	RATURE REVIEW	29
	3.1	De	sign of buried pipes	29
	3.	1.1	Design principles	29
	3.	1.2	Loads on buried pipes	30
	3.	1.3	Stresses in buried pipes	33
	3.2	Co	prosion of pipes	34
	3.	2.1	The corrosion mechanism of concrete sewers	34
	3.	2.2	Corrosion mechanism of cast iron water mains	38
	3.3	Se	rvice life prediction and structural reliability analysis	41
	3.	3.1	Background	41
	3.	3.2	Theory of reliability analysis	43
	3.	3.3	Generalisation of a basic reliability problem	46
	3.	3.4	Reliability of structural systems	48
	3.	3.5	Sensitivity analysis	50
	3.	3.6	Background and methods for reliability analysis of pipes	52
	3.	.3.7 ast ir	Gaps in the current state of the art of reliability analysis of concrete sewers	and 59
	3.4	Su	mmary	60

4 AN	DE NAL	EVE YSI	LOPING METHODS FOR TIME DEPENDENT RELIAB S OF PIPES	(LITY
	4 1	Bac	ckground	64
4	4 2	Sel	ection of the appropriate method	65
2	4.3	Firs	st passage probability method	66
2	4.4	Ga	mma process concept	68
	4.4	.1	Problem formulation	69
	4.4 dat	l.2 ta	Developing gamma distributed degradation model with available corrosion 71	ı depth
	4.4 coi	l.3 rrosi	Developing gamma distributed degradation model in case of unavailabition depth data	lity of 75
4	4.5	Mo	onte Carlo simulation method	76
4	4.6	Sur	mmary	78
5	AF 79	PPL]	ICATION OF THE DEVELOPED METHODS TO CONCRETE SEV	WERS
	5.1	Pre	eface	79
	5.2	Cas	se study	80
	5.3	Rel	liability analysis considering individual failure mode	81
	5.3	5.1	Corrosion model and definition of failure mode (limit state function)	81
	5.3	5.2	Calculation of the probability of failure	83
	5.3	5.3	Verification of the results of the probability of failure	87
	5.3	8.4	Sensitivity analysis	87
	5.4	Rel	liability analysis considering multi failure mode	92
	5.4	1.1	Corrosion model and definition of failure modes (limit state functions)	92
	5.4	ł.2	Calculation of the probability of failure	97
	5.4	.3	Verification of the results of the probability of failure	99
	5.4	4.4	Sensitivity analysis	100
	5.5	Sur	mmary	103
6 PI	AF PES	PPL]	ICATION OF THE DEVELOPED METHODS TO CAST IRON WA	ATER 105
(5.1	Pre	eface	105
(5.2	Rel	liability analysis considering individual failure mode	105
	6.2	2.1	Case Study	105
	6.2	2.2	Corrosion model and definition of failure mode (limit state function)	106

6.2.3	Calculation of the probability of failure	113	
6.2.4	Verification of the results of the probability of failure	117	
6.2.5	Sensitivity analysis	118	
6.3 Rel	iability analysis considering multi failure mode	121	
6.3.1	Case study		
6.3.2	Corrosion model and definition of failure modes (limit state functions).		
6.3.3	Calculation of the probability of failure		
6.3.4	Verification of the results of the probability of pipe system failure		
6.3.5	Sensitivity analysis		
6.4 Su	nmary	132	
7 DISCU	JSSION AND ANALYSIS OF THE RESULTS		
7.1 Co	mparison of the two time developed methods		
7.1.1	First passage probability theory		
7.1.2	Gamma distributed degradation (GDD) model	137	
7.2 Rel	iability analysis and service life prediction of concrete sewers		
7.2.1	Results of individual failure mode analysis		
7.2.2	Results of multi failure mode analysis	139	
7.2.3	Sensitivity analysis and effectiveness of variables	140	
7.3 Rel	iability analysis and service life prediction of cast iron water pipes	142	
7.3.1	Results of individual failure mode analysis	143	
7.3.2	Results of multi failure mode analysis	144	
7.3.3	Sensitivity analysis and effectiveness of variables	147	
8 CONC	LUSION AND RECOMMENDATIONS		
8.1 Co	nclusion		
8.2 Rec	commendations for further research	155	
REFEREN	CES	158	
APPENDI	X 1- Codes and programming		
A1.1Mul	ti Failure Mode Reliability Analysis for Concrete Sewer		
A1.2Mul	A1.2Multi Failure Mode Reliability Analysis for Cast Iron Water Pipe		
APPENDI	X 2-List of Publications	174	
APPENDI	X 3- Published Works	175	

TABLES

Table 1.1 Frequency of pipe breakage for different materials (Breaks/100km/year), M	lisiunas
(2005)	18
Table 3.1 Stresses on buried pipes	34
Table 4.1 Typical values for exponential parameter, b, in different deterioration types.	71
Table 5.1 Statistical data for the basic random variables	82
Table 5.2 Formulisation of different resistance modes of the concrete sewer	94
Table 6.1 Geometry and stresses in selected pipes	106
Table 6.2 Values of basic random variables used in the case study	123
Table 6.3 Models for stresses on buried pipes considered in this study	125
Table 7.1 Service life (in years) for selected pipes	143

FIGURES

Figure 2.1 The process and methodologies that are used in current research to reach the a	aims
Eigure 2.1 (a) Hegger earth programs distribution (b) Olander/Medified Olander Ba	28
Processing Distribution (a) Derig/Manual Uniform Drossung Distribution	21
Fressure Distribution, (c). Paris/Manual Uniform Pressure Distribution	
(1989)	
Figure 3.3 Rate of internal and external corrosion for cast iron pipes, Marshall (2001)	40
Figure 3.4 A section of one of London's Victorian water mains, (a) External corrosion.	, (b) 41
Figure 3.5 Two random variable joint density function fRS(r, s), marginal density funct	ions
fR and fS and failure domain D, (Melchers (1999))	48
Figure 3.6 Limit state surface $G(x)=0$ and its linearised version $G_L(x)=0$ in the space of	f the
basic variables, (Melchers (1999)).	48
Figure 3.7 System definitions	49
Figure 3.8 Basic structural system reliability problem in two dimensions showing fai	ilure
domain D (and D1), (Melchers (1999))	50
Figure 4.1 Schematic time dependent reliability problem, (Melchers (1999))	65
Figure 4.2 Time dependent degradation model in case of two inspections	73
Figure 4.3 Gamma distributed degradation (GDD) Model in case of availability of corro	sion
depth data	75
Figure 4.4 Gamma distributed degradation (GDD) Model in case of unavailability	y of
corrosion depth data	76
Figure 5.1 Probability of failure for different auto-correlation coefficient, ρd , from	first
passage probability method	85
Figure 5.2 Probability of failure from gamma distributed degradation (GDD) model	86
Figure 5.3 Verification of the results from the two methods by Monte Carlo simula	ition
Figure 5.4 Polotive contributions of rendem variables in failure function	0/
Figure 5.4 Relative contributions of random variables in fandre function	000
Figure 5.5 Effect of supplide concentration on service file of the sewer	09
right 5.0 Effect of the fatto of the width of the scheam surface to the perimeter of exposed well (hP) on service life of the server	80
Exposed wall (DT) on service life of the server	00
Figure 5.8 Paliability index vs. coefficient of variation of [DS] for various values of	90 F tha
nipeline elansed time	01
Figure 5.9 Reliability index vs. coefficient of variation of h/P' for various values of	91 ? the
nipeline elansed time	01
Figure 5.10 Reliability index vs. coefficient of variation of alkalinity (4) for various value	91
the pipeline elanced time	02 02
Figure 5.11 System combination of the four limit state functions for multi failure m	node
reliability analysis of the concrete sewer	97
Figure 5.12 Probability of system failure from first passage probability method	98
Figure 5.13 Probability of system failure from GDD model	
\mathcal{O}^{*}	

Figure 5.14 Verification of the results from the two methods by Monte Carlo simulation method
Figure 5.15 Probability of system failure of the concrete sewer for different values of basic
$\Gamma_{111111111111111111111111111111111111$
Figure 5.16 Comparison of range values of basic random variables
Figure 6.1 Corrosion rates for cast iron pipes (reproduced from Marshall (2001))
Figure 6.2 External semi-elliptical surface pit on a pipe
Figure 6.3 Internal semi-elliptical surface pit in a pipe
Figure 6.4 Probability of pipe collapse for different ρKI , for case 1 (external corrosion),
using first passage probability method (D = 305mm, $\lambda_{K_i} = 0.41$)
Figure 6.5 Probability of pipe collapse for different ρKI , for case 2 (internal corrosion),
using first passage probability method (D = 254mm, $\lambda_{K_I} = 0.41$)
Figure 6.6 Probability of pipe collapse for Case 1 (external corrosion) with different
Figure 6.7 Probability of pine collapse for Case 2 (internal corresion) with different
diameters, using GDD Model
Figure 6.8 Verification of the results from GDD model by Monte Carlo simulation method
for Case 1 (external corrosion) with different diameters
Figure 6.9 Verification of the results from GDD model by Monte Carlo simulation method
for Case 2 (internal corrosion) with different diameters
Figure 6.10 Probability of pipe failure for different K_c for case 1 (external corrosion), using
GDD Model (D = 305 mm)
Figure 6.11 Relative contribution to the variance of failure function for corrosion multiplying
constant (K), and corrosion exponential constant (n) , a) Case 1, External Corrosion, b) Case
2, Internal Corrosion
Figure 6.12 Probability of failure due to external corrosion (case 1) Vs coefficient of
variation for various values of pipeline elapsed life ((a) corrosion multiplying constant. K.
and (b) corrosion exponential constant. n)
Figure 6 13 Probability of failure due to internal corrosion (case 2) Vs coefficient of variation
for various values of pipeline elapsed life ((a) corrosion multiplying constant K and (b)
corrosion exponential constant n) 121
Figure 6.14 Stresses and cracks on a nine wall (σh : hoop stress and σa : axial stress) 124
Figure 6.15 Probability of the nine system failure (internal corrosion)
Figure 6.16 Probability of the pipe system failure (internal corrosion).
Figure 6.17 Varification of the results from GDD model by Monte Carlo simulation method
(internal corresion)
Figure 6.18 Varification of the regults from CDD model by Monte Carle simulation method
(automal correction)
(External control of random variables in ring failure at different times for
external correspon
CAUCITICAI COLIOSIOII
Figure 6.20 Consistivity notio of random variables subjected to subserve 1 compasing for 1:00-
Figure 6.20 Sensitivity ratio of random variables subjected to external corrosion for different

Figure 6.21 Probability of the pipe system failure due to external corrosion for different
coefficients of variation of k at different times
Figure 6.22 Probability of the pipe system failure due to external corrosion for different
coefficients of variation of <i>n</i> at different times
Figure 7.1 Probability of system failure for different concrete cover, ao, GDD model 139
Figure 7.2 Probability of pipe failure for different cases, using GDD Model ($D = 305$ mm)
individual failure mode analysis
Figure 7.3 Probability of pipe system failure with different corrosion, multi failure mode
analysis146
Figure 7.4 Probability of the pipe failure with different limit states (external corrosion)146
Figure 8.1Contribution of this research in reliability analysis and service life prediction of
pipelines

List of Symbols

- a: depth of the equivalent rectangular stress block, (mm)
- A: the acid-consuming capability of the wall material
- A_s : Area of tension reinforcement in length b, (mm²/m)
- a_0 : Concrete cover (mm)
- b: unit length of pipe, 1000mm
- B_1 : crack control coefficient for effect of spacing and number of layers of reinforcement
- c: the average rate of corrosion (mm/year)
- C_1 : Crack control coefficient for type of reinforcement
- d: distance from compression face to centroid of tension reinforcement, (mm)
- d_b : diameter of rebar in inner cage, (mm)
- [DS]: Dissolved sulphide concentration (mg/l)
- f_c' : design compressive strength of concrete, (MPa)
- f_y : design yield strength of reinforcement, (MPa)
- F: crack width control factor
- F_c : factor for effect of curvature on diagonal tension (shear) strength in curved components

 F_d : factor for crack depth effect resulting in increase in diagonal tension (shear) strength with decreasing d

- F_N : coefficient for effect of thrust on shear strength
- *h*: overall thickness of member (wall thickness), (mm)
- *i*: coefficient for effect of axial force at service load stress
- k: Acid reaction factor
- *J*: is pH-dependent factor for proportion of H_2S
- w: the width of the stream surface
- P': perimeter of the exposed wall
- M_s : Service load bending moment acting on length b, (Nmm/m)
- M_u : factored moment acting on length b, (Nmm/m)

 N_s : Axial thrust acting on length b, service load condition (+ when compressive, - when tensile), (N/m)

 N_u : Factored axial thrust acting on length b, (+ when compressive, - when tensile), (N/m)

- s: is the slope of the pipeline
- t: elapsed time
- u: is the velocity of the stream (m/sec)
- V_b : basic shear strength of length b at critical section
- Φ : The average flux of H₂S to the wall
- $Ø_f$: strength reduction factor for flexure
- ϕ_{v} : strength reduction factor for shear
- Δ : reduction in wall thickness due to corrosion, (mm)

 Δ_{max} : Maximum permissible reduction in wall thickness (structural resistance or limit), (mm)

- σ_F : hoop stress due to internal fluid pressure
- σ_L : frost pressure
- σ_P : axial stress due to internal fluid pressure
- σ_S : soil pressure
- σ_{T_e} : Thermal stress
- σ_V , Traffic stress

1 INTRODUCTION

Buried pipes are subject to chemical and mechanical loading in their environment of service and these stresses cause failure that is costly to repair. Methods to predict pipe performance are poorly developed and require improvement, through the introduction of time dependent reliability analytical tools. This is the subject of this research.

This chapter presents the background and significance of the subject of this research. The need for improved reliability analysis and service life prediction for concrete sewers and cast iron water pipes is established. A review of pipe failures within water and wastewater systems is given, and the costs of failure are outlined.

The outcome of this research is a model for improved reliability analysis that is tested on real-life data. This model can help asset managers to develop a risk-informed and cost-effective strategy for the management and maintenance of corrosion-affected pipelines. Improved reliability analytical tools can assist design engineers to develop pipeline systems with longer service lives.

1.1 Background and research significance

Pipelines are widely used engineering structures for collecting wastewater or for the distribution of water in urban and/or rural areas. Most of the time, pipelines are placed underground, surcharged by soil weight and traffic loads. Evidently, underground pipelines are required to resist the influence of the external loads (soil and traffic) and internal fluid pressure (ASCE (60) 2007, ACPA 2007, Moser and Folkman (2008)).

In many cases underground pipelines are required to withstand particular environmental hazards. Corrosion of pipe material is the most common form of pipeline deterioration and should be considered in both strength and serviceability analysis of buried pipes (Ahammed

& Melchers (1997), Sharma et al. (2008)). The current study focuses on two categories of buried pipes: concrete sewers and cast iron water mains.

In the UK there is approximately 310,000 km of sewer pipes with an estimated total asset value of £110 billion (OFWAT 2000). The investment for repair and maintenance of this infrastructure is approximately £40 billion for the period of 1990 to 2015 (The Urban Waste Water Treatment Directive, 91/271/EEC, 2012). It has been known that sewer collapses are predominantly caused by the deterioration of the pipes. For cementitious sewers, sulphide corrosion is the primary cause of these collapses (Pomeroy (1976), ASCE (69) 2007).

In Los Angeles USA, approximately 10% of the sewer pipes are subject to significant sulphide corrosion, and the costs for the rehabilitation of these pipelines are roughly estimated at £325 million (Zhang et al. (2008)). As an example of an European country, in Belgium, the cost of sulphide corrosion of sewers is estimated at £4 million per year, representing about 10% of total cost for wastewater collection and treatment systems (Vincke (2002)). These statistics indicate that sewer systems are faced with high emergency repair and renewal costs, and frequent charges arising from increasing rates of deterioration. On the other hand, budget limitations are significantly restricting sewer systems and reducing their capabilities in terms of addressing these needs. Therefore to eliminate the high costs associated with sewer failures, sewer system managers need to generate proactive asset management strategies and prioritise inspection, repair, and renewal needs of sewers pipes by utilising reliability analysis. The failure assessment and reliability analysis of sewers can help asset managers to provide an improved level of service and publicity, gain approval and funding for capital improvement projects, and manage operations and maintenance practices more efficiently (Grigg (2003) and Salman and Salem (2012)).

In water distribution systems, although cast iron pipes are being phased out of the water pipeline network in the UK, a significant portion of current networks are comprised of cast

iron pipes with some of them up to 150 years old. There are approximately 335,000 km of water mains in the UK and more than 60% is estimated to be cast iron pipes (Water UK 2007). In the UK, the failure rate of cast iron pipes can be as high as 3000 failures per year (i.e., 10 bursts/1000 km/year) (UKWIR 2002). Of many mechanisms for pipe failures, corrosion of cast iron has been found to be the most predominant, which is linked to almost all pipe failures (Misiunas (2005)).

Data from other countries in the world also shows that, on average, cast iron has been the dominating material for water distribution pipes before the 1960s. Therefore the average age of cast iron pipes in existing networks has been estimated to be 50 years (Rajani and Kleiner, (2004), Misiunas (2005)). Due to their long term use, the aging and deterioration of pipes are inevitable and indeed many failures have been reported worldwide (Atkinson et al. (2002), Misiunas (2005), Rajani *and* Tesfamariam (2007) and EPA/600 2012). Depending on the country, compared with other types of pipe material, cast iron pipes have the highest frequency of breaks as shown in Table 1.1. It has been established (e.g., Yamini and Lence (2010) and EPA/600 2012) that the corrosion of cast iron is the most common form of deterioration of the pipes and it is a matter of concern for both the safety and serviceability of pipes. It is also well known that the consequence of the failure of water pipes can be socially, economically and environmentally devastating, causing, e.g. enormous disruption of daily life, massive costs for repair, widespread flooding and then pollution, and so on. This warrants a thorough assessment of the likelihood of pipe failures and their remaining safe life which is the topic of the present research.

Source	Cast Iron	Ductile Iron	PVC
NRC (1995)	36	9.5	0.7
Weimer (2001)	27	3	4
Pelletier et al. (2003)	55	20	2

Table 1.1 Frequency of pipe breakage for different materials (Breaks/100km/year), Misiunas (2005)

Large investments are required for building new wastewater collection systems and/or water supply infrastructure. It is unlikely to replace the existing pipe networks completely over a short period of time. Therefore the solution is to maintain and rehabilitate the existing pipes. Accurate prediction of the service life of pipes is essential to optimize strategies for maintenance and rehabilitation in the management of pipe assets. Service life (of building component or material) is the period of time after installation during which all the properties exceed the minimum acceptable values when routinely maintained (ASTM E632-82(1996)).

The basis for making quantitative predictions of the service life of structures is to understand the mechanisms and kinetics of many degradation processes of the material whether it is steel, concrete or other materials. Material corrosion in concrete sewers and/or cast iron water pipes is a matter of concern for both strength and serviceability functions. Loss of wall thickness through general corrosion affects the strength of the pipe. To that effect, incorporating the effect of corrosion into the structural analysis of a pipeline is of paramount importance. There are several parameters which may affect corrosion rate and hence the reliability of pipelines. To consider uncertainties and data scarcity associated with these parameters, various researches on probabilistic assessment of buried pipes have been undertaken (De Belie (2004), Sadiq et al. (2004), Davis et al. (2005), Kleiner et al. (2006), Davis et al. (2008), Salman and Salem (2012)). Since the deterioration of buried pipelines is uncertain over time, it should ideally be represented as a stochastic process. A stochastic process can be defined as a random function of time in which for any given point in time the value of the stochastic process is a random variable depending on some basic random variables. Therefore a robust method for reliability analysis and service life prediction of corrosion affected pipes should be a time dependent probabilistic (i.e., stochastic) method which considers randomness of variables to involve uncertainties in a period of time.

In most of the literature, failure and reliability assessment of pipes has been carried out by considering one failure mode (Davis et al. (2005), Desilva et al. (2006), Moglia et al. (2008), Yamini (2009) and Zhou (2011)). However in reality, even in simple cases composed of just one element, various failure modes such as flexural failure, shear failure, buckling, deflection, etc, may exist. To have a more accurate reliability analysis and failure assessment, multi failure mode of concrete sewers and cast iron pipes are also considered in the current study.

For a comprehensive reliability analysis, evaluation of the contributions of various uncertain parameters associated with pipeline reliability can be carried out by using sensitivity analysis techniques. Sensitivity analysis is conducted as a main part of reliability analysis from which the effect of different variables on service life of pipes can be investigated. Sensitivity analysis is the study of how the variation in the output of a model (numerical or otherwise) can be apportioned, qualitatively or quantitatively, to different sources of variation (Saltelli et al. (2004)). Among the reasons for using sensitivity analysis are:

• To identify the factors that have the most influence on reliability of the pipe

• To identify factors that may need more research to improve confidence in the analysis.

• To identify factors that are insignificant to the reliability analysis and can be eliminated from further analysis.

• To identify which, if any, factors or groups of factors interact with each other.

1.2 Structure of thesis

This thesis is organised in 7 chapters as follows:

Chapter 1 - Introduction: This chapter describes the background and significance of the research and the structure of the thesis.

Chapter 2 – Scope of the research: aims and objectives of the research together with the methodologies which are used to address the research objectives are explained in Chapter 2.

Chapter 3 - Literature review: This chapter describes the relevant existing published research works in the areas of design of buried pipelines, corrosion mechanism, reliability analysis, service life prediction and sensitivity analysis methods for infrastructure management in general and for buried pipes in particular.

Chapter 4 – Developing methods for time dependent reliability analysis of pipes: in this chapter time dependent reliability analysis methods are introduced and developed for pipeline reliability analysis.

Chapter 5 – Application of the developed methods for concrete sewers: The two developed approaches in chapter 4 (i.e., first passage probability theory and gamma distributed degradation model) are applied for reliability analysis of a concrete sewer case study in the UK. The results are verified by using Monte Carlo simulation method.

Chapter 6 – Application of the developed methods for cast iron water pipes: The results of application of first passage probability theory and gamma distributed degradation model for reliability analysis of cast iron pipes in the UK are discussed in this chapter. A comparison between the methods is made and the results are verified by Monte Carlo simulation method.

Chapter 7 – Discussion and analysis of the results: The obtained results from application of the proposed methods in chapters 5 and 6 are discussed and analysed in this chapter. A comparison among the different purposed methods for reliability analysis of pipes in this research is presented and weakness and strengths of each method are emphasised. The

chapter outlines how the results address the set objectives of the research and fills the gaps which had been found in literature review

Chapter 8 – Conclusion and recommendations: Conclusions and guidelines for reliability analysis and service life prediction of buried pipes as have been concluded from this study are presented in this chapter. Recommendations are also outlined to address the further research which is needed to develop the area of reliability analysis and service life prediction of corrosion affected buried pipes.

The significance of this research was described in this chapter. A review of failure of concrete sewers and cast iron water pipes reveals that there is a vital need for improved reliability analysis and service life prediction of buried infrastructure, to allow infrastructure managers to improve the management of these assets.

In the next chapter, the aims and objectives of the research are defined. The methodologies used to address the objectives are discussed

2 SCOPE OF THE RESEARCH

In the previous chapter the importance of reliability analysis of concrete sewers and cast iron pipes were described. To address the gap in knowledge outlined in chapter 1, the aims and objectives of the research are defined in this chapter. The methods proposed to meet the objectives of this research are presented.

2.1 Research Aim and Objectives

The significance and necessity of reliability analysis of concrete sewers and cast iron water pipes was discussed in the previous chapter. Apart from some sporadic research on the subject, there is a lack of a reliable methodology and a comprehensive research in the area of reliability analysis of corrosion affected concrete sewers and cast iron pipes. Therefore the aims of the current research were set as follows:

- To develop reliability methods for assessment of buried pipes (i.e., concrete sewers and cast iron water pipes)
- To apply the developed methods to predict service life of concrete sewers and cast iron water pipes in the UK

In order to achieve the aims of the research, the following research objectives were set for this study:

- 1) To understand and investigate the design procedure of buried pipes and their behaviour under various loading conditions.
- To adopt models of structural deterioration (i.e., corrosion) for concrete sewer pipes and cast iron water mains.
- To examine and understand reliability theory and methods in application to pipes.
- To develop rational methods for reliability analysis and service life prediction of corrosion affected buried pipes.
- To test the developed methods/models to concrete sewer pipes and cast iron water pipes

2.2 Research Methodology

The methodologies which are used to address each of the objectives of the research are explained below:

In addressing the appointed objective number 1, a comprehensive literature survey is carried out on the subject to acquire solid knowledge of the design procedure of buried pipes and their behaviour under various loading conditions. Structural reliability analysis and failure assessment of buried pipes can not be achieved without an extensive knowledge about loading and stresses conditions, design principles and failure modes of buried pipes. Current state-of-the-art of research on the design principles for buried pipes and their behaviour under various loads is reviewed. Recently published design manuals and codes of practice are used for this purpose (e.g., ASCE 15-98 (2000), ACPA (2007) and Moser and Folkman (2007)).

ASCE 15-98 (2000) presents the standard practice for direct design of buried precast concrete pipes. It is an appropriate design manual for the design of concrete sewers. Limit state functions (failure functions) which need to be considered in a comprehensive reliability analysis of concrete sewers can be extracted from this design manual. Other references such as ACPA (2007) and Moser and Folkman (2007), give more technical details which support the standardised procedures and formulations in the design manual.

Similarly, for cast iron pipes design handbooks, reports and frequently cited technical papers are used for understanding the design principles, stresses and failure modes which need to be considered for assessment and analysis of cast iron pipes (e.g., Ahammed and Melchers (1997), Rajani et al. (2000) and Moser and Folkman (2007)).

To adopt models of structural deterioration for concrete sewers and cast iron water mains (objective number 2), it is necessary to investigate how buried pipes including concrete sewers and cast iron water pipes deteriorate and how to incorporate the effect of corrosion as a time dependent process of deterioration in the structural analysis of the pipeline. Therefore,

a comprehensive literature review is carried out to understand the chemical and mechanical corrosion mechanisms in concrete sewers and in cast iron water pipes.

Corrosion of buried pipelines is uncertain over time; therefore it should ideally be represented as a stochastic process. In this study, corrosion models taken from key reference works are used in the limit state functions (failure functions) developed in this work.

The general form of a sulphide corrosion model for concrete sewers has not changed since the mid 70's after Pomeroy (1976)'s work. The final form of the formulation for sulphide corrosion rate (ASCE No.60, 2007) is selected in this research and the corrosion depth is adopted from that formulae. By considering corrosion as a stochastic process, the variables in the formulations would be random variables and the corrosion model will have a form of stochastic model.

Likewise, corrosion models for cast iron water pipes are studied inclusively and the most acceptable form of the models is selected for further analysis. Recently published literature such as Rajani and Kleiner (2001), Melchers (2005 a, b) and Melchers (2008) are used to elaborate how a proper model for cast iron corrosion can be adopted.

A comprehensive study is carried out to understand the principles of structural reliability analysis (objective number 3). The focus of the literature review is on reliability analysis of corrosion affected structures in general and buried pipes in particular.

An in depth mathematical study and practice on probability theory and numerical methods should be carried out as a preface for reliability theory. References such as Papoulis and Pillai (2002) and Rubinstein and Kroese (2008) are used for this purpose. Reference books (such as Melchers (1999) and Ditlevsen and Madsen (1996)) and frequently cited papers are also used to understand the principles of reliability theory and the application history of the reliability analysis of pipelines.

A good understanding of probability theory especially in the area of statistical characteristics

of random variables, probability density functions and stochastic processes is achieved by an extensive literature review. This approach facilitates the chance of quantifying the random variables associated with corrosion models leading to improved reliability analysis.

Since the Monte Carlo simulation method is used to verify the results obtained from the analytical reliability analysis methods, a full understanding of this method is required. References such as Melchers (1999) and Rubinstein and Kroese (2008) are used for learning and practicing Monte Carlo simulation technique and frequently cited literature (such as Sadiq et al. (2004) and Zhou (2011)) are used to investigate the adoptability of using this simulation method for pipeline reliability analysis.

To include corrosion mechanism as a time dependent process in reliability analysis of buried pipes, the focus should be on time dependent techniques to calculate the reliability and remaining service life of the pipes. After a comprehensive literature review the most adoptable time dependent methods are developed for reliability analysis of concrete sewers and cast iron pipes (objective number 4) by using advanced analytical mathematics. Probability theory is employed to develop analytical models for deterioration and reliability analysis of pipeline systems.

In addressing the appointed objective number 5, case studies on concrete sewers and cast iron water pipes in the UK are selected for application of the developed reliability analysis and service life prediction methods. A set of CCTV data on concrete sewers from city of Harrogate in the UK is used for the case study of reliability analysis of concrete sewers. Likewise, for cast iron pipes, a set of corrosion measurement data in the UK is taken from Marshall (2001)'s report.

The results from each analytical method and for each case (concrete sewer and/or cast iron pipe) are discussed and the methods are compared. Verification of the results is also carried out by using Monte Carlo simulation method.

MATLAB software is used as a strong programming tool for coding and calculations both for analytical methods and the numerical method (i.e., Monte Carlo).

Sensitivity analysis also is performed to identify the most important parameters that affect pipeline reliability and failure. Sensitivity indexes presented by frequently cited literature are used for this purpose.

To summarise, the application of these methods, the following subjects are investigated:

- The factors that affect and control the process of corrosion in concrete sewers and in cast iron water pipes
- Modelling corrosion process stochastically, to involve uncertainties of random variables which affect the corrosion rate
- Comparison between the developed time dependent reliability analysis methods
- Sensitivity analysis to assess the effectiveness of different parameters on reliability of concrete sewers and cast iron water pipes

Figure 2.1 briefly illustrates the process, steps and methodologies that are used in this research to reach the set aims and objectives.

The developed methodologies in this research can be used as rational tools for decision makers with regard to strengthening and rehabilitation of existing pipelines. Accurate prediction of service life of pipeline system can help structural engineers and asset managers to obtain a cost-effective strategy in the management of the system.

The output of this research will be a methodology that will permit infrastructure managers and construction professionals to:

- Predict service life of buried pipeline systems by a rational and reliable time dependent analysis
- Prioritise of design parameters and random variables by sensitivity analysis techniques.

The aims and objectives for this research were defined in this chapter and the methods outlined. However, to investigate the state-of-the-art of the reliability analysis of buried pipes, it is necessary to understand the engineering design of buried pipes, the corrosion mechanisms responsible for failure and principles of reliability analysis. This is dealt with in the next chapter, where a comprehensive literature review is presented.

Reliability analysis and service life prediction of pipelines



Figure 2.1 The process and methodologies that are used in current research to reach the aims and objectives

3 LITERATURE REVIEW

This chapter investigates the deficiencies in the field of reliability analysis and service life prediction of concrete sewers and cast iron pipes. A comprehensive literature survey is undertaken to gain solid knowledge of the design process, corrosion mechanism involved and methods of reliability analysis. The relevant literature on service life prediction and methods for sensitivity analysis for buried pipes is reviewed in this chapter, to enable the basis for a novel approach for reliability prediction to be established.

3.1 Design of buried pipes

3.1.1 Design principles

The design of buried pipes constitutes a wide ranging and complex field of engineering, which has been the subject of extensive study and research in the world over a period of many years.

There are two main stages for designing of water and wastewater pipes: a) Hydraulic design, and b) Structural design. In the hydraulic design stage, the focus is on determination of the demand of the system for collecting and conveying of water or wastewater. Based on this the diameter of the pipe is estimated. In the second stage, focus is on determination of structural capacity or strength, including details like wall thickness and/or reinforcement. This section discusses the structural design of buried rigid pipes. It introduces and compares different existing design methods. The structural properties of the pipe are analysed to ensure the pipe can safely sustain external and internal loads during its service life time, without loss of its function and without detriment to the environment.

A set of performance criteria must be met when the pipe is subjected to loads. As for other structures, there are two categories of performance criteria for underground rigid pipes: ultimate limit state and serviceability limit state.

The ultimate limit state is represented by the strength of the pipe and is reached when the pipe collapses or fails in general. Flexural and shear failures are two main ultimate limit states that are considered in design and assessment (ASCE 15-98, 2000). Serviceability limit states may be measured by cracking or other functional requirements (for example leakage, deformation beyond allowable limits (for flexible pipes) and excessive movement at the joints).

The principle for the design of a pipe is to ensure that both serviceability and ultimate limit states are not reached. This includes consideration of one or more of the following conditions: strain, stress, bending moment and normal force or load bearing capacity, in the ring or longitudinal direction as appropriate; and water tightness.

The design of a buried pipe involves the selection of an appropriate pipe strength and a bedding combination which is able to sustain the most adverse permanent and transient loads to which the pipeline will be subjected over its design life.

3.1.2 Loads on buried pipes

All pipes shall be designed to withstand the various external and internal loadings to which they are expected to be subjected, during construction and operation. The external loadings include loads due to the backfill, most severe surface surcharge or traffic loading (live load) likely to occur, and self-weight of the pipe and water weight. The internal pressure in the pipeline, if different from atmospheric, shall also be treated as a loading.

Earth load

Beginning in 1910, Anson Marston developed a method for calculating earth loads above a buried pipe based on the understanding of soil mechanics at that time. Marston's formula is considered for calculation of earth load on buried pipes in all codes of practice and manuals (such as BS EN 1295-1 and ASCE No.60). The general form of Marston's equation is:

$$W = C\gamma B^2 \tag{3.1}$$

Where W is the vertical load per unit length acting on the pipe because of gravity soil loads, γ is the unit weight of soil; B is the trench width or pipe width, depending on installation condition; and C is a dimensionless load coefficient depending on soil and installation type (available in design manuals).

The pressure distribution around the pipe from the applied loads (W) and bedding reaction shall be determined from a soil-structure analysis or a rational approximation. Acceptable pressure distribution diagrams from soil-structure analysis are the Heger Pressure Distribution (Figure 3.1a) for use with the Standard Installations; the Olander/Modified Olander Radial Pressure Distribution (Figure 3.1b) or the Paris/Manual Uniform Pressure Distribution (Figure 3.1c).



Figure 3.1 (a). Heger earth pressure distribution, (b). Olander/Modified Olander Radial Pressure Distribution, (c). Paris/Manual Uniform Pressure Distribution (ACPA, 1993)

Pipe and flow dead loads

The dead load of the pipe weight shall be considered in the design based on material density. The dead load of fluid in the pipe also shall be based on the unit weight of the stream (ASCE 15-98, 2000).

Live load

In designing buried pipes, it is necessary to consider the impact of live loads (surcharge) as well as the dead loads. Live loads become a greater consideration when a pipe is installed with shallow cover under un-surfaced road way, railroads and/or airport runways and taxiways. Surcharge loads are calculated using Boussinesq's theory (Moser and Folkman 2008), for various vehicle wheel loading patterns, representing the most severe loadings which might apply in various locations.

Both concentrated and distributed superimposed live loads should be considered in the structural design of sewers. The following equation for determining loads due to superimposed concentrated load, such as a truck wheel load has been presented by ASCE No.60 (2007):

$$W_{sc} = \frac{C_s PF}{L} \tag{3.2}$$

Where

 W_{sc} = the live load on the sewer in kg/m of length

P= the concentrated load (kg)

F= the impact factor

 C_s = the load coefficient, a function of $\frac{B_c}{2H}$ and $\frac{L}{2H}$ where H is the height of fill from the top of pipe to ground surface in (m) and B_c is the width of the sewer in (m)

L= the effective length of sewer in (m)

For the case of superimposed load distributed over an area of considerable extent, the formula

for load on pipe is (ASCE No.60, 2007):

$$W_{sd} = C_s p F B_c \tag{3.3}$$

Where

 W_{sd} =the load on the pipe, (N/m)

p= the intensity of distributed load, (N/m²)

F= impact factor

 B_c = the width of the sewer pipe, (m)

 C_s = the load coefficient, which is a function of $\frac{D}{2H}$ and $\frac{M}{2H}$, and D and M are width and length, respectively, of the area over which the distributed load acts.

H= height from the top of the sewer to ground surface, (m)

3.1.3 Stresses in buried pipes

Rajani et al. (2000) developed a formulation for total external stresses including all circumferential and axial stresses. σ_{θ} is hoop or circumferential stress, which is equal to $\sigma_F + \sigma_S + \sigma_L + \sigma_V$, where σ_F is hoop stress due to internal fluid pressure, σ_S is soil pressure, σ_L is frost pressure and σ_V is traffic stress.

Similarly axial stress, σ_x , would be equal to $\sigma_{Te} + \sigma_{fr} + (\sigma_S + \sigma_L + \sigma_V) \nu_p$ where σ_{Te} is stress related to temperature difference, σ_{fr} is axial stress due to internal fluid pressure, ν_p is pipe material Poisson's ratio and other parameters have already mentioned. Equations and references used for the above mentioned stresses have been presented in Table 3.1.

Stress Type	Model*	Reference
σ_F , hoop stress due to internal fluid pressure	pD 2d	Rajani et al. (2000)
σ_S , soil pressure	$\frac{3K_m \gamma B_d^2 C_d E_P d D}{E_P d^3 + 3K_d p D^3}$	Ahammed & Melchers (1994)
σ_L , frost pressure	f_{frost} . σ_S	Rajani et al. (2000)
σ_V , Traffic stress	$\frac{3 \text{ K}_{\text{m}} \text{ I}_{\text{c}} \text{ C}_{\text{t}} \text{ F} \text{ E}_{\text{P}} \text{ d} \text{ D}}{\text{A}(\text{E}_{\text{P}} \text{ d}^3 + 3\text{K}_{\text{d}} \text{ p} \text{ D}^3)}$	Ahammed & Melchers (1994)
σ_{T_e} , Thermal stress	$- E_P \alpha_P \Delta T_e$	Rajani et al. (2000)
σ_P , axial stress due to internal fluid pressure	$\frac{p}{2} \Big(\frac{D}{d} - 1 \Big) \nu_p$	Rajani et al. (2000)

Table 3.1 Stresses on buried pipes

*: Notations introduced in list of symbols

3.2 Corrosion of pipes

3.2.1 The corrosion mechanism of concrete sewers

Sewer pipes deteriorate at different rate depending on specific local conditions and are not determined by age alone. There have been numerous cases of severe damage to concrete pipes, where it has been necessary to replace the pipes before the desired service life has been reached. There are many cases in which sewer pipes designed to last 50 to 100 years have failed due to H₂S corrosion in 10 to 20 years. In extreme cases, concrete pipes have collapsed in as few as 3 years (Pomeroy 1976). The most corrosive agent that leads to the rapid deterioration of concrete pipelines in sewers is H₂S. Approximately 40% of the damage in concrete sewers can be attributed to biogenous sulphuric acid attack. Sulphide corrosion, which is often called microbiologically induced corrosion, has two distinct phases as follows:

• The conversion of sulphate in wastewater to sulphide, some of which is released as gaseous hydrogen sulphide

• The conversion of hydrogen sulphide to sulphuric acid, which subsequently attacks susceptible pipeline materials.

The surface pH of new concrete pipe is generally between 11 and 13. Cement contains calcium hydroxide, which neutralizes the acids and inhibits formation of oxidizing bacteria when the concrete is new. However, as the pipe ages, the neutralizing capacity is consumed, the surface pH drops, and the sulphuric acid-producing bacteria become dominant. In active corrosion areas, the surface pH can drop to 1 or even lower and can cause a very strong acid attack. The corrosion rate of the sewer pipe wall is determined by the rate of sulphuric acid generation and the properties of the cementitious materials. As sulphides are formed and sulphuric acid is produced, hydration products in the hardened concrete paste (calcium silicon, calcium carbonate and calcium hydroxide) are converted to calcium sulphate, more commonly known by its mineral name, gypsum (ASCE 69 1989). The chemical reactions involved in sulphide build-up can be explained as follows.

Sulphate, generally, abundant in wastewater, is usually the common sulphur source, though other forms of sulphur, such as organic sulphur from animal wastes, can also be reduced to sulphide. The reduction of sulphate in the presence of waste organic matter in a wastewater collection system can be described as follows:

$$SO_4^{-2} + Organic matter + H_2O \longrightarrow 2HCO_3 - +H_2S$$
 (3.4)
Bacteria

The H₂S gas in the atmosphere can be oxidized on the moist pipe surfaces above the water line by bacteria (Thiobacillus), producing sulphuric acid according to the following reaction (Meyer 1980):

Bacteria

As sulphides are formed and sulphuric acid is produced, hydration products in the hardened

concrete paste (calcium silicate, calcium carbonate and calcium hydroxide) are converted to calcium sulphate. The chemical reactions involved in corrosion of concrete are

$$H_{2}SO_{4} + CaO.SiO_{2}.2H_{2}O \rightarrow CaSO_{4} + Si(OH)_{4} + H_{2}O$$

$$H_{2}SO_{4} + CaCO_{3} \rightarrow CaSO_{4} + H_{2}CO_{3}$$

$$(3.7)$$

$$H_{2}SO_{4} + Ca(OH)_{2} \rightarrow CaSO_{4} + 2H_{2}O$$

$$(3.8)$$

Gypsum does not provide much structural support, especially when wet. It is usually present as a pasty white mass on concrete surfaces above the water line. As the gypsum material is eroded, the concrete loses its binder and begins to spall, exposing new surfaces. This process will continue until the pipeline fails or corrective actions are taken. Sufficient moisture must be present for the sulphuric acid-producing bacteria to survive, however; if it is too dry, the bacteria will become desiccated, and corrosion will be less likely to occur. Figure 3.2 shows the process of sulphide build-up in a sewer system.


Figure 3.2 Process occurring in sewer under sulphide build up conditions, ASCE No.69, (1989)

Concrete corrosion rate

The rate of corrosion of a concrete sewer can be calculated from the rate of production of sulphuric acid on the pipe wall, which is in turn dependent upon the rate that H₂S is released from the surface of the sewage stream. The average flux of H₂S to the exposed pipe wall is equal to the flux from the stream into the air multiplied by the ratio of the surface area of the stream to the area of the exposed pipe wall, which is the same as the ratio of the width of the stream surface (*b*) to the perimeter of the exposed wall (\dot{P}). The average flux of H₂S to the wall is therefore calculated as follows (Pomeroy 1976):

$$\Phi = 0.7(su)^{3/8} j[DS](b/\dot{P})$$
(3.9)

Where s is pipe slope, u is velocity of stream (m/s), j is pH-dependent factor for proportion of

 H_2S , [DS] is dissolved sulphide concentration (mg/lit). A concrete pipe is made of cementbonded material, or acid-susceptible substance, so the acid will penetrate the wall at a rate inversely proportional to the acid-consuming capability (*A*) of the wall material. The acid may partly or entirely react. The proportion of acid that reacts is variable (*k*), ranging from 100% when the acid formation is slow, to perhaps 30% to 40% when it is formed rapidly. Thus, the average rate of corrosion (mm/year) can be calculated as follows

$$c = 11.5k\Phi(1/A) \tag{3.10}$$

Where k is the factor representing the proportion of acid reacting, to be given a value selected by the judgement of the engineer and A is the acid-consuming capability, alkalinity, of the pipe material, expressed as the proportion of equivalent calcium carbonate. A value for granitic aggregate concrete ranges from 0.17 to 0.24 and for calcareous aggregate concrete, A ranges from 0.9 to 1.1 (ASCE No.60, 2007). Substituting Equation (3.9) into Equation (3.10):

$$c = 8.05k \times (su)^{3/8}j.[DS] \times \frac{b}{P'A}$$
 (3.11)

Therefore the reduction in wall thickness in elapsed time t, is:

$$d(t) = c.t = 8.05k. (su)^{3/8}j. [DS] \times \frac{b}{P'A}.t$$
 (3.12)

3.2.2 Corrosion mechanism of cast iron water mains

The predominant deterioration mechanism of iron-based pipes is electro-chemical corrosion with the damage occurring in the form of corrosion pits. The damage to iron is often identified by the presence of graphitisation, a result of iron being leached away by corrosion. Either form of metal loss represents a corrosion pit that grows with time and reduces the thickness and mechanical resistance of the pipe wall. This process eventually leads to the breakage of the pipe.

Corrosion pits have a variety of shapes with characteristic depths, diameters (or widths), and lengths. They can develop randomly along any segment of pipe and tend to grow with time at a rate that depends on environmental conditions in the immediate vicinity of the pipeline (Rajani and Makar (2000)).

The corrosion rate of in-service cast iron pipes is believed to be higher in the beginning and then decreases over time as corrosion appears to be a self-inhibiting process (Shreir et al. (1994)). Furthermore, due to the variation of service environment it is rare that the corrosion occurs uniformly along the pipe but more likely locally in the form of a corrosion pit.

A number of models for corrosion of cast iron pipes have been proposed to estimate the depth of corrosion pit (e.g., Sheikh et al. (1990), Ahammed and Melchers (1997), Kucera and Mattsson (1987), Rajani et al. (2000) and Sadiq et al. (2004)). For example, Sheikh et al. (1990) suggested a linear model for corrosion growth in predicting the strength of cast iron pipes. A decade later, Rajani et al. (2000) proposed a two-phase corrosion model where the first phase is a rapid exponential pit growth and the second is a slow linear growth. There are debates in the research community as to whether the corrosion rate can be assumed linear or otherwise. A widely accepted model of corrosion as measured by the depth of corrosion pit is of a power law which was first postulated for atmospheric corrosion by Kucera and Mattsson (1987) and can be expressed as follows:

$$a = kt^n \tag{3.13}$$

Where t is exposure time and k and n are empirical constants largely determined from experiments and/or field data.

For underground corrosion, the constants are typically functions of localised conditions including soil type, the availability of oxygen and moisture and properties of pipeline material. In many cases it may be possible to use past experience to derive estimates for the two constants in Equation 3.13, but with somewhat more effort than would be necessary to estimate a constant corrosion rate as used conventionally (Ahammed and Melchers (1997)). Rajani et al. (2000) proposed a two-phase corrosion model to accommodate this self-inhibiting process:

$$a = \alpha t + \beta (1 - e^{-\lambda t})$$
(3.14)

Where α , β and λ are constant parameters.

In the first phase of the above equation there is a rapid exponential pit growth and in the second phase there is a slow linear growth. This model was developed based on a data set that lacked sufficient points in the early exposure times. Therefore prediction of pit depth in the first 15-20 years of pipe life should be considered highly uncertain when Equation 3.14 is used.

An example of field data which shows the rate of internal and external corrosion for cast iron pipes has been illustrated in Figure 3.3 (Marshall (2001)). As it can be concluded from this data, external corrosion has higher rate than internal corrosion especially during early stages. In Figure 3.4, a sample of a cast iron pipe taken from London water mains in Victorian time (i.e., 1800-1900) also shows the severity of external corrosion compared with internal corrosion.



Figure 3.3 Rate of internal and external corrosion for cast iron pipes, Marshall (2001)



Figure 3.4 A section of one of London's Victorian water mains, (a) External corrosion, (b) Internal corrosion

3.3 Service life prediction and structural reliability analysis

3.3.1 Background

Reliability analysis and the prediction of the service life of structures is one of the major challenges for infrastructure managers and structural engineers. Historically, reliability theory has most often been introduced in the military, aerospace and electronics fields (Cheung and Kyle (1996)). Over the past number of years, the significance of reliability theory has been increasingly realised in the area of civil engineering. The structural reliability began as a subject for academic research about 50 years ago (Freudenthal (1956)). The topic has grown rapidly during the last three decades and has evolved from being a topic for academic research to a set of well-developed or developing methodologies with a wide range of practical applications.

Structural reliability can be defined as the probability that the structure under consideration has a sufficient performance throughout its service life. Reliability methods are used to estimate the service life of structures.

In addition to the prediction of initial service life, reliability methods are effective tools to evaluate the efficiency of repair and replacement. The impact of any repair and maintenance option upon the future performance of the structure can be evaluated by decision makers using reliability analysis methods.

Furthermore, reliability analysis of a structure or a system can be used at the conceptual design stage to evaluate various design choices and to determine the impact that their implementation could have upon their service lives.

The uncertain nature of the loadings and the performance aspects of structures could have led the planners to probabilistic approaches for service life assessment. In probabilistic methods for dealing with uncertainties, the safety and service/performance requirements are measured by their reliabilities. The reliability of a structure or a component is defined as its probability of survival (Melchers (1999)):

$$P_s = 1 - P_f \tag{3.15}$$

Where

P_s: Probability of survival

P_f : Probability of failure

Failure can be defined in relation to different possible failure modes, commonly referred as limit states. Reliability is considered to be the probability that these limits will not be exceeded and is equal to the probability of survival. Each of the limit state function variables is attributed to a probability density function that presents its statistical properties.

To summarise, structural reliability analysis can be generally used for the following purposes:

- Service life prediction of existing structures, for funding allocation to most critical parts of the structure or infrastructure
- Evaluation of the effect of repair, maintenance and rehabilitation actions on the service life of the structure (ability to examine the consequences of potential action or inaction relative to operational and maintenance procedures).
- To be used at the conceptual design stage to evaluate various design choices and to determine the impact that their implementation would have upon the service lives

To predict the service life of existing structures, information is required on the present

condition of the structure, rates of degradation, past and future loading, and definition of the failure of the structure. Based on remaining life predictions, cost-benefit analysis can also be made on whether or not a structure should be repaired, rehabilitated, or replaced.

3.3.2 Theory of reliability analysis

In the past, the design of structural systems considered all loads and strengths as deterministic values. The strength of an element was determined in such a way that it withstood the load within a certain margin. The ratio between the strength and the load was denoted a safety factor.

This safety factor was considered as a measure of the reliability of the structure. However, uncertainties in the loads, strengths and in the modelling of the systems require that methods based on probabilistic techniques in a number of situations have to be used. A structure is usually required to have a satisfactory performance in the expected service life, i.e. it is required that it does not collapse or becomes unsafe and that it fulfils certain functional requirements.

In order to estimate the reliability by using probabilistic concepts it is necessary to introduce stochastic variables and/or stochastic processes/fields and to introduce failure and non-failure behaviour of the structure under consideration.

Generally the main steps in a reliability analysis for service life prediction are:

- 1. Identify the significant failure modes of the structure.
- 2. Decompose the failure modes in series systems or parallel systems of single components (only needed if the failure modes consist of more than one component).
- 3. Formulate failure functions (limit state functions) corresponding to each component in the failure modes.
- 4. Identify the stochastic variables and the deterministic parameters in the failure functions. Further specify the distribution types and statistical parameters for the stochastic variables and the dependencies between them.

- 5. Estimate the reliability of each failure mode (illustration of how reliability or inversely the probability of failure changes with time).
- 6. Evaluate the reliability result by performing sensitivity analyses.

Typical failure modes to be considered in a structural reliability analysis are yielding, corrosion, buckling (local and global), fatigue and excessive deformations.

The failure modes (limit states) are generally divided in:

<u>Ultimate limit states</u> correspond to the maximum load carrying capacity which can be related to e.g. formation of a mechanism in the structure, excessive plasticity, rupture due to corrosion and instability (buckling).

<u>Conditional limit states</u> correspond to the load-carrying capacity if a local part of the structure has failed. A local failure can be caused by an accidental action or by fire. The conditional limit states can be related to e.g. formation of a mechanism in the structure, exceedance of the material strength or instability (buckling).

<u>Serviceability limit states</u> are related to normal use of the structure, e.g. excessive deflections, local damage and excessive vibrations.

The fundamental quantities that characterise the behaviour of a structure are called the basic variables and can be denoted X = (X1, ..., Xn) where n is the number of basic stochastic variables. Typical examples of basic variables are loads, strengths, dimensions and material properties. The basic variables can be dependent or independent. A stochastic process can be defined as a random function of time in which for any given point in time the value of the stochastic process is a random variable.

The uncertainty modelled by stochastic variables can be divided in the following groups:

<u>Physical uncertainty</u>: or inherent uncertainty is related to the natural randomness of a quantity, for example the uncertainty in the yield stress due to production variability.

<u>Measurement uncertainty</u>: is the uncertainty caused by imperfect measurements of for example a geometrical quantity.

<u>Statistical uncertainty</u>: is due to limited sample sizes of observed quantities.

<u>Model uncertainty</u>: is the uncertainty related to imperfect knowledge or idealizations of the mathematical models used or uncertainty related to the choice of probability distribution types for the stochastic variables.

All the above types of uncertainty can usually be treated by the reliability methods. Another type of uncertainty which is not covered by these methods is gross errors or human errors. These types of errors can be defined as deviation of an event or process from acceptable engineering practice.

Generally, methods to measure the reliability of a structure can be divided in four groups, see Madsen and Krenk (1986):

• Level I methods: The uncertain parameters are modelled by one characteristic value, as for example in codes of practice based on the partial safety factor concept.

• Level II methods: The uncertain parameters are modelled by the mean values and the standard deviations, and by the correlation coefficients between the stochastic variables. The stochastic variables are implicitly assumed to be normally distributed. The reliability index method is an example of a level II method.

• Level III methods: The uncertain quantities are modelled by their joint distribution functions. The probability of failure is estimated as a measure of the reliability.

• Level IV methods: In these methods the consequences (cost) of failure are also taken into account and the risk (consequence multiplied by the probability of failure) is used as a measure of the reliability. In this way different designs can be compared on an economic basis taking into account uncertainty, costs and benefits.

Level I methods can e.g. be calibrated using level II methods, level II methods can be calibrated using level III methods, etc.

Several techniques can be used to estimate the reliability for level II and III methods, including the following methods:

• **Simulation techniques**: Samples of the stochastic variables are generated and the relative number of samples corresponding to failure is used to estimate the probability of failure. The simulation techniques are different in the way the samples are generated. Monte Carlo method is the major simulation method for structural reliability analysis.

• **FORM techniques**: In First Order Reliability Methods the limit state function (failure function) is linearized and the reliability is estimated using level II or III methods.

• **SORM techniques**: In Second Order Reliability Methods a quadratic approximation to the failure function is determined and the probability of failure for the quadratic failure surface is estimated.

• **Time dependent reliability techniques**: when a structure is subjected to a time dependent degradation process, probabilistic time dependent methods can be used. First passage probability theory has been introduced for time dependent reliability analysis (Melchers (1999)). Gamma process concept also has the potential of usage as a model for reliability analysis of structures subject to monotonic degradation processes (van Noortwijk and Pandey (2003)). These methods are discussed and developed in Chapter 4 for reliability analysis of concrete sewers and cast iron pipes.

In level IV methods the consequences of failure can be taken into account. In cost-benefit analyses the expected total cost-benefit for a structure in its expected lifetime is maximized. For a detailed introduction to structural reliability theory references are made to the following textbooks: Melchers (1999), Thoft-Christensen & Baker (1982) and Ditlevsen & Madsen (1996).

3.3.3 Generalisation of a basic reliability problem

In a basic reliability problem only one load effect, S, can be resisted by one resistance, R. The load and the resistance are expressed by a known probability density function, f_S and f_R

respectively.

Considering the definition of safety, the structure will be marked as failed if its resistance, R, is less than the stress resultant, S, action on it. Therefore the probability of failure can be stated as follows:

$$P_f = P[R - S \le 0] = P[G(R, S) \le 0]$$
(3.16)

Where G(R,S) is termed the limit state function, and the probability of failure is identical with the probability of limit state violation. In figure 3.5, the above equation is represented by the hatched failure domain D, so that the failure probability becomes (Melchers (1999)):

$$P_f = P(R - S \le 0) = \int_D \int f_{RS}(r, s) dr ds$$
(3.17)

where $f_{R,S}(r, s)$ is the joint (bivariate) density function.



Figure 3.5 Two random variable joint density function $f_{RS}(r, s)$, marginal density functions f_R and f_S and failure domain D, (Melchers (1999))

With the limit state function expressed as G(X), the generalisation of the equation (3.17) becomes:

$$P_f = P[G(X) \le 0] = \int \dots \int_{G(X) \le 0} f_X(X) d_X$$
(3.18)

Here $f_X(X)$ is the joint probability density function for n-dimensional vector X of basic variables. Figure 3.6 shows generalisation of the reliability problem.



Figure 3.6 Limit state surface G(x)=0 and its linearised version $G_L(x)=0$ in the space of the basic variables, (Melchers (1999))

When both the load effect (S) and the pipe resistance (R) are independent and of normal distribution, the integral in Equation 3.17 can be determined from (Melchers, 1999):

$$P_{f}(t) = \Phi\left[\frac{-(\mu_{R} - \mu_{S})}{(\sigma_{S}^{2} + \sigma_{R}^{2})^{1/2}}\right] = \Phi(-\beta)$$
(3.18)

where Φ is the standard normal distribution function, μ is the mean and σ is the standard deviation of random variables. β is known as safety index or reliability index.

3.3.4 Reliability of structural systems

In some cases of reliability analysis even in a simple structure composed of just one element,

various limit states such as bending action, shear, buckling, axial stress, deflection, etc, may apply. Such a composition is referred to as the 'structural system'.

Individual structural components and subsystems have typical service lifespans that do not necessarily coincide with one another. In the reliability evaluation of structural systems it is described how the individual limit states interact on each other and how the overall systems reliability can be estimated when the individual failure modes are combined in a series or parallel system.

In a series system (also called a weakest link system), attainment of any one element limit state constitutes failure of the structure. All components of a parallel system (also called a redundant system) must fail for a system failure to occur. Combining parallel and series subsystems can make more complex systems (Figure 3.7).

If in a system reliability problem each failure mode is represented by a limit state equation Gi(x)=0 in basic variable space, the direct extension of the fundamental reliability problem (Equation 3.18) is

$$P_f = \int_{D \in X} \dots \int f_X(X) d_X \tag{3.19}$$



Figure 3.7 System definitions

Where X represents the vector of all basic random variables (load, strength of members, member properties, sizes, etc.) and D (and D1) is the domain in X defining failure of the

system. This is defined in terms of the various failure modes as $G_i(X) \le 0$.

In Figure 3.8 expression of Equation 3.20 is defined in a two dimensional X space.

As it was defined previously, a parallel system can fail only when all its contributory components have reached their limit states. This means that, in contrast with the situation for series systems, the behavioural characteristics of the system components are significantly important in defining system failure.



Figure 3.8 Basic structural system reliability problem in two dimensions showing failure domain D (and D1), (Melchers (1999))

3.3.5 Sensitivity analysis

Sensitivity analysis is widely accepted as a necessary part of reliability analysis of structures and infrastructure. The effect of variables on the reliability of a pipeline can be analysed by doing a comprehensive sensitivity analysis. In view of the large number of variables that affect the corrosion process and the limit state function, it is of interest to identify those variables that affect the failure most so that more research can focus on those variables.

Sensitivity analysis should be carried out to provide quantitative information necessary for classifying random variables according to their importance. These measures are essential for reliability-based service life prediction of deteriorating materials and structures.

Sensitivity analysis provides the degree of variation of limit state functions or measures at a

specific point characterized by a realisation of all random variables. Similarly to the conventional sensitivity measure in the reliability approaches, the sensitivity measure, S, can be defined as follows (Kong and Frangopol (2005)):

$$S_{G(X)}(X_i) = \frac{\partial G(X)}{\partial X_i} = \lim_{\epsilon \to 0} \frac{G(X+\epsilon) - G(X)}{\epsilon}$$
(3.20)

where G is a performance function of X; X and ε are vectors; and ε is a small perturbation. An element X_i of X can be any type of variable or parameter. For instance, it can be a mean or a standard deviation of a variable, or a deterministic parameter. For a complex system, the sensitivity measure can be computed by using the numerical differentiation method rather than by an analytical approach (Kong and Frangopol (2005)).

Different sensitivity indexes have been introduced. In the current research, relative contribution and sensitivity ratio will be discussed and will be used.

a) Relative Contribution

A sensitivity index that can be used in a comprehensive reliability analysis is the relative contribution of each variable in limit state function. The relative contribution (α_x^2) of each random variable (x) to the variance of the limit state function is introduced as follows (Ahammed and Melchers (1994)):

$$\alpha_{\rm X}^2 = \frac{\left(\frac{\partial_{\rm G}}{\partial_{\rm X}}\sigma_{\rm X}\right)^2}{\sigma_{\rm G}^2} \tag{3.21}$$

where σ_x is standard deviation of the random variable x and σ_G^2 is the variance of the limit state function. Variables with higher values of α_x^2 contribute more in limit state function than other variables; therefore more focus and study needs to be carried out to determine the accurate values for such variables.

b) Sensitivity Ratio (SR)

A method of sensitivity analysis applied in many different models in science, engineering,

and economics is the Sensitivity Ratio (SR), also known as the elasticity equation. The ratio is equal to the percentage change in output (e.g., probability of failure) divided by the percentage change in input for a specific input variable, as shown in the following equation (EPA 540, 2001):

$$SR = \frac{\left(\frac{Y_2 - Y_1}{Y_1}\right) \times 100\%}{\left(\frac{X_2 - X_1}{X_1}\right) \times 100\%}$$
(3.22)

where, Y_1 = the baseline value of the output variable using baseline values of input variables Y_2 = the value of the output variable after changing the value of one input variable X_1 = the baseline point estimate for an input variable

 X_2 = the value of the input variable after changing X_1

Risk estimates are considered most sensitive to input variables that yield the highest absolute value for SR. The basis for this equation can be understood by examining the fundamental concepts associated with partial derivatives. In fact, SR is equivalent to normalized partial derivative. Variables with higher values of sensitivity ratios are more effective on the limit state function or the probability of failure (EPA 540, 2001).

3.3.6 Background and methods for reliability analysis of pipes

Since large investment is required for building new urban water supply infrastructure it is unlikely to replace the existing pipe networks completely over a short period of time. Therefore the resort has to be maintenance and rehabilitation of existing pipes. To have an optimum strategy for maintenance and rehabilitation plans in the management of pipe asset, accurate prediction of the service life of pipes is essential. But this cannot be achieved without an accurate method for reliability analysis in which the likelihood of pipe failures is determined.

Reliability analysis can cover a wide domain of failure assessment of structures and infrastructure including both service life prediction and failure rate prediction. It should be noted that there is a clear distinction between the two terms:

Failure rate prediction of pipes: When the result of reliability analysis and/or failure assessment is presented as a number of failures within a period of time (e.g., breaks/year), it should be ideally considered as failure rate prediction.

Service life prediction of pipes: When in a study, service life of pipe(s), in terms of time, is investigated; the study should be named as service life prediction.

As a comparison, there is considerably less literature in the field of service life prediction compared to the failure rate prediction of pipes. It needs to be clarified that the focus of this study is on service life prediction of buried pipes. To that effect, the outcome of a comprehensive literature survey on reliability analysis and service life prediction of concrete sewers and iron based water pipes (including cast iron water pipes) is presented in this section.

Kleiner and Rajani (2001) define two classes of methods for service life prediction: deterministic and probabilistic methods. Deterministic methods do not consider variation in any variables that affect pipe behaviour and failure, whilst probabilistic methods consider some or all variables as random variables. The input parameters and output results of pipe deterioration models are heavily dependent on the type of methodology chosen. To find out the gaps and limitations of each model, the models are briefly explained here.

<u>Deterministic models</u> often use laboratory tests and sample specimens to find the necessary information, therefore the relationships between components are certain. Variations and uncertainties in variables are not considered in deterministic methods while probabilistic methods consider some or all variables as random variables.

Kaempfer and Berndt (1999) undertook a laboratory experiment to predict service life of concrete sewers subjected to sulphide corrosion. They used deterministic parameters from an accelerated laboratory test to predict the service life of concrete sewers. Since their study was

53

in laboratory conditions, they suggested that for more accurate result the data from real sewage conditions is necessary.

Rajani et al. (2000) proposed a method to estimate the remaining service life of cast iron water pipes by considering that the corrosion pits reduce the structural capacity of the pipes. The residual capacity of the pipes was calculated by a reiterative model, based on corrosion pit measurement and the anticipated corrosion rate. The method was deterministic in the view that it does not consider the uncertainties involved in all factors contributing to the corrosion and subsequent failures.

A comprehensive deterministic life time assessment has been carried out by Kienow and Kienow (2001). They performed sulphide corrosion modelling as part of screening analyses to support the prioritization of sewer evaluation efforts for a sewer inspection and evaluation in the City of Fresno, California. The project consisted of the inspection and evaluation of approximately 90 kilometre of concrete sewers in sizes ranging from 30cm to 70cm in diameter.

Deb et al. (2002) presented a deterministic model based on analysing the growth of corrosion pits on cast iron (CI) pipes, loss of wall thickness and the strength reduction of the pipe over time. Kim et al. (2007) developed prediction models for CI pipes using the assessment of residual tensile strength based on pit characteristics and fracture toughness. Results illustrated that the proposed models using tensile strength and fracture toughness of CI pipes successfully estimated the residual life of water pipes. Analysis of the results showed that the determination of fracture toughness may be more reliable than considering only the pit depth.

<u>Probabilistic models</u> are often used when historical failure or inspection data is limited or unavailable. These models specifically analyse the effective parameters on pipe performance rather than evaluating the previous pipe failure history. Uncertainties are involved by considering random variables. Usually, these methods are applied to pipes where the process of deterioration and factors for failure are well understood.

Various frameworks have been proposed to model the behaviour of underground pipelines for different types of material, using the reliability-based concept. Ahammed and Melchers (1994, 1995 and 1997) reported a comprehensive and continuous study on the reliability analysis of underground steel pipelines. To consider uncertainty associated with the rate of corrosion and the uncertain location of its occurrence, they used a probabilistic approach (first-order second-moment reliability method, FOSM) for the analysis of pipeline reliability. In 1994, they defined the failure mode as exceeding the sum of total stresses from the maximum allowable stress (yield strength of the pipe). They considered three types of existing stresses caused by internal pressure and external pressure:

- The circumferential stress due to internal fluid pressure
- The bending stress in the circumferential direction produced in the pipe wall by external soil loading
- The circumferential bending stresses produced in the pipe wall due to external traffic loads

Taking into account an empirical time-dependent corrosion model, resulted in a nonlinear limit state function that required an iterative solution technique for the calculation of reliability index and for sensitivity analysis.

In 1995 they modelled the growth of corrosion pits to assess the service life of liquid carrying metallic pipelines. In the pitting (localized corrosion) model, two corrosion related parameters (including pipeline dimension and liquid flow) were treated as probabilistic variables.

The limit state function was defined as the difference between the allowable fluid loss and the estimated fluid loss through the pit holes.

To calculate the probability of failure, they used the level II FOSM reliability method. This method requires an iterative solution, when the random variables are not normally distributed

55

and the limit state function is nonlinear.

They developed their model in further studies (Ahammed and Melchers (1997)) by considering the effect of following longitudinal stresses:

- Longitudinal tensile stresses as a result of Poisson's ratio effect from the outward radial action of the internal fluid pressure
- Longitudinal thermal stresses
- Longitudinal stresses due to bending as a result of unevenness or settlement of the pipeline bedding

Camarinopoulos et al. (1999) used a combination of approximate quadrature analytical and Monte Carlo method to evaluate the multiple integrals in their reliability analysis for cast-iron buried water pipes. They also used the model to assess the sensitivity of structural reliability to the variation of some important parameters such as wall thickness, unsupported length and external corrosion coefficient.

Yves and Patrick (2000) also presented a method to calculate the reliability of the buried water pipes using maintenance records and the Weibull distribution for underlying variables. The method appears to rely entirely on the historical data, which in most cases is unknown.

Benmansour and Mrabet (2002) studied the reliability of buried concrete sewers by using the FOSM method. They studied two typical cases to assess the effect of loads on the circumferential and longitudinal behaviour of pipes. Therefore two limit state functions that they considered in their study were based on longitudinal cracking and circular cracking due to bending moments. They did not consider corrosion as a deterioration process, therefore their methodology was a simple time independent method.

Sadiq et al. (2004) used Monte Carlo simulations to perform the reliability analysis of cast iron water mains, considering axial and hoop stresses as acting loads in a limit state function. The reduction in the factor of safety (FOS) of water mains over time was computed, with a failure defined as a situation in which FOS becomes smaller than 1. The Monte Carlo simulations

yielded an empirical probability density function of time to failure, to which a lognormal distribution was fitted leading to the derivation of a failure hazard function.

Davis et al (2005) used Weibull probability distribution to account for variation in the degradation rate of asbestos cement sewers. A tensile failure model was developed that simulates degradation until failure under in-service loading conditions. Simulated failure times were then fitted to a Weibull probability distribution, which allows the expected time to first failure to be calculated at different locations along the pipeline. They found a reasonable agreement between predicted failure times and recorded failures for a period of 8 years.

Amirat et al (2006) used reliability analysis to assess the effect of both the residual stresses generated during manufacturing process and in-service corrosion of underground steel pipes. During the service life of a pipe, residual stress relaxation occurs due to the loss of pipe thickness as material layers are consumed by corrosion. First they focused on the influence of residual stresses in uncorroded pipelines in order to identify the sensitivity of system parameters. In the second step, a probabilistic-mechanical model was used to couple the residual stress model with the corrosion model, in order to assess the aging effects through the pipe service life. For long term corrosion, the reliability analysis incorporated the residual stress relaxation resulting from wall thickness losses. The probability of failure of the pipeline was then evaluated for different corrosion rates varying from the atmospheric baseline to very active corrosion processes.

DeSilva et al. (2006) presented a condition assessment and probabilistic analysis to estimate failure rates in metallic pipelines. A Level II first-order-second moment (FOSM) analysis was combined with condition assessment data to determine the probability of failure. Davis and Marlow (2008) developed a physical probabilistic failure model for service life prediction of CI pipelines subject to corrosion under internal pressure and external loading. A limitation within their study was that the model only considered internal pressure and in-

plane bending; therefore, the resulting failure mode was only shown with a longitudinal fracture, which is just one type of several failure modes. Other failure modes such as circumferential fractures were not considered.

Moglia et al. (2008) looked at the exploration of a CI pipe failure model utilising fracture mechanics of the pipe failure process. The first model generated was simple, which allows explorations of additional model assumptions. Throughout numerous assumptions, the model improved drastically. An elementary method, FOSM, was initially used but proved to yield inaccurate results. A new approach to the model evaluated the nominal tensile strength of pipes, which could determine the maximum corrosion defect. To account for the uncertainty or randomness within the data, a Weibull distribution was utilized adding stochasticity to the corrosion rate. The proposed model calculated failure rates based on historical data using a random Poisson statistical process. The maximum likelihood estimator used within the Poisson distribution was used to calculate the failure rate of the historical data sets. A case study was employed utilising small diameter reticulation mains. By modeling various assumptions into the simulated model, the predicted and observed failure rates were included within the observed data model.

Yamini (2009) also used different reliability analysis methods (Monte Carlo simulation, First Order Reliability Method (FORM) and Second Order Reliability Method (SORM)) for failure analysis of cast iron water mains. In his study two failure modes were considered individually. A failure mode was defined as the point at which the corrosion depth is more than the maximum acceptable decrease in pipe wall thickness and another failure mode was defined as the time at which total stresses exceed the pipe strength capacity.

Lee et al (2010) also used a first order reliability method (FORM) to evaluate the timedependent reliability index for a fully deteriorated piping component rehabilitated with Fibre

58

Reinforced Plastic (FRP), considering the demand of internal fluid pressure, external soil pressure and traffic loading.

Zhou (2011) developed a methodology to carry out the time-dependent reliability evaluation of a pressurized steel gas pipeline containing an active corrosion defect by taking into account the time-dependency of the internal pressure.

3.3.7 Gaps in the current state of the art of reliability analysis of concrete sewers and cast iron pipes

In the previous section, the former works on reliability analysis and service life prediction of underground pipes was reviewed. The comprehensive literature review revealed the areas lacking in knowledge in the subject. In this section the gaps that have been found specifically in the area of reliability analysis and service life prediction of concrete sewers and cast iron water pipes are declared.

a) Reliability analysis of concrete sewers

Despite some sporadic works on prediction of failure rates in sewer systems (Ana et al. (2008) and Savic et al. (2009)), there has not been sufficient research on service life prediction of concrete sewers. The lack of literature is more remarkable in the field of probabilistic service life prediction of concrete sewers.

The literature survey in this research also showed that although the mechanism of sulphide corrosion in concrete sewers has been reasonably understood, effectiveness and contribution of corrosion parameters on the service life of pipelines have not been studied by researchers. This lack of knowledge necessitates an extensive sensitivity analysis on reliability analysis of concrete sewers.

b) Reliability analysis of cast iron water pipes

Compared to concrete sewers, the study on probabilistic service life prediction of cast iron

water pipes as a metal base infrastructure has been considered by several researchers in recent years (Rajani et al. (2000), Sadiq et al. (2004) and Davis and Marlow (2008)).

In most cases a Monte Carlo simulation method has been used as a strong numerical method for reliability analysis of buried pipes. The mathematical complexity of analytical methods for time dependent reliability analysis has made these methods less effective for failure assessment. This complexity is expanded when multi failure mode assessment (i.e., more than one limit state function) needs to be considered for a more realistic and comprehensive study. Parameters which are involved in failure and service life of cast iron water pipes have been clarified in previous researches (e.g., Rajani et al. (2000) and Sadiq et al. (2004)). However, the significance and contribution level of these parameters on failure function(s) and service life of cast iron pipelines have not been studied previously. This is what is dealt with in sensitivity analysis of reliability of cast iron water pipes in chapter 6.

A review of most recent research literature (Sadiq et al. (2004), Moglia et al. (2008), Yamini (2009) and Clair and Sinha (2012)) also suggests that in most reliability analyses for buried pipes, multi failure modes are rarely considered; while the real condition in practice, necessitates consideration of multi failure modes analysis.

3.4 Summary

At the first step of reliability analysis and service life prediction of buried pipes, it is necessary to gain knowledge about the design principles and loads affecting buried pipes. The performance criteria for in service loads were introduced in this chapter as ultimate limit state and serviceability limit state. Flexural and shear failures are two main ultimate limit states that are considered in design and assessment. Serviceability limit states may be measured by cracking condition. These criteria should be met for a buried pipe to be safe and in service. This chapter reviewed the loads and stresses acting on buried pipes. The formulations for stresses will be used in reliability analysis, where the limit state function (failure mode) is checked.

Material corrosion is the most common form of buried pipe deterioration and is a matter of concern for both the strength and durability functions. Loss of wall thickness through general corrosion affects the strength of a pipe. Evidently, how to incorporate the effect of corrosion in the structural analysis of a pipeline is of practical importance. This chapter outlined the corrosion formulation for concrete sewers and for cast iron water mains.

Parameters affecting sulphide corrosion in a concrete sewer were introduced based on literature review and reliable references. Metal corrosion in cast iron water mains was also discussed and common formulation for corrosion in those pipes were outlined.

The principles of structural reliability analysis were discussed and the previous studies for reliability analysis and service life prediction of buried pipes were presented based on a comprehensive literature review.

Failure rate prediction has been considered in most of the previous researches in the field of reliability analysis and the failure assessment of sewers and water pipes. In contrast, service life prediction of these pipes has been considered less. This lack of literature is more remarkable for concrete sewers compared with cast iron water pipes. To that effect, one of the main contributions of the current research is to fill the gap in the area of service life prediction of both concrete sewers and cast iron pipes.

There are several parameters which may affect corrosion rate and hence the service lives of pipes. In conventional methods for their service life prediction, these parameters are considered to be deterministic. However, in reality there are uncertainties associated to these parameters. An approach to uncertainty representation is to represent each uncertain variable as a random variable with mean and standard deviation. Therefore for reliability analysis of corrosion affected pipes, using probabilistic methods is necessary. While the process of corrosion is time dependent, the reliability analysis methods which are used for corrosion

61

affected pipes should also be time dependent. Therefore, among the reliability analysis techniques which were noted in section 3.3.2, using time dependent reliability techniques can be the best approach for dealing with reliability analysis and service life prediction of corrosion affected buried pipes. The methods are developed in chapter 4 to be applied for the reliability analysis of concrete sewers and cast iron water mains.

Another gap in the current state of the art of reliability analysis of pipes which was found in the literature review, is the lack of research on the multi failure mode analysis. Therefore the emphasis in the current study is to consider multi failure modes in reliability analysis and service life prediction of concrete sewers and cast iron water pipes.

The evaluation of the contributions of various uncertain parameters associated with pipeline life assessment can be carried out by using sensitivity analysis techniques. The effectiveness and contribution of corrosion parameters on the service life of concrete sewers and cast iron water pipes which have not been deeply studied by other researchers, are investigated in the current research.

In this chapter, the design of buried pipes was reviewed. The service loads, corrosion mechanism and principles of service life prediction were presented. A critical analysis of methods for reliability analysis and service life prediction of concrete sewers and cast iron water pipes was presented. In chapter 4, the novel methodologies proposed for development of two time-dependent reliability methods of analysis are presented.

62

4 DEVELOPING METHODS FOR TIME DEPENDENT RELIABILITY ANALYSIS OF PIPES

The deficiencies in the area of the reliability analysis and service life prediction of concrete sewers and cast iron pipes were identified and discussed in chapter 3. In this chapter, two new methods for the time-dependent reliability analysis of corrosion affected pipes are presented. These methods will allow for:

- Application to all type of corrosion-affected structures and/or pipelines,
- The capability to consider multi failure mechanisms/modes
- Consideration of the scarcity of monitoring data from real world examples

The outcome of this approach is potentially a robust tool for use by infrastructure managers

to predict and mitigate pipe failure.

4.1 Background

As it was concluded in the previous chapter, probably the most viable approach to predict the structure's reliability or its service life under future performance conditions is through probability-based techniques involving time dependent reliability analyses.

By using these techniques a quantitative measure of structural reliability is provided to integrate information on design requirements, material and structural degradation, damage accumulation, environmental factors, and non-destructive evaluation technology. The technique can also investigate the role of in-service inspection and maintenance strategies in enhancing reliability and extending service life. Several non-destructive test methods that detect the presence of a defect in a structure tend to be qualitative in nature in that they indicate the presence of a defect but may not provide quantitative data about the defect's size, precise location, and other characteristics that would be needed to determine its impact on structural performance. None of these methods can detect a given defect with certainty. The imperfect nature of these methods can be described in statistical terms. This randomness affects the calculated reliability of a component.

Structural loads, engineering material properties, and strength-degradation mechanisms are random. The resistance, R(t), of a structure and the applied loads, S(t), both are stochastic functions of time. At any time, t, the safety limit state, G(R, S, t), is (Melchers (1999)):

$$G(R, S, t) = R(t) - S(t)$$
 (4.1)

Making the customary assumption that R and S are statistically independent random variables, the probability of failure resulting from Equation 4.1, $P_f(t)$, is (Melchers (1999)):

$$P_f(t) = P[G(t) \le 0] = \int_0^\infty F_R(x) f_s(x) dx$$
(4.2)

in which $F_R(x)$ and $f_S(x)$ are the probability distribution function of R and density function of S respectively. Equation 4.2 provides quantitative measure of structural reliability and

performance, provided that Pf can be estimated and validated.

The probability that failure occurs for any one load application is the probability of limit state violation. Roughly, it may be represented by the amount of overlap of the probability density functions f_R and f_s in Figure 4.1. Since this overlap may vary with time, P_f also may be a function of time.



Figure 4.1 Schematic time dependent reliability problem, (Melchers (1999))

4.2 Selection of the appropriate method

As it was mentioned in section 3.3.6 of Chapter 3, between the two categories of reliability analysis methods, probabilistic methods should be considered for reliability analysis and service life prediction of buried pipes. For corrosion affected pipes, the method should also be time dependent.

It can be concluded from the a comprehensive literature review in chapter 3 that probabilistic time dependent methods need to be developed for use in reliability analysis of corrosion affected buried pipes. The methods include: first passage probability method and gamma process concept method.

In the following sections these methods are developed to be applied for corrosion affected pipes. Monte Carlo simulation method is also discussed to be used as a strong simulation technique for reliability analysis. This technique is used in the next chapters to verify the results obtained from the application of the two probabilistic time dependent methods (i.e., first passage probability and gamma process concept).

4.3 First passage probability method

The service life of a pipe or structure in general is a time period at the end of which the pipe stops performing the functions it is designed and built for. As it is mentioned earlier, to determine the service life for pipes, a limit state function (G(t) = R(t) - S(t)) is introduced. Where S(t) is the action (load) or its effect at time t and R(t) is the acceptable limit (resistance) for the action or its effect. With the limit state function of Equation (4.1), the probability of pipe (structural) failure, p_f , can be determined by:

$$P_f(t) = P[G(t) \le 0] = P[S(t) \ge R(t)]$$
(4.3)

At a time that $P_f(t)$ is greater than a maximum acceptable risk in terms of the probability of pipe failure, p_a , it is the time the pipe becomes unsafe or unserviceable and requires replacement or repairs. This can be determined from the following:

$$P_{\rm f}(T_{\rm L}) \ge P_{\rm a} \tag{4.4}$$

where T_L is the service life for the pipe for the given assessment criterion and acceptable risk. In principle, the acceptable risk, p_a , can be determined from a risk-cost optimization of the pipeline system during its whole service life. This is beyond the scope of this research and will not be discussed herein but can be referred to Mann and Frey (2011) and Dawotola et al. (2012).

Equation 4.3 represents a typical upcrossing problem in mathematics and can be dealt with using time-dependent reliability methods. Time-dependent reliability problems are those in

which either all or some of basic variables are modeled as stochastic processes. In this method, the structural failure depends on the time that is expected to elapse before the first occurrence of the action process S(t) upcrossing an acceptable limit (the threshold) L(t) sometime during the service life of the structure $[0, T_L]$. Equivalently, the probability of the first occurrence of such an excursion is the probability of failure $P_f(t)$ during that time period. This is known as "first passage probability" and can be determined by (Melchers (1999)):

$$P_f(t) = 1 - [1 - P_f(0)]e^{-\int_0^t v d_t}$$
(4.5)

Where $P_f(0)$ is the probability of structural failure at time t = 0 and v is the mean rate for the action process S(t) to upcross the threshold R(t).

The upcrossing rate in Equation 4.5 can be determined from the Rice formula (Melchers (1999)):

$$\nu = \nu_R^+ = \int_R^\infty (\dot{S} - \dot{R}) f_{S\dot{S}}(R, \dot{S}) d_{\dot{S}}$$
(4.6)

where ν_R^+ is the upcrossing rate of the action process S(t) relative to the threshold R, \dot{R} is the slope of R with respect to time, $\dot{S}(t)$ is the time derivative process of S(t) and f_{SS} is the joint probability density function for S and \dot{S} . An analytical solution to Equation 4.6 has been derived for a deterministic threshold R in Li and Melchers (1993) as follows:

$$v_R^+ = \frac{\sigma_{\hat{S}|S}}{\sigma_S} \, \emptyset\left(\frac{R-\mu_S}{\sigma_S}\right) \left\{ \emptyset\left(-\frac{\dot{R}-\mu_{\hat{S}|S}}{\sigma_{\hat{S}|S}}\right) - \frac{\dot{R}-\mu_{\hat{S}|S}}{\sigma_{\hat{S}|S}} \Phi\left(-\frac{\dot{R}-\mu_{\hat{S}|S}}{\sigma_{\hat{S}|S}}\right) \right\}$$
(4.7)

where \emptyset and Φ are standard normal density and distribution functions respectively, μ and σ denote the mean and standard deviation of *S* and \dot{S} , represented by subscripts and " | " denotes the condition. For a given Gaussian stochastic process with mean function $\mu_S(t)$, and auto-covariance function $C_{SS}(t_i, t_j)$, all terms in Equation 4.7 can be determined, based on the theory of stochastic processes as detailed in Papoulis and Pillai (2002) as follows.

$$\mu_{\dot{S}|S} = E[\dot{S}|S=R] = \mu_{\dot{S}} + \rho \frac{\sigma_{\dot{S}}}{\sigma_{S}}(R-\mu_{S})$$
(4.8a)

$$\sigma_{\dot{S}|S} = \left[\sigma_{\dot{S}}^2 (1 - \rho^2)\right]^{1/2} \tag{4.8b}$$

Where

$$\mu_{\dot{S}} = \frac{d\mu_S(t)}{dt} \tag{4.8c}$$

$$\sigma_{\dot{S}} = \left[\frac{\partial^2 C_{SS}(t_i, t_j)}{\partial t_i \partial t_j} \big|_{i=j}\right]^{1/2}$$
(4.8d)

$$\rho = \frac{C_{SS}(t_i, t_j)}{\left[C_{SS}(t_i, t_i).C_{SS}(t_j, t_j)\right]^{1/2}}$$
(4.8e)

And the cross-covariance function is:

$$C_{SS}(t_i, t_j) = \frac{\partial C_{SS}(t_i, t_j)}{\partial t_j}$$
(4.8f)

Because it is unlikely that the corrosion depth in a given pipe exceeds the wall thickness at the beginning of structural service, the probability of failure due to corrosion at t = 0 is zero, i.e., $P_f(0) = 0$. The solution to Equation 4.5 can be expressed, after substituting Equation 4.7 into Equation 4.5, and considering that R is constant ($\dot{R} = 0$) therefore:

$$P_f(t) = \int_0^t \frac{\sigma_{\dot{S}|S}(t)}{\sigma_S(t)} \oint \left(\frac{R - \mu_S(t)}{\sigma_S(t)}\right) \left\{ \oint \left(-\frac{\mu_{\dot{S}|S}(t)}{\sigma_{\dot{S}|S}(t)}\right) + \frac{\mu_{\dot{S}|S}(t)}{\sigma_{\dot{S}|S}(t)} \Phi \left(\frac{\mu_{\dot{S}|S}(t)}{\sigma_{\dot{S}|S}(t)}\right) \right\} d_\tau$$
(4.9)

For Equation (4.9) to be of practical use, i.e., determining the probability of serviceability failure due to corrosion over time, there is a requirement to develop a stochastic model for the corrosion depth. This is dealt with in section 5.3.1 in Chapter 5.

4.4 Gamma process concept

To deal with data scarcity and uncertainties, using stochastic models for time dependent reliability analysis of corrosion affected buried pipes can be considered. In order to model monotonic progression of a deterioration process, the stochastic gamma process concept can be used for modeling the reduction of pipe wall thickness due to corrosion. The gamma process is a stochastic process with independent, non-negative increments having a gamma distribution with an identical scale parameter and a time-dependent shape parameter.

A stochastic process model, such as gamma process, incorporates the temporal uncertainty associated with the evolution of deterioration (e.g. Bogdanoff and Kozin (1985), Nicolai et al. (2004), van Noortwijk and Frangopol (2004)).

The gamma process is suitable to model gradual damage monotonically accumulating over time, such as wear, fatigue, corrosion, crack growth, erosion, consumption, creep, swell, a degrading health index, etc. For the mathematical aspects of gamma processes, see Dufresne et al. (1991), Ferguson and Klass (1972), Singpurwalla (1997), and van der Weide (1997).

To the best of authors' knowledge, Abdel-Hameed (1975) was the first to propose the gamma process as a model for deterioration occurring randomly in time. In his paper he called this stochastic process the "gamma wear process". An advantage of modeling deterioration processes through gamma processes is that the required mathematical calculations are relatively straightforward.

4.4.1 Problem formulation

The mathematical definition of the gamma process is given in Equation (4.10). Given that a random quantity *d* has a gamma distribution with shape parameter $\alpha > 0$ and scale parameter $\lambda > 0$ if its probability density function is given by:

$$Ga(d|\alpha,\lambda) = \frac{\lambda^{\alpha}}{\Gamma(\alpha)} d^{\alpha-1} e^{-\lambda d}$$
(4.10)

Let $\alpha(t)$ be a non-decreasing, right continuous, real-valued function for $t \ge 0$, with $\alpha(0) \equiv 0$. $\Gamma(\alpha)$ denotes gamma function of α with mathematical definition of $\Gamma(\alpha) = (\alpha - 1)!$. The gamma process is a continuous-time stochastic process $\{d(t), t \ge 0\}$ with the following properties:

1. d(0) = 0 with probability one;

2.
$$d(\tau) - d(t) \sim Ga(\alpha(\tau) - \alpha(t), \lambda)$$
 for all $\tau > t \ge 0$;

3. d(t) has independent increments.

Let d(t) denote the deterioration at time $t, t \ge 0$, and let the probability density function of d(t), in accordance with the definition of the gamma process, be given by

$$f_{d(t)}(d) = Ga(d|\alpha(t),\lambda)$$
(4.11)

with mean and variance as follows:

$$E(d(t)) = \frac{\alpha(t)}{\lambda}$$
(4.12)

$$Var(d(t)) = \frac{\alpha(t)}{\lambda^2}$$
(4.13)

A pipe is said to fail when its corrosion depth, denoted by d(t), is more than a specific threshold (a_0) . Assuming that the threshold a_0 is deterministic and the time at which failure occurs is denoted by the lifetime *T*. Due to the gamma distributed deterioration, Equation 4.11, the life time distribution can then be written as:

$$F(t) = \Pr(T \le t) = \Pr(d(t) \ge a_0) = \int_0^{a_0} f_{d(t)}(d) d_d = \frac{\Gamma(\alpha(t), a_0 \lambda)}{\Gamma(\alpha(t))}$$
(4.14)

Where $\Gamma(\nu, x) = \int_{t=x}^{\infty} t^{\nu-1} e^{-t} dt$ is the incomplete gamma function for $x \ge 0$ and $\nu > 0$.

To model corrosion in a pipe, in terms of a gamma process, the question that remains to be answered is how its expected deterioration increases over time. The expected corrosion depth at time t may be modelled empirically by a power law formulation (Ahammed and Melchers (1997)):

$$\alpha(t) = ct^b \tag{4.15}$$

for some physical constants c > 0 and b > 0.

Because there is often engineering knowledge available about the shape of the expected deterioration in terms of the exponential parameter b in Equation 4.15, this parameter may be assumed constant. The typical values for b from some examples of expected deterioration according to a power law are presented in the Table 4.1.

Deterioration type	Exponential parameter, <i>b</i>	Reference	

degradation of concrete due to reinforcement corrosion	1	Ellingwood & Mori (1993)
sulfate attack	2	Ellingwood & Mori (1993)
diffusion-controlled aging	0.5	Ellingwood & Mori (1993)
creep	1/8	Cinlar et al. (1977)
expected scour-hole depth	0.4	Hoffmans & Pilarczyk (1995) and van Noortwijk & Klatter (1999)

Table 4.1 Typical values for exponential parameter, b, in different deterioration types

The reliability analysis approach which is developed in this section by using gamma process concept is entitled 'Gamma Distributed Degradation, GDD' model.

In the event of expected deterioration in terms of a power law (i.e., Equation 4.15), the parameters c and λ can be estimated by using statistical estimation methods. The estimation procedure is discussed for the two scenarios including a case with available corrosion depth data and a case of unavailability of corrosion depth data.

4.4.2 Developing gamma distributed degradation model with available corrosion depth data

In this section using gamma distributed degradation (GDD) model for reliability analysis of corrosion affected pipes in case of availability of corrosion depth data is discussed. The data of corrosion depth can be achieved by periodical inspections.

To model the corrosion as a gamma process with shape function $\alpha(t) = ct^b$ and scale parameter λ , the parameters c and λ should be estimated. For this purpose, statistical methods are suggested. The two most common methods that can be used for parameter estimation are the maximum likelihood and method of moments. Both methods for deriving the estimators of c and λ were initially presented by Cinlar et al. (1977) and were developed by van Noortwijk and Pandey (2003).

a) Maximum Likelihood Estimation

In statistics, maximum-likelihood estimation (MLE) is a method of estimating the parameters of a statistical model. When applied to a data set and given a statistical model, MLE provides estimates for the model's parameters.

In general, for a fixed set of data and underlying statistical model, the method of maximum likelihood selects values of the model parameters that produce a distribution that gives the observed data the greatest probability (i.e., parameters that maximize the likelihood function). Given that *n* observations are denoted by $x_1, x_2, ..., x_n$, the principle of maximum likelihood assumes that the sample data set is representative of the population. This has a probability density function of $f_x(x_1, x_2, ..., x_n; \theta)$, and chooses that value for θ (unknown parameter) that most likely caused the observed data to occur, i.e., once observations $x_1, x_2, ..., x_n$ are given, $f_x(x_1, x_2, ..., x_n; \theta)$ is a function of θ alone, and the value of θ that maximizes the above probability density function is the most likely value for θ .

In the current study a typical data set consists of inspection times t_i , i = 1, ..., n where $0 = t_0 < t_1 < t_2 < \cdots < t_n$, and corresponding observations of the cumulative amounts of deterioration d_i , i = 1, ..., n are assumed to be given as inputs of the model. Figure 4.2 schematically shows a time dependent degradation model in the case of two inspections with a deterministic path.

The maximum-likelihood estimators of c and λ can be determined by maximising the
logarithm of the likelihood function of the increments. The likelihood function of the observed deterioration increments $\delta_i = d_i - d_{i-1}$, i = 1, ..., n is a product of independent gamma densities (van Noortwijk and Pandey (2003)):

$$l(\delta_{1}, \dots, \delta_{n} | c, \lambda) = \prod_{i=1}^{n} f_{d(t_{i})-d(t_{i-1})}(\delta_{i}) = \prod_{i=1}^{n} \frac{\lambda^{c[t_{i}^{b}-t_{i-1}^{b}]}}{\Gamma(c[t_{i}^{b}-t_{i-1}^{b}])} \delta_{i}^{c[t_{i}^{b}-t_{i-1}^{b}]} e^{-\lambda\delta_{i}}$$
(4.16)



Figure 4.2 Time dependent degradation model in case of two inspections

To maximize the logarithm of the likelihood function, its derivatives are set to zero. It follows that the maximum likelihood estimator of λ is:

$$\hat{\lambda} = \frac{\hat{c}t_n^b}{d_n} \tag{4.17}$$

where \hat{c} must be computed iteratively from the following equation:

$$\sum_{i=1}^{n} \left[t_{i}^{b} - t_{i-1}^{b} \right] \left\{ \psi \left(\hat{c} \left[t_{i}^{b} - t_{i-1}^{b} \right] \right) - \log \delta_{i} \right\} = t_{n}^{b} \log \left(\frac{\hat{c} t_{n}^{b}}{d_{n}} \right)$$
(4.18)

where the function $\psi(x)$ is the derivative of the logarithm of the gamma function:

$$\psi(x) = \frac{\dot{\Gamma}(x)}{\Gamma(x)} = \frac{\partial \log \Gamma(x)}{\partial x}$$
(4.19)

b) Method of Moments

In statistics, the method of moments is a method of estimation of population parameters such as mean and variance by equating sample moments with unobservable population moments and then solving those equations for the quantities to be estimated. Assuming transformed times between inspections as $w_i = t_i^b - t_{i-1}^b$, i = 1, ..., n, the method-of-moments estimates c and λ can be found from (van Noortwijk and Pandey (2003)):

$$\frac{\hat{c}}{\hat{\lambda}} = \frac{\sum_{i=1}^{n} \delta_i}{\sum_{i=1}^{n} w_i} = \frac{d_n}{t_n^b} = \bar{\delta}$$
(4.20)

$$\frac{d_n}{\widehat{\lambda}} \left(1 - \frac{\sum_{i=1}^n w_i^2}{\left[\sum_{i=1}^n w_i\right]^2} \right) = \sum_{i=1}^n \left(\delta_i - \overline{\delta} w_i \right)^2 \tag{4.21}$$

The first equation from both methods (i.e., Equations 4.17 and 4.20) are the same and the second equation in the method of moments is simpler since it does not necessarily require iterations to find the unknown parameter (\hat{c}).

The flowchart in Figure 4.3 illustrates the gamma distributed degradation model in case of availability of corrosion measurements. To use this procedure, at least two measures of corrosion depth should be available for calculation of δ_i in Equations 4.18 and 4.21.



Figure 4.3 Gamma distributed degradation (GDD) Model in case of availability of corrosion depth data

4.4.3 Developing gamma distributed degradation model in case of unavailability of corrosion depth data

In practice, most of the time for reliability analysis of corrosion affected pipes, data such as corrosion depth are not available. Therefore, a method should be developed for such cases of using the gamma distributed degradation model. As it was mentioned in section 4.4.1, in order to calculate the probability of failure over elapsed time (Equation 4.14), the parameters corresponding to shape and scale parameters (α and λ) should be estimated. The steps for this purpose are:

- a) Determining the approximate moments (mean and variance)
- b) Estimating values for α and λ by using Equations 4.12 and 4.13

Assuming $X_1, X_2, ..., X_n$ as basic random variables, moment approximation (i.e., step (a)) can be carried out by expanding the function $Y = Y(X_1, X_2, ..., X_n)$ in a Taylor series about the point defined by the vector of the means $(\mu_{X_1}, \mu_{X_2}, ..., \mu_{X_n})$. By truncating the series, the mean and variance are (Papoulis and Pillai (2002)):

$$E(Y) \approx Y(\mu_{X_1}, \mu_{X_2}, \dots, \mu_{X_n}) + \frac{1}{2} \sum_{i=1}^n \sum_{j=1}^n \frac{\partial^2 Y}{\partial X_i \partial X_j} \operatorname{cov}(X_i, X_j)$$
(4.22)

$$\operatorname{var}(\mathbf{Y}) \approx \sum_{i}^{n} \sum_{j}^{n} c_{i} c_{j} \operatorname{cov}(\mathbf{X}_{i}, \mathbf{X}_{j})$$
(4.23)

The flowchart in Figure 4.4 illustrates the gamma distributed degradation model in case that corrosion measurements are not available. The procedure will be used for reliability analysis of concrete sewers and cast iron pipes in Chapters 5 and 6.



Figure 4.4 Gamma distributed degradation (GDD) Model in case of unavailability of corrosion depth data

4.5 Monte Carlo simulation method

Monte Carlo simulation has been successfully used for reliability analysis of different structures and infrastructure (e.g., Camarinopoulos et al. (1999), Melchers (1999), Sadiq et al. (2004) and Yamini (2009)). Hence, the method is used as a verification method to check the results which are obtained from application of the two time dependent analytical method (i.e, first passage probability method and gamma distributed degradation model).

Monte Carlo simulation techniques involve sampling at random to artificially simulate a large number of experiments and to observe the results. To use this method in structural reliability analysis, a value for each random variable is selected randomly (\hat{x}_i) and the limit state function $(G(\hat{x}))$ is checked. If the limit state function is violated (i.e. $G(\hat{x}) \leq 0$), the structure or the system has failed. The experiment is repeated many times, each time with randomly chosen variables. If N trials are conducted, the probability of failure then can be estimated by dividing the number of failures to the total number of iterations:

$$P_f \approx \frac{n(G(\hat{x}) \le 0)}{N} \tag{4.24}$$

The accuracy of Monte Carlo simulation result depends on the sample size generated and, in the case when the probability of failure is estimated, on value of the probability (the smaller the probability of failure, the larger the sample size needed to ensure the same accuracy). The accuracy of the failure probability estimates can be checked by calculating their coefficient of variation (e.g., Melchers (1999)).

In order to improve the accuracy of estimating the probability of ultimate strength failure, while keeping the computation time within reasonable limits, variance reduction techniques (e.g., importance sampling, Latin hypercube, and directional simulation) can be employed. However, in cases that the main emphasis is on serviceability failure, which can be estimated by a crude Monte Carlo simulation with very good accuracy within a relatively short computation time, such techniques are not necessary to be used (Val and Chernin (2009)).

Importance sampling is a variance reduction technique that can be used in the Monte Carlo method (Melchers (1999)). The idea behind importance sampling is that certain values of the input random variables in a simulation have more impact on the parameter being estimated than others. If these "important" values are emphasized by sampling more frequently, then the estimator variance can be reduced. Hence, the basic methodology in importance sampling is to choose a distribution which "encourages" the important values. The use of "biased" distributions will result in a biased estimator if it is applied directly in the simulation. However, the simulation outputs are weighted to correct use of the biased distribution, and this ensures that the new importance sampling estimator is unbiased.

The fundamental issue in implementing importance sampling simulation is the choice of the biased distribution which encourages the important regions of the input variables. Choosing or designing a good biased distribution is the "art" of importance sampling. The rewards for a

good distribution can be significant run-time savings; the penalty for a bad distribution can be longer run times than for a general Monte Carlo simulation without importance sampling. The details of the Monte Carlo method including sampling techniques can be found in Ditlevsen and Madesn (1996), Melchers (1999) and Rubinstein and Kroese (2008).

4.6 Summary

In this chapter, two probabilistic methods for time-dependent reliability analysis were presented. These two methods are the: (a) first passage probability method and (b) the gamma distributed degradation (GDD) model.

The gamma distributed degradation model was considered for two different scenarios: (a) where corrosion-depth data is available and, (b) where it is not.

The *availability* of deterioration data has already been considered in a previous study of sea defense structures (van Noortwijk et al. 2007). However, the current research will develop this approach for corrosion-affected buried pipes. In addition the same approach will be modified for corrosion scenarios where data is unavailable. This work has already been published in peer-reviewed journal papers, illustrating the novelty and robustness of this approach.

The application of first passage probability theory together with a Monte Carlo simulation, incorporating the new code developed in this work, the improved prediction of service performance has been achieved (Mahmoodian and Li 2011a and 2011b). Further work on the use of gamma process concept (Mahmoodian and Alani 2013a & 2013b)) has shown that this approach is very straight forward as the gamma distribution fits the monotonic progression of the corrosion processes.

In the next two chapters, the first passage probability method and the gamma distributed degradation model devised in this study will be applied to real case data from concrete sewers

78

and cast iron water mains in the UK. This will check the capability of the methods for the reliability analysis of corrosion affected buried pipes in practice.

5 APPLICATION OF THE DEVELOPED METHODS TO CONCRETE SEWERS

5.1 Preface

In this chapter the two model approaches outlined in chapter 4 (i.e., first passage probability method and gamma distributed degradation model) are used for the reliability analysis and service life prediction of a concrete sewer pipeline in the UK. The results are verified by a Monte Carlo simulation and the weakness and strength of each method is discussed.

For a comprehensive time-dependent reliability analysis of corrosion-affected concrete sewers, the following steps need to be followed:

- Modelling the corrosion mechanism
- Defining the failure mode(s) (limit state function(s))
- Calculation of the probability of failure
- A sensitivity analysis

Two conditions are considered for the reliability analysis and service life prediction of the sewer pipeline in the UK. Firstly, the analysis is carried out based on an individual limit state function (i.e., considering one failure mode), and secondly, to have a more representative assessment of different failures to test reliability. The novelty of this approach is the consideration of all possible corrosion induced failure modes that exist in practice. The effectiveness and contribution of the different variables (on the failure probability) provides new insight to the sensitivity analysis of concrete sewers.

5.2 Case study

The two developed methodologies in chapter 4 for time dependent reliability analysis of pipes are applied on CCTV data from surveys of concrete sewers in Harrogate in the United Kingdom. Harrogate is a spa town in North Yorkshire, England. The town is a tourist destination with a population of approximately 75000.

The study considers concrete sewers with a 500mm diameter and 25mm of internal concrete cover.

To incorporate the effect of increments in populations of residents and essentially the increase in the flow rate during the system's life time, the modeling assumes flow rates corresponding to relative depths (i.e., depth/diameter) of 0.2, 0.4 and 0.6, each occurring over a period of 25 years.

Reliability analysis and service life prediction of this concrete sewer pipeline is of interest for

asset managers in North Yorkshire, England, to develop a risk-informed and cost-effective strategy in the management and maintenance of concrete sewers. The analysis can also help infrastructure managers to develop rehabilitation or replacement strategies for existing pipe networks with a view to better management of the pipe asset.

In the following sections two conditions are considered for the assessment, firstly the pipeline reliability is analysed by considering individual failure mode and secondly the reliability analysis is carried out by considering multi failure mode.

5.3 Reliability analysis considering individual failure mode

5.3.1 Corrosion model and definition of failure mode (limit state function)

a) Corrosion Model

The corrosion mechanism in concrete sewers was discussed in section 3.2.1 and models for the rate of corrosion and wall thickness reduction were presented in the form of Equations 3.11 and 3.12 respectively.

Recalling the Equation 3.12 from chapter 3, the reduction in wall thickness in elapsed time t, is:

$$d(t) = c.t = 8.05k. (su)^{3/8} j. [DS] \times \frac{b}{p'_{4}}.t$$
(3.12)

Where k is the factor representing the proportion of acid reaction, s is pipe slope, u is the velocity of stream (m/s), j is pH-dependent factor for proportion of H₂S, [DS] is dissolved sulphide concentration (mg/lit), A is acid-consuming capability or Alkalinity, b is the width of the stream surface, \dot{P} is the perimeter of the exposed wall and t is time.

To consider uncertainties about wall thickness reduction due to corrosion, a stochastic model is presented. Considering Equation 3.12, basic random variables affecting thickness reduction includes: k, u, j, [DS], b/P' and A.

The wall thickness reduction due to corrosion is a function of basic random variables as well

as time. It can be expressed as:

$$d(t) = f(k, u, j, [DS], b/P', A, t)$$
(5.1)

Where k, u, j, [DS], b/P' and A are the basic random variables, the probabilistic information of which are presumed available.

Values for the basic random variables in the current case study are presented in Table 5.1.

Basic Variables	Units	Mean	Standard deviation
K	-	0.8	0.1
j	-	0.2	0.04
[DS]	mg/L	1	0.5
u	m/sec	0.6	0.1
b/P'	-	h/D= 0.2 b/P'= 0.36 h/D= 0.4 b/P'= 0.55	0.11
		b/P' = 0.71	0.23
А	-	0.2	0.06

Table 5.1 Statistical data for the basic random variables

b) Definition of failure mode (limit state function)

According to the ASCE manual No.69 (1989), one of the performance criteria related to the stability of concrete sewers is to control the wall thickness reduction under an acceptable limit (normally concrete cover). In the theory of structural reliability this criterion can be expressed in the form of a limit state function as follows:

$$G(d_{max}, d, t) = d_{max}(t) - d(t)$$
(5.2)
where:

d: Reduction in wall thickness due to corrosion (or corrosion depth), (mm)

d_{max}: Maximum permissible reduction in wall thickness (structural resistance or limit), (mm) t: elapsed time

 d_{max} may change with time although in most practical cases it has a constant value prescribed in design codes of practice and manuals.

5.3.2 Calculation of the probability of failure

With the limit state function introduced in the form of Equation 5.2, the probability of failure of the concrete pipe due to the reduction of its wall thickness can be determined by:

$$P(t) = P[G(d_{\max}, d, t) \le 0] = P[d(t) \ge d_{\max}(t)]$$
(5.3)

The two developed methods for time dependent reliability analysis in chapter 4 (i.e., first passage probability method and gamma distributed degradation model are applied for calculation of the probability of failure of the concrete sewer case study in Harrogate in the UK.

a) Using first passage probability method

Equation 5.3 is a typical upcrossing problem that can be solved by using first passage probability theory. In a time dependent reliability problem all or some of basic random variables are modelled as stochastic processes. For the above problem, the sewer failure depends on the time that is expected to elapse before the first occurrence of the stochastic process, d(t), upcrosses a critical limit (the threshold, d_{max}) sometime during the service life of the sewer.

As it was described in section 4.3, the probability of failure of a pipe can be determined by using first passage probability theory from Equation 4.9. Considering Equation 5.3 as the failure definition, reduction of wall thickness (*d*) is the action (load) and d_{max} is its effect or the acceptable limit (resistance). Therefore Equation 4.9 can be reproduced with *d* replacing *S* and d_{max} replacing *R*.

$$P_{f}(t) = \int_{0}^{t} \frac{\sigma_{\dot{d}|d}(t)}{\sigma_{d}(t)} \emptyset\left(\frac{d_{max} - \mu_{d}(t)}{\sigma_{d}(t)}\right) \left\{ \emptyset\left(-\frac{\mu_{\dot{d}|d}(t)}{\sigma_{\dot{d}|d}(t)}\right) + \frac{\mu_{\dot{d}|d}(t)}{\sigma_{\dot{d}|d}(t)} \Phi\left(\frac{\mu_{\dot{d}|d}(t)}{\sigma_{\dot{d}|d}(t)}\right) \right\} d_{\tau}$$
(5.4)

For a given Gaussian stochastic process with mean function $\mu_d(t)$, and auto-covariance function $C_{dd}(t_i, t_j)$, all terms in Equation 5.4 can be determined as outlined in section 4.3 by using the following formulations:

$$\mu_{\dot{d}|d} = E[\dot{d} \mid d = d_{\max}] = \mu_{\dot{d}} + \rho_d \frac{\sigma_{\dot{d}}}{\sigma_d} (d_{\max} - \mu_d)$$
(5.5a)

$$\sigma_{\dot{d}|d} = [\sigma_{\dot{d}}^2 (1 - \rho_d^2)]^{1/2}$$
(5.5b)

where

$$\mu_{\dot{d}} = \frac{d\mu_d(t)}{dt} \tag{5.5c}$$

$$\sigma_{\dot{d}} = \left[\frac{\partial^2 C_{dd}(t_i, t_j)}{\partial t_i \partial t_j}\Big|_{i=j}\right]^{1/2}$$
(5.5d)

$$\rho_{d} = \frac{C_{d\dot{d}}(t_{i},t_{j})}{\left[C_{dd}(t_{i},t_{i}) \cdot C_{\dot{d}\dot{d}}(t_{j},t_{j})\right]^{1/2}}$$
(5.5e)

and the cross-covariance function is

$$C_{dd}(t_i, t_j) = \frac{\partial C_{dd}(t_i, t_j)}{\partial t_j}$$
(5.5f)

$$C_{dd}(t_i, t_j) = \lambda_d^2 \rho_d \mu_d(t_i) \mu_d(t_j)$$
(5.5g)

Where λ_d is the coefficient of variation of the wall thickness reduction which is determined based on Monte Carlo simulations and ρ_d is (auto-) correlation coefficient for the wall thickness reduction between two points in time t_i and t_j . Therefore all variables in Equation 5.4 can be determined.

To estimate the probability of failure due to corrosion, a critical limit for the wall thickness reduction should be established. ASCE manual No.69 (1989) considers exposure of reinforcement as a criterion for failure. Therefore the maximum acceptable limit for wall thickness reduction (*i.e.*, d_{max}) can be considered equal to the thickness of concrete cover

 $(d_{max} = a_0, \text{ concrete cover}).$

The probability of failure due to wall thickness reduction is computed using Equation 5.4 and the results are shown in Figure 5.1. As can be concluded from this figure, the effect of autocorrelation coefficient (ρ_d) in the probability of failure is negligible, specially for the area of interest (i.e., lower probability of failure).



Figure 5.1 Probability of failure for different auto-correlation coefficient, ρ_d , from first passage probability method

b) Using Gamma distributed degradation (GDD) Model

The average rate of wall thickness reduction within time in a concrete sewer is calculated through Equation 3.12 as explained in section 3.2.1. Defining failure as the time when all concrete cover is corroded, the developed algorithm (GDD model) in section 4.4.3 is used for formulation of the probability of failure based on the gamma process concept.

Given that a random deterioration (i.e., corrosion depth, d) has a gamma distribution with shape parameter $\alpha > 0$ and scale parameter $\lambda > 0$, formulation for calculation of the probability of failure for a corrosion affected pipe is developed as it was mentioned in section 4.4.

The failure was defined as the time that all concrete cover on the reinforcement is corroded

(ASCE manual No.69, (1989)), therefore the concrete sewer is said to fail when its corrosion depth, denoted by d(t), is more than a special value (for instance concrete cover, a_0). Assuming that the time at which failure occurs is denoted by the lifetime *T*, due to the gamma distributed deterioration, Equation 4.14 can be used for calculation of the probability of failure.

$$F(t) = \Pr(T \le t) = \Pr(d(t) \ge a_0) = \int_0^{a_0} f_{d(t)}(d) d_d = \frac{\Gamma(\alpha(t), a_0 \lambda)}{\Gamma(\alpha(t))}$$
(4.14)

Where $\alpha(t) = ct^b$ is the shape parameter with physical constants c > 0 and b > 0 and λ is the scale parameter. The parameters $\alpha(t)$ and λ can be estimated by using the estimation method explained in section 4.4.3. For the exponential parameter b, a value of one is assumed (b=1) based on some examples of expected deterioration that have been presented in the Table 4.1.

Figure 5.2 shows the results obtained for the probability of pipe failure by using the GDD method.



Figure 5.2 Probability of failure from gamma distributed degradation (GDD) model

5.3.3 Verification of the results of the probability of failure

To verify the results obtained from the two developed methods in the previous section, Monte Carlo simulation method (see section 4.5 for the detail) is used. The result of the probability of failure is presented in Figure 5.3.

To be able to compare the results obtained from the two different methods, graphs in Figures 5.1 and 5.2 are also illustrated in Figure 5.3. The graph for first passage probability method is taken from the result for $\rho_d = 0.5$.

The comparison shows that the probabilities of failure predicted by the two methods are in good agreements and they can be verified by the results of Monte Carlo simulation method, particularly for small probabilities which are of most practical interest.



Figure 5.3 Verification of the results from the two methods by Monte Carlo simulation method

5.3.4 Sensitivity analysis

For a comprehensive reliability analysis it may be appropriate to assess the effect of different random variables on the service life of the sewer. It is of interest to identify those variables that affect the wall thickness reduction most, so that further studies can focus on those variables. For this purpose, relative contributions of variables to the variance of the limit state function were calculated using Equation 3.22.

Figure 5.4 illustrates sensitivities (α^2) or relative contribution of variables to the variance of the limit state function calculated based on the definition in Equation 3.22, section 3.3.5. It can be concluded that while the contribution of some variables are relatively small, some other variables have considerable contribution in the limit state function. High values of contribution of variables mean that the sensitivity to those variables is more dependent on the actual value of their coefficient of variation. In such cases, more concern should be taken in order to determine relevant parameter values.



Figure 5.4 Relative contributions of random variables in failure function

The GDD model was run to elaborate on the effect of variables with high values of relative contribution (i.e., [DS], A and b/\hat{P}) on the probability of failure. As can be concluded from Figure 5.5, a change in sulphide concentration has a considerable effect on service life of the

concrete sewer.



Figure 5.5 Effect of sulphide concentration on service life of the sewer

Figure 5.6 also shows the effect of sewage flow quantity on the service life of the sewer. The ratio of the width of the stream surface to the perimeter of the exposed wall (b/\dot{P}) has a significant effect on corrosion rate of concrete and consequently the service life of the sewer decreases dramatically by increasing the ratio of b/\dot{P} .



Figure 5.6 Effect of the ratio of the width of the stream surface to the perimeter of the exposed wall (b/\dot{P}) on service life of the sewer

To investigate more about the effect of concrete properties on the service life of concrete sewers, more studies on the effect of change in Alkalinity (acid consuming capability) of

concrete on the probability of failure were carried out. The result of changing alkalinity from 0.14 to 0.22 is presented in Figure 5.7.



Figure 5.7 Probability of failure for different values of alkalinity, A

Further sensitivity studies were carried out to investigate the effect on the reliability index of the level of variability (i.e. coefficient of variation) of each of the major random variables. The reliability index (β) was chosen for this work in preference to the probability of failure, mainly to facilitate the interpretation of the results. Although these two quantities are directly related (Equation 3.19), the interpretation of results would be more appropriate when dealing with reliability index rather than probability of failure.

The random variables chosen for this study were three significant variables in terms of the α^2 contribution in Figure 5.4 (i.e., [DS], A, b/P').

The coefficient of variation for each of these random variables was varied from 0 to 0.5 in steps of 0.1. The coefficient of variation of all other variables was kept constant at the values given in Table 5.1. Figures 5.8 to 5.10 illustrate the results for four different pipeline lifespans (t). A period of service life from t=20yr to t=35yr which results in practical reliability indexes from 1.5 to 4.5 is selected for this study.



Figure 5.8 Reliability index vs. coefficient of variation of [DS] for various values of the pipeline elapsed time



Figure 5.9 Reliability index vs. coefficient of variation of b/P' for various values of the pipeline elapsed time

The results show that the reliability index decreases as time and the coefficient of variation of random variables increases. It is also observed that the variability of these three random variables for low values of t has a more significant effect on the reliability index.



Figure 5.10 Reliability index vs. coefficient of variation of alkalinity (A) for various values of the pipeline elapsed time

5.4 Reliability analysis considering multi failure mode

In this section the case of the concrete sewer pipeline in Harrogate UK, is analysed by considering more than one possible failure mode. This study which is called multi failure mode reliability analysis will eventually end up with a more reliable service life estimation.

5.4.1 Corrosion model and definition of failure modes (limit state functions)

a) Corrosion model

The same corrosion model as presented in Equation (3.22) with k, u, j, [DS], b/P' and A as the basic random variables is considered for the multi failure mode reliability analysis of the concrete sewer.

b) Definition of failure modes (limit state functions)

As it was mentioned earlier (section 4.1), failure can be defined in relation to different possible mechanisms and in the theory of structural reliability, it can be described by a limit state function as it was presented in Equation (4.1):

$$G(R, S, t) = R(t) - S(t)$$
 (4.1)

Where G(R, S, t) is limit state function, S(t) is the action (load effect) at time t and R(t) is

the critical limit (resistance) for the action or its effect. The probability that failure occurs for any load application is the probability of the limit state violation.

For concrete sewers, failure does not necessarily imply structural collapse (ultimate strength failure) but in most cases is indicated by loss of structural serviceability, as characterised by concrete cracking and/or concrete cover loss. In a comprehensive reliability analysis it is of interest to take into account both serviceability failure and ultimate strength failure.

In some cases of reliability analysis of a structure, various limit states such as bending, shear, cracking and deflection may apply in a composition referred to as a 'system'. The detail of system reliability analysis was explained in section 3.3.4. As it was mentioned in that section, in a series system, attainment of any limit state constitutes failure of the structure. All components of a parallel system must fail for system failure to occur.

A concrete sewer can fail in multiple modes due to different limit state violations. Therefore, the probability of the sewer failure should be determined using the methods of systems reliability analysis.

Considering design codes of practice and manuals for reinforced concrete pipe design (ASCE 15-98, (2000) and ASCE 60, (2007)), the four following failure modes (limit state functions) should all be considered for buried concrete sewers:

Flexural limit state:
$$G_1(M_u, M_s, t) = M_u(t) - M_s(t)$$
 (5.6a)

Shear limit state:
$$G_2(V_b, V_s, t) = V_b(t) - V_s(t)$$
 (5.6b)

Excessive crack limit state:
$$G_3(F, F_{cr}, t) = F(t) - F_{cr}(t)$$
 (5.6c)

Cover loss limit state:
$$G_4(\Delta, a_o, t) = \Delta(t) - a_o(t)$$
 (5.6d)

where M_u, V_b, F and Δ are flexural strength, shear strength, crack control factor and concrete thickness reduction respectively. Formulisation of these four resistance modes for a reinforced concrete pipe are presented in Table 5.2.

 M_s, V_s, F_{cr} and a_o are flexural stress, shear stress, crack control limit and concrete cover

which are considered as thresholds for limit state functions.

The formulae presented by design codes for the resistance modes have stationary formats; while in the case of corrosion, the wall thickness of the pipe is a time dependent parameter (i.e., decreases within the time). Hence, the time dependent format of each formula is given in the last column of Table 5.2.

Failure Mechanism		Stationary formulisation		
		Equation Source		Time dependent formulisation
Strength	Flexural Failure	$M_{u} = A_{s}f_{y}\left(d - \frac{a}{2}\right) + N_{u}\left(\frac{h - a}{2}\right)$	ASCE 15-98 (2000)	$M_u(t) = A_s f_y \left(d - \frac{a}{2} \right) + N_u \left(\frac{h(t) - a}{2} \right)$
	Shear Failure	$V_b = 0.083b\phi_v dF_{vp}\sqrt{f_c'} \left(1.1 + 63 \times \frac{A_s}{bd}\right) \left[\frac{F_d F_N}{F_c}\right]$ $F_N = 1 + \frac{N_u}{3.5bh}$	ASCE 15-98 (2000)	$V_b(t) = 0.083b\phi_v dF_{vp}\sqrt{f_c'} \left(1.1 + 63 \times \frac{A_s}{bd}\right) \left[\frac{F_d F_N(t)}{F_c}\right]$ $F_N(t) = 1 + \frac{N_u}{3.5bh(t)}$

Table 5.2 Formulisation of different resistance modes of the concrete sewer

serviceability	Excessive Crack width	$F = \frac{B_1}{52500_f dA_s} \left[\frac{M_s + N_s \left(d - \frac{h}{2} \right)}{ij} - \left(0.083C_1 bh^2 \sqrt{f_c'} \right) \right]$ $j \approx 0.74 + 2.54e/d$ $i = \frac{1}{1 - \frac{jd}{e}}$ $e = \frac{M_s}{is} + d - \frac{h}{2}$	ASCE 15-98 (2000)	$F(t) = \frac{B_1}{5250\phi_f dA_s} \left[\frac{M_s + N_s \left(d - \frac{h(t)}{2} \right)}{i(t)j(t)} - \left(0.083C_1 bh(t)^2 \sqrt{f_c'} \right) \right]$ $j(t) \approx 0.74 + 2.54 \frac{e(t)}{d}$ $i(t) = \frac{1}{1 - \frac{j(t)d}{e(t)}}$ $e(t) = \frac{M_s}{M_s} + d - \frac{h(t)}{2}$
	Loss of Concrete Cover	$\Delta = h - d - \frac{d_b}{2}$	ASCE 60 (2007)	$\Delta(t) = h(t) - d - \frac{d_b}{2}$

Symbols

a: depth of the equivalent rectangular stress block, (mm)

A: the acid-consuming capability of the wall material

 A_s : Area of tension reinforcement in length b, (mm²/m)

b: unit length of pipe, 1000mm

 B_1 : crack control coefficient for effect of spacing and number of layers of reinforcement

c: the average rate of corrosion (mm/year)

 C_1 : Crack control coefficient for type of reinforcement

d: distance from compression face to centroid of tension reinforcement, (mm)

 d_b : diameter of rebar in inner cage, mm

[DS]: Dissolved sulphide concentration (mg/l)

 f_c' : design compressive strength of concrete, (MPa)

- f_y : design yield strength of reinforcement, (MPa)
- F: crack width control factor

 F_c : factor for effect of curvature on diagonal tension (shear) strength in curved components

 F_d : factor for crack depth effect resulting in increase in diagonal tension (shear) strength with decreasing d

 F_N : coefficient for effect of thrust on shear strength

h: overall thickness of member (wall thickness), (mm)

i: coefficient for effect of axial force at service load stress

k: Acid reaction factor

J: is pH-dependent factor for proportion of H_2S

w: the width of the stream surface

P': perimeter of the exposed wall

 M_s : Service load bending moment acting on length b, (Nmm/m)

 M_u : factored moment acting on length b, (Nmm/m)

 N_s : Axial thrust acting on length b, service load condition (+ when compressive, - when tensile), (N/m)

 N_u : Factored axial thrust acting on length b, (+ when compressive, - when tensile), (N/m)

s: is the slope of the pipeline

t: elapsed time

u: is the velocity of the stream (m/sec)

 V_b : basic shear strength of length b at critical section

 Φ : The average flux of H₂S to the wall

 $Ø_f$: strength reduction factor for flexure

 ϕ_{ν} : strength reduction factor for shear

 Δ : reduction in wall thickness due to corrosion, (mm)

 Δ_{max} : Maximum permissible reduction in wall thickness (structural resistance or limit), (mm)

The four limit states (i.e., Equations 5.6a to 5.6d) can be classified in the two main categories of failure modes, namely serviceability limit states and ultimate strength limit states. If a pipe loses its flexural strength and/or its shear strength it has completely failed. Therefore flexural limit state and shear limit state are considered as ultimate strength limit state functions. On the other hand, if a pipe cracks or loses its cover, it is not necessarily failed structurally, but it is failed from a serviceability point of view. Therefore crack limit state and cover loss limit state can be considered as serviceability limit states.

As mentioned earlier (section 4.1), each failure mode happens when the limit state function is violated (i.e., $G_i \leq 0$). To consider all the four modes as a system, it is necessary to clarify the combination of the limit state functions. Figure 5.11 presents the combination which is suggested for a system reliability analysis of the concrete sewer; it is a combination of series and parallel systems. The two serviceability limit states (crack and cover loss) are considered

parallel, because violation of them individually does not fail the whole system. On the other hand violation of flexural limit state and/or shear limit state will cause the failure of the whole system and therefore these two limit states are set in a series combination.



Figure 5.11 System combination of the four limit state functions for multi failure mode reliability analysis of the concrete sewer

5.4.2 Calculation of the probability of failure

According to the theory of systems reliability, the probability of failure for a series system

 $(P_{fs}(t))$ can be estimated by (Thoft-Christensen and Baker (1982)):

$$max[P_{f_i}(t)] \le P_{f_s}(t) \le 1 - \prod_{i=1}^{m} \left[1 - P_{f_i}(t) \right]$$
(5.7)

where $P_{f_i}(t)$ is the probability of failure due to the *i*th failure mode of pipe and *m* is the number of failure modes considered in the system.

Considering the system configuration presented in Figure 5.11 and the upper bound of Equation 5.7, the probability of failure of the whole concrete sewer system can be calculated by the following equation:

$$P_{fs}(t) = 1 - (1 - P_{f_1})(1 - P_{f_2}) \cdot (1 - P_{f_3} \cdot P_{f_4})$$
(5.8)

Where:

 P_{f_1} is the probability of flexural failure P_{f_2} is the probability of shear failure P_{f_3} is the probability of crack failure P_{f_4} is the probability of cover loss failure

For reliability analysis of the sewer, initially, the probability of failure in time t for each failure mode ($P_{f_i}(t)$, i=1,...,4) is estimated by using one of the two proposed methods (first passage probability and/or GDD model). Then Equation 5.8 is used for calculation of the probability of the sewer system failure ($P_f(t)$), considering all four possible failure modes.

a) Using first passage probability method

The results of using first passage probability method (section 4.3) for calculation of the probability of system failure is shown in Figure 5.12.



Figure 5.12 Probability of system failure from first passage probability method

b) Using Gamma Distributed Degradation (GDD) Model

The developed algorithm for GDD model in section 4.4.3 is used for calculation of the probability of system failure. The results of using gamma distributed degradation (GDD) model for calculation of the probability of the sewer system failure are shown in Figure 5.13.



Figure 5.13 Probability of system failure from GDD model

5.4.3 Verification of the results of the probability of failure

The Monte Carlo simulation method (see section 4.5 for the detail) is performed for verification of the results from the two methods in the previous section. Figure 5.14 shows the comparison of the results of the probability of system failure from the three methods (first passage probability, GDD model and Monte Carlo simulation). The graph for first passage probability method is taken from the result for auto-correlations equal to 0.5.

The comparison shows that the probabilities of system failure predicted by the two methods are in good agreements and it can be verified by the results of the Monte Carlo simulation method, particularly for small probabilities which are of most practical interest.



Figure 5.14 Verification of the results from the two methods by Monte Carlo simulation method

5.4.4 Sensitivity analysis

As it was mentioned earlier, for a comprehensive pipeline assessment, the effect of variables on the failure of the concrete sewer can be analysed by performing sensitivity analysis. In view of the large number of variables that affect the corrosion process, and hence the limit state functions, it is of interest to identify those variables that affect the failure most so that more research can focus on those variables.

Unlike individual failure mode assessment (section 5.3.5), the concept of relative contribution can not be used for multi failure mode analysis. It is simply because Equation 3.22 has been presented for calculation of relative contribution in case of individual limit state function. Hence, a new parametric method is developed and applied for sensitivity analysis of the concrete sewer in case of multi failure mode assessment.

To assess how the change in the values of the six random variables (k, u, j, [DS], b/P and A) can affect the service life of the concrete sewer system, the values for each variable are changed from $\mu_i - 2\sigma_i$ to $\mu_i + 2\sigma_i$ (where μ_i is the mean of the random variable and σ_i is its standard deviation). Assuming a Gaussian distribution for the random variables, this range corresponds to 95.4 percent of the possible values of the variable. The results of the analysis by using Monte Carlo simulation method are illustrated in Figure 5.15 (a-f).



Figure 5.15 Probability of system failure of the concrete sewer for different values of basic random variables

It can be concluded that among all variables, the effect of [DS], A and b/\dot{P} on the probability of failure of the sewer is highly remarkable. The disparity shown within the graphs for each of these three variables means that the sensitivity of the failure of the pipeline is more dependent on their actual value. In such cases, more concern should be taken in order to determine relevant parameter values.

For a better comparison of the effectiveness of random variables on the service life of the pipeline, the results in Figures 5.15(a-f) are summarised in the form of Figure 5.16. In this figure, a range value is defined for each random variable as the difference between the maximum and minimum values of the probability of failure in each elapsed time. Therefore a higher range value means wider resultant values for the probability of failures. This figure illustrates which variables contribute most to the probability of the failure of the system. To clarify, this means that the variables have more effect on the service life of the pipeline.

The significance of the three major variables (i.e, [DS], b/\hat{P} and A) on the failure of the concrete sewer had also been concluded in individual failure mode analysis in section 5.3.5.



Figure 5.16 Comparison of range values of basic random variables

Of these three major random variables, the analysis shows the particular significance of dissolved sulfide concentration in the probability of failure of the pipeline, indicating that dissolved sulfide concentration is the most significant variable on the reliability of concrete sewers.

5.5 Summary

In this chapter a comprehensive study for the reliability analysis of a concrete sewer pipeline in the UK subjected to a time dependent deterioration (i.e. corrosion) was performed. Variables which affect the corrosion were investigated and a corrosion model for this type of buried pipe was presented. To deal with uncertainties and scarcity of monitoring data, the use of time dependent reliability analysis methods for failure analysis and service life prediction of the concrete sewer was considered.

The two developed methods for time dependent reliability analysis (i.e., first passage probability and GDD model) in Chapter 4, were applied for the concrete sewer pipeline in the UK. The assessment was carried out by considering two scenarios: individual failure mode and multi failure mode.

In an individual failure analysis scenario, loss of concrete cover was assumed as the criterion for failure of the sewer. Four failure modes including flexural, shear, cracking and cover loss were assumed as possible failure modes for multi failure mode analysis. These modes of failure, which are set into the two categories of serviceability and ultimate strength limit states, were then put into a system configuration consisting of a combination of series and parallel systems.

The results from the two developed methods in chapter 4 (i.e., first passage probability method and GDD model) showed a significant agreement with the results from Monte Carlo simulation method. Although the result from first passage probability method depends to the amount of assumed auto-correlation coefficient (ρ), this study showed that the effect of auto-correlation coefficient on the probability of failure is negligible.

Another concern is that to begin with first passage probability procedure, it is necessary to estimate the coefficient of variation of the deteriorating parameter (i.e. λ_d in Equation 5.5g). As it was mentioned in section 5.3.2 this estimation can be carried out by using a Monte

103

Carlo simulation. It means that using first passage probability method (as an analytical method) still needs application of a numerical simulation (e.g. Monte Carlo) to estimate required parameters (i.e. λ_d). This estimation would be time consuming and will involve errors in calculations.

The obtained results in this chapter showed that a gamma distributed degradation (GDD) model can appropriately model the monotonic behaviour of the aging and deterioration process of pipes. The service life of a concrete sewer can be predicted with high accuracy for different levels of probability of failure.

In addition to the assessment of existing pipes, the reliability analysis methods used in this case study can also show how changes in design parameters of sewers (such as concrete cover) can affect the service life of the pipeline.

Sensitivity analysis has also been undertaken to identify factors that affect the probability of pipe failure due to corrosion. The analysis showed less significant contribution of some variables in failure functions. Therefore, it would not be necessary to consider those parameters as random variables and they can be treated as deterministic constant values for further studies.

The results showed that among six random variables, [DS], b/\dot{P} and A have the most effect on the probability of sewer failure. This effect is more considerable for lower values of time, which means special attention should be taken for accurate determination of these variables for new pipelines.

The new methods adopted in chapter 4 were applied to the reliability analysis of a UK concrete sewer case study. The results were verified by using a Monte Carlo simulation, and more realistic results from multi-failure assessment were achieved. The degree of sensitivity to different variables was also investigated. An extensive discussion and critical analysis of these results is presented in chapter 7.

6 APPLICATION OF THE DEVELOPED METHODS TO CAST IRON WATER PIPES

6.1 Preface

The applicability of the two time-dependent reliability analysis methods proposed in chapter 4 is tested by two UK case studies involving cast iron water pipes. Two different scenarios including individual and multi failure modes are considered during the analysis. The steps used for the reliability analysis of concrete sewers in chapter 5 are followed here, and involve:

- Modelling the corrosion mechanism of cast iron pipes
- Defining the failure mode(s) (limit state function(s))
- Calculation of the probability of failure
- A sensitivity analysis

In each case, both internal and external corrosion are considered and the results are compared and discussed. Although the research on reliability of cast iron pipes has been done by other researchers, the novelty of this study is consideration of both strength failure and fracture failure, in hoop and axial planes, leading to a more realistic and reliable outcome, as demonstrated from the Monte Carlo simulation.

6.2 Reliability analysis considering individual failure mode

6.2.1 Case Study

Underground cast iron pipes in a typical conurbation of the UK are considered here for illustration of the proposed methodologies. The pipes selected have a diameter ranging from 254 to 406 mm and the wall thickness ranging from 16 to 20.3 mm correspondently as shown in Table 6.1. The pipes are made of cast iron with the fracture toughness in a range of $K_C = 7.66$ to 9.25 MPa/m^{0.5} (Marshall (2001)).

Diameter	Thickness	Fracture toughness	Hoop Stress
(mm)	(mm)	Kc (MPa/m ^{0.5})	(MPa)
254	16.0	7.66	44.9
305	17.5	7.86	49.6
406	20.3	9.25	62

Table 6.1 Geometry and stresses in selected pipes

Since the purpose is to demonstrate the application of the two time dependent reliability analysis methods for predicting the service life of cast iron pipes subjected to pitting corrosion, the loads that are applied to the pipes over the service life are taken from the design codes, BS EN 1295 (1997). It has been assumed that the overburden forces can be calculated considering main road traffic and a burial depth of 0.9 m, using the formula given in the national design standard BS EN 1295 (1997). A pressure surge of 1.6 MPa has also been assumed. The stresses produced by these loads are summarized in Table 6.1 for various diameters and thicknesses.

6.2.2 Corrosion model and definition of failure mode (limit state function)

a) Corrosion model

It has been well known that the predominant deterioration mechanism for cast iron pipes is electro-chemical corrosion in the form of corrosion pits. Each spot of metal loss represents a corrosion pit that grows with time and reduces the thickness and mechanical resistance of the pipe wall. This process eventually leads to the collapse of the pipe.

As it was mentioned in chapter 3, a number of models for corrosion of cast iron pipes have been proposed to estimate the depth of corrosion. There are debates in the research community as to whether the corrosion rate can be assumed linear or otherwise (e.g., Kucera and Mattsson (1987), Sheikh et al. (1990), Ahammed and Melchers (1997), Rajani et al. (2000), Sadiq et al. (2004)). The widely used model for corrosion pit (as mentioned in chapter3) is expressed in the form of the following:

$$a = kt^n \tag{6.1}$$

where t is the exposure time and K and n are empirical coefficients which in practice are obtained by fitting the model to experimental data.

To avoid controversy of employing a published model, which may have not gained public acceptance, and more importantly, since the purpose of this research is to propose reliability analysis methods that only use corrosion as an example, the modelling of corrosion pit is based on the experimental data from a UK Water Industry Research report (Marshall (2001)). This corrosion rate data for using in this case study has been illustrated in Figure 6.1. As the regression of available data fits a power law very well with high R-squire value (Figure 6.1, $R^2 = 0.959$ for internal corrosion and $R^2 = 0.857$ for external corrosion), the corrosion can be modelled, for both external and internal corrosion, as follows:

$$\mu_a = 2.54t^{0.32} \text{ for external corrosion}$$
(6.2)

$$\mu_a = 0.92t^{0.4} \text{ for internal corrosion}$$
(6.3)

where μ_a denotes the mean value for the depth of corrosion pit.



Figure 6.1 Corrosion rates for cast iron pipes (reproduced from Marshall (2001))

b) Definition of the failure mode (limit state function)

Cast iron is a brittle material and its failure can be characterised by fracture. The main effect of corrosion pits in cast iron pipes is the localised cracking (or fissuring) in the pipe wall which leads to stress concentration at or around the tip of the corrosion pit and associated reduction in the wall thickness of the pipe and hence the pipe capacity. With the growth of corrosion the pit propagates to a critical size such that the pipe cannot withstand the stresses incurred in the pipe and hence the wall is fractured and the pipe fails.

According to the theory of fracture mechanics, when a structural element (e.g., the pipe wall) under a nominal stress contains a sharp crack (e.g., the corrosion pit), the stress field ahead of the sharp crack (pit) can be described and measured by a single factor, known as stress intensity factor K. It is a parameter that amplifies the effect of stress field at the tips of crack leading to fracture. In essence, *K* serves as a scale factor to define the magnitude of the crack-tip stress field and is related to the geometrical parameters and stress types of the element.

The factor K is a function of far-field stress level σ , the size of the pit, a, the shape and orientation of the pit, and dimensions of the element where the pit occurs. This relationship can be expressed as a general form of (Hertzberg 1996):

$$K(t) = \sigma \sqrt{2\pi r} f(\theta, t) \tag{6.4}$$

where r and θ are the polar coordinates of the pit (crack) tip and f(θ , t) is the correction factor allowing for various geometries of the pit and structural element which also changes with time due to pit growth. Exact expressions for the stress intensity factor K(t) vary and depend on the mode of fracture and geometry of the pit and element. In general, there are three deformation modes of fracture (Hertzberg 1996), which are opening mode (Mode I), in-plane shear mode (Mode II) and out-of-plane shear or tear mode (Mode III). Since Mode I is found to be the dominant cracking condition in pipes under normal service conditions (Marshal 2001), in this research, only Mode I is considered in the failure analysis of the pipe (i.e., a
crack plane is perpendicular to the direction of the stress incurred).

If K_I is stress intensity factor for crack in mode I and K_c is defined as the critical stress intensity factor, known as fracture toughness, beyond which the pipe cannot withstand the pit crack, the probability of pipe collapse (failure) can be defined as follows:

$$P_f(t) = P[K_I(t) \ge K_c] \tag{6.5}$$

In practical application of Equation 6.5 to service life prediction of cast iron pipes, the main effort lies in developing a model of stress intensity factor K(t), i.e., the action process, since the fracture toughness K_c is a material property (ASTM-E1820-01 2001). In this research, two typical scenarios of corrosion pits are considered: (i) external corrosion; and (ii) internal corrosion. Since the focus is only on the hoop stress that leads to Mode I fracture, the following two cases are considered in determining the stress intensity factors of cast iron pipes.

Case 1 External surface pit under hoop stress. The typical example of this case is a pipe under internal pressure (e.g., water) and subjected to external corrosion. Figure 6.2 shows an idealised section of a cast iron pipe with external corrosion pit which is assumed to be semielliptical. In Figure 6.2, 2c is the length of the semi-elliptical pit, a is the radial depth of the pit through the pipe wall, R is the inner radius, d is the thickness of the pipe wall and the angle θ is used to describe the position around the semi-elliptical pit which varies between $0 \le \theta \le \pi$. In this case, the stress intensity factor for Mode I fracture of a semi-elliptical surface pit K_{II} can be determined as follows (Raju and Newman 1982):

$$K_{I1} = \sigma_{hoop} \sqrt{\pi \frac{a}{Q}} F_e(\frac{a}{d}, \frac{a}{c}, \frac{d}{R}, \theta)$$
(6.6)

where σ_{hoop} is the hoop stress caused by internal pressure in the pipe, F_e is the boundary correction factor for a semi-elliptical surface pit located outside the pipe. Since the growth of pit (i.e., α), increases with time and hence K_{I1} is time-variant.

In Equation (6.6), Q is the semi-elliptical pit correction factor and can be determined from (Raju and Newman 1982):

$$Q = 1 + 1.464 \left(\frac{a}{c}\right)^{1.65}$$

$$(6.7)$$

Figure 6.2 External semi-elliptical surface pit on a pipe

Also in Equation 6.6, F_e can be determined from

$$F_{e} = \frac{d}{R} \left(\frac{R^{2}}{R_{o}^{2} - R^{2}}\right) \left[2G_{0} + 2\frac{a}{R_{o}}G_{1} + 3\left(\frac{a}{R_{o}}\right)^{2}G_{2} + 4\left(\frac{a}{R_{o}}\right)^{3}G_{3}\right]$$
(6.8)

where R_o is outer radius of the pipe and G_j (j = 0, 1, 2, 3) are the influence coefficients for jth stress distribution on crack (pit) surface. The influence coefficients G_j depend on the geometry of the pits and the pipe as represented by *a*, *c*, *d* and *R*. Exact values of G_j for a semi-elliptical external surface pit with given geometric parameters can be obtained from finite element analysis as discussed in details in Raju and Newman (1982). For the calculation of probability of failure, using Equation 6.5, an analytical expression can be developed for G_j based on mathematical regression as outlined in the following.

Raju and Newman (1982) employed finite element method to determine the stress intensity factors for pipes with various geometry and corrosion pits. In their study the exact values of the influence coefficients, i.e., G_j (j = 0, 1, 2, 3) are tabulated. Based on these tables

analytical expressions for G_j (j = 0, 1, 2, 3) can be obtained using mathematical regressions for given values of geometry of a pipe and corrosion pit. Using regression methods can change un-continuous data from the tables to continuous formulae which can be used in calculation of F_e and consequently in the reliability analysis method.

To obtain the maximum value of K_{I1} , it is measured at the deepest point of the pit, i.e., $\theta = 90^{\circ}$. It is also assumed that c, the half-length, and a, the radial depth of the semi-elliptical surface pit, are equal (Hertzberg 1996). In addition for most pipes used by industry the ratio of wall thickness to radius, i.e., d/R, is in a range of 0.1-0.2.

Using least squares regression method for the data from tables of Raju and Newman (1982) and for $\frac{d}{R} = 0.1$ and $\frac{2\theta}{\pi} = 1.0$, the expressions for influence coefficients G_j for semi-elliptical surface crack on the outside surface of a pipe can be derived as follows:

$$G_0 = \left(5.7981 \left(\frac{a}{c}\right)^2 - 9.1619 \frac{a}{c} + 3.5187\right) \frac{a}{d} - 1.2154 \left(\frac{a}{c}\right)^2 + 1.8072 \frac{a}{c} + 0.4044 \qquad (6.9a)$$

$$G_{1} = \left(2.0031 \left(\frac{a}{c}\right)^{2} - 3.1684 \frac{a}{c} + 1.2436\right) \frac{a}{d} - 0.3644 \left(\frac{a}{c}\right)^{2} + 0.6521 \frac{a}{c} + 0.4135$$
(6.9b)

$$G_{2} = \left(1.0237 \left(\frac{a}{c}\right)^{2} - 1.6222 \frac{a}{c} + 0.6468\right) \frac{a}{d} - 0.1531 \left(\frac{a}{c}\right)^{2} + 0.3594 \frac{a}{c} + 0.3726 \qquad (6.9c)$$

$$G_3 = \left(0.6387 \left(\frac{a}{c}\right)^2 - 1.0082 \frac{a}{c} + 0.4028\right) \frac{a}{d} - 0.0802 \left(\frac{a}{c}\right)^2 + 0.2521 \frac{a}{c} + 0.3328 \qquad (6.9d)$$

Case 2 Internal surface pit under hoop stress. The internal surface corrosion pit can also be assumed to be semi-elliptical with length 2c and radial depth a as shown in Figure 6.3. In this case, the stress intensity factor for an internal semi-elliptical surface pit K_{I2} can also be determined from Equation 6.6 with F_i replacing F_e as follows (Raju and Newman 1982):

$$K_{I2} = \sigma_{hoop} \sqrt{\pi \frac{a}{Q}} F_i(\frac{a}{d}, \frac{a}{c}, \frac{d}{R}, \theta)$$
(6.10)

where F_i is the boundary correction factor for a semi-elliptical surface pit located inside the pipe and can be determined from:

$$F_{i} = \frac{d}{R} \left(\frac{R_{o}^{2}}{R_{o}^{2} - R^{2}}\right) \left[2G_{0} - 2\frac{a}{R}G_{1} + 3\left(\frac{a}{R}\right)^{2}G_{2} - 4\left(\frac{a}{R}\right)^{3}G_{3}\right]$$
(6.11)

Figure 6.3 Internal semi-elliptical surface pit in a pipe

It is assumed that *c*, the half-length, and *a*, the radial depth, of the semi-elliptical surface pit, are equal (Hertzberg 1996). Likewise the external corrosion pit, the expressions for the influence coefficients G_j (j = 0, 1, 2, 3) for semi-elliptical surface crack on the inside surface of a pipe can be derived as follows:

$$G_0 = \left(3.6877 \left(\frac{a}{c}\right)^2 - 5.84671 \frac{a}{c} + 2.2834\right) \frac{a}{d} - 0.6844 \left(\frac{a}{c}\right)^2 + 1.0071 \frac{a}{c} + 0.6665 \quad (6.12a)$$

$$G_{1} = \left(1.28448 \left(\frac{a}{c}\right)^{2} - 2.0459 \frac{a}{c} + 0.8361\right) \frac{a}{d} - 0.1921 \left(\frac{a}{c}\right)^{2} + 0.3943 \frac{a}{c} + 0.495 \qquad (6.12b)$$

$$G_{2} = \left(0.6521 \left(\frac{a}{c}\right)^{2} - 1.0462 \frac{a}{c} + 0.4442\right) \frac{a}{d} - 0.0658 \left(\frac{a}{c}\right)^{2} + 0.2305 \frac{a}{c} + 0.411$$
(6.12c)

$$G_3 = \left(0.295 \left(\frac{a}{c}\right)^2 - 0.502 \frac{a}{c} + 0.2403\right) \frac{a}{d} + 0.051 \left(\frac{a}{c}\right)^2 + 0.0669 \frac{a}{c} + 0.3851$$
(6.12d)

It should be noted that, in this study conservative assumptions and simplified form of Equations 6.6 to 6.12 have been adopted to determine the maximum values of K_{I1} and K_{I2} ,

which is based on the worst case scenario and measured at the deepest point of the pit, i.e., θ =90°.

6.2.3 Calculation of the probability of failure

a) Using first passage probability method

The first passage probability method is proposed in this section to quantitatively assess the probability of the pipe failure over a period of time. The concept of stress intensity in fracture mechanics is employed to establish the failure criterion, i.e., the limit state function used to determine the probability of pipe failure.

As it was already mentioned in section 4.2.1, probability of failure of a pipe can be determined by using first passage probability theory from Equation 4.9. Considering Equation 6.5 as the failure definition, stress intensity factor (K_I) is the action (load) and Kc is its effect or the acceptable limit (resistance). Therefore Equation (4.9) can be reproduced with K_I replacing *S* and *Kc* replacing *R*.

$$P_{f}(t) = \int_{0}^{t} \frac{\sigma_{\dot{K}_{I}|K_{I}}(t)}{\sigma_{K_{I}}(t)} \phi\left(\frac{K_{c} - \mu_{K_{I}}(t)}{\sigma_{K_{I}}(t)}\right) \left\{ \phi\left(-\frac{\mu_{\dot{K}_{I}|K_{I}}(t)}{\sigma_{K_{I}|K_{I}}(t)}\right) + \frac{\mu_{\dot{K}_{I}|K_{I}}(t)}{\sigma_{\dot{K}_{I}|K_{I}}(t)} \Phi\left(\frac{\mu_{\dot{K}_{I}|K_{I}}(t)}{\sigma_{K_{I}|K_{I}}(t)}\right) \right\} d_{\tau}$$
(6.13)

For a given Gaussian stochastic process with mean function $\mu_{K_i}(t)$, and auto-covariance function $C_{K_iK_i}(t_i, t_j)$, all terms in Equation 6.13 can be determined as outlined in section 4.3 as follows:

$$\mu_{\dot{K}_{I}|K_{I}} = E[\dot{K}_{I} | K_{I} = Kc] = \mu_{\dot{K}_{I}} + \rho_{K_{I}} \frac{\sigma_{\dot{K}_{I}}}{\sigma_{K_{I}}} (Kc - \mu_{K_{I}})$$
(6.14a)

$$\sigma_{\dot{K}_{I}|K_{I}} = [\sigma_{\dot{K}_{I}}^{2} (1 - \rho_{K_{I}}^{2})]^{1/2}$$
(6.14b)

where

$$\mu_{\dot{K}_{I}} = \frac{d\mu_{K_{I}}(t)}{dt}$$
(6.14c)

$$\sigma_{\dot{K}_{i}} = \left[\frac{\partial^{2} C_{K_{i}K_{i}}(t_{i}, t_{j})}{\partial t_{i}\partial t_{j}}\Big|_{i=j}\right]^{1/2}$$
(6.14d)

$$\rho_{K_{I}} = \frac{C_{K_{I}\dot{K}_{I}}(t_{i},t_{j})}{\left[C_{K_{I}K_{I}}(t_{i},t_{i}) \cdot C_{\dot{K}_{I}\dot{K}_{I}}(t_{j},t_{j})\right]^{1/2}}$$
(6.14e)

and the cross-covariance function is

$$C_{K_i \dot{K}_i}(t_i, t_j) = \frac{\partial C_{K_i K_i}(t_i, t_j)}{\partial t_j}$$
(5.14f)

$$C_{K_{I}K_{I}}(t_{i},t_{j}) = \lambda_{K_{I}}^{2}\rho_{K_{I}}\mu_{K_{I}}(t_{i})\mu_{K_{I}}(t_{j})$$
(6.14g)

Where λ_{K_I} is the coefficient of variation of the stress intensity factor which is determined based on Monte Carlo simulations and ρ_{K_I} is (auto-) correlation coefficient for the stress intensity factor between two points in time t_i and t_j . Therefore all variables in Equation 6.13 can be determined.

Values of the influence coefficients, i.e., Gj (j = 0, 1, 2, 3) in Equations 6.8 and 6.11 are calculated by using equations 6.9(a-d) and 6.12(a-d) for external and internal corrosion respectively. The stresses produced by the existing loads are used from the values given in Table 6.1 for various diameters and thicknesses.

With this preparation, the stress intensity factors for two different cases can be calculated by Equations 6.6 and 6.10, respectively. The time-dependent probability of pipe failure can finally be calculated using Equation 6.13.

Typical results for each case are shown in Figures 6.4 and 6.5. These Figures indicate that the effect of the auto-correlation of the stress intensity factors between two points in time (i.e., ρ_{K_I}) on pipe collapse can be significant which justifies that using first passage probability method without having sufficient data (i.e., auto-correlation coefficient) for this example is not a desired method.



Figure 6.4 Probability of pipe collapse for different ρ_{K_I} , for case 1 (external corrosion), using first passage probability method (D = 305mm, $\lambda_{K_I} = 0.41$)



Figure 6.5 Probability of pipe collapse for different ρ_{K_I} , for case 2 (internal corrosion), using first passage probability method (D = 254nm, $\lambda_{K_I} = 0.41$)

b) Using Gamma distributed degradation (GDD) Model

With the failure definition in Equation 6.5, the developed algorithm (GDD model) in section 4.4.2 is used for formulation of the probability of failure based on gamma process concept. Given that a random process (i.e., stress intensity factor, K_I) has a gamma distribution with

shape parameter $\alpha > 0$ and scale parameter $\lambda > 0$, formulation for calculation of the probability of failure for a corrosion affected pipe is developed as it was mentioned in section 4.4.

Assuming that the time at which failure occurs is denoted by the lifetime T, due to the gamma distributed deterioration, Equation 4.14 can be reproduced for calculation of the probability of failure.

$$F(t) = \Pr(T \le t) = \Pr(K_I(t) \ge K_c) = \int_0^{K_c} f_{K_I(t)}(K_I) d_{K_I} = \frac{\Gamma(\alpha(t), K_c \lambda)}{\Gamma(\alpha(t))}$$
(6.15)

Where $\alpha(t) = ct^{b}$ is the shape parameter with physical constants c > 0 and b > 0 and λ is the scale parameter. The parameters $\alpha(t)$ and λ can be estimated by using the estimation methods explained in section 4.4.2. For the exponential parameter b, a value of one is assumed (b=1) based on some examples of expected deterioration that have been presented in Table 4.1.

Figures 6.6 and 6.7 show the results obtained for the probability of pipe failure by using the GDD model for different cases.



Figure 6.6 Probability of pipe collapse for Case 1 (external corrosion) with different diameters, using GDD Model



Figure 6.7 Probability of pipe collapse for Case 2 (internal corrosion) with different diameters, using GDD Model

6.2.4 Verification of the results of the probability of failure

To verify the results obtained from GDD model in the previous section, Monte Carlo simulation method is used. A MATLAB code was written to perform Monte Carlo simulation (see section 4.5 for the detail). The results are illustrated in Figures 6.8 and 6.9, involving all the results from the GDD model for each case. The results from using first passage probability are not considered for comparison because of high dependency to the value of auto-correlation.

The comparison shows that the probabilities of failure predicted by GDD model are in good agreements with the results of Monte Carlo simulation method, particularly for small probabilities which are of most practical interest.



Figure 6.8 Verification of the results from GDD model by Monte Carlo simulation method for Case 1 (external corrosion) with different diameters



Figure 6.9 Verification of the results from GDD model by Monte Carlo simulation method for Case 2 (internal corrosion) with different diameters

6.2.5 Sensitivity analysis

To assess the effect of the pipe material property (i.e., toughness) on the probability of failure, the GDD model is run for different toughness of the pipe material (K_c) . The results of

this parametric sensitivity analysis are illustrated in Figure 6.10. These results are selfevident; the tougher the pipe is (i.e., the greater the fracture toughness), the smaller the probability of its failure.

In view of variables that affect the corrosion process, it is of interest to identify the degree of contribution of variables so that more research can focus on the most effective variable. The contribution of these variables in the failure function is calculated by using relative contribution concept (Chapter 3, Equation 3.22).

 α_x^2 represents the relative contribution of random variables (k and n) in the violation of the limit state function. Figure 6.11 shows the degree of contribution of each variable during the time of service.



Figure 6.10 Probability of pipe failure for different K_c for case 1 (external corrosion), using GDD Model (D = 305mm)

Further sensitivity studies were carried out to investigate the effect on probability of failure of the level of variability (i.e. coefficient of variation) of each of corrosion model parameters (i.e. k and n). The coefficient of variation for each of these parameters was varied from \bullet to 0.5 in steps of 0.1. Figures 6.12 and 6.13 illustrate the results for three different pipeline

elapsed lives (t). Generally the probability of failure is more sensitive to the variation of the coefficient of variation of exponential constant (n).



Figure 6.11 Relative contribution to the variance of failure function for corrosion multiplying constant (K), and corrosion exponential constant (n), a) Case 1, External Corrosion, b) Case 2, Internal Corrosion

It is also observed that the variability of the parameters (k and n) for low values of t, has more significant effect on the probability of failure.



Figure 6.12 Probability of failure due to external corrosion (case 1) Vs coefficient of variation for various values of pipeline elapsed life ((a) corrosion multiplying constant, K, and (b) corrosion exponential constant, n)



(a) (b)
 Figure 6.13 Probability of failure due to internal corrosion (case 2) Vs coefficient of variation for various values of pipeline elapsed life ((a) corrosion multiplying constant, *K*, and (b) corrosion exponential constant, *n*)

6.3 Reliability analysis considering multi failure mode

Cast iron pipes can fail in many modes which in general can be summarized in two categories: loss of strength due to the reduction of wall thickness of the pipes, and loss of toughness due to the stress concentration at the tips of cracks or defects. Even in one category there can be many mechanisms that cause failure. For the example of strength failure it can be caused by hoop stress or axial stress in the pipes. A review of most recent research literature (Sadiq et al. (2004), Moglia et al. (2008), Yamini (2009) and Clair and Sinha (2012)) suggests that current research on pipe failures focuses more on loss of strength than loss of toughness. As it was mentioned in section 3.3.7(b) literature review also revealed that in most reliability analyses for buried pipes, multi failure modes are rarely considered even in practice this is the reality. Therefore the aim of this section is to consider multi failure modes in reliability analysis and service life prediction for cast iron pipes. Both loss of strength and toughness of the pipe are considered. A system reliability method is employed in calculating the probability of pipe failure over time, based on which the service life of the pipe can be estimated. Sensitivity analysis is also carried out to identify those factors that affect the pipe behavior most.

6.3.1 Case study

A cast iron pipe of 254 mm diameter and effective length of 6.5 m is considered as an example to illustrate the proposed method. Other data for the cast iron pipe required for calculation is presented in Table 6.2. Totally 15 random variables are involved in the problem in which their statistical data have been presented in the Table.

6.3.2 Corrosion model and definition of failure modes (limit state functions)

a) Corrosion model

Same as individual failure analysis, the widely used corrosion model (i.e., Equation (6.1)) is selected for the multi failure mode reliability analysis of the cast iron pipe. Therefore the statistical values (mean and standard deviation) for k and n in Equation (6.1) are again taken from the mathematical regression (Figure 6.1) to the data from Marshall (2001). Based on this data mean and standard deviation for corrosion coefficients (k and n) have been estimated (Table 6.2).

b) Definition of the failure modes (limit state functions)

Buried pipes are not only subjected to mechanical actions (loads) but also environmental actions that cause the corrosion of pipes. Corrosion related defects would subsequently cause fracture of cast iron pipes. In the presence of corrosion pit, failure of a pipe can be attributed to two mechanisms: (1) the stresses in the pipe exceed the corresponding strength or (2) the stress intensity exceeds fracture toughness of the pipe. Based on these two failure modes, two limit state functions can be established as follows.

Symbol	Variable	<u> </u>	Units	Min.	Mean	St. Dev.	Max	References
Р	Internal Pressure		MPa	0.35	0.45	0.1	0.7	EPB 276 (2004)
D	Inner diameter		mm	240	254	14.28	260	BS78-2 (1965)
d	Wall thickness		mm	-	16	0.7	-	BS78-2 (1965)
K _m	Bending moment coefficient		-	-	0.235	0.04	-	Sadiq et al. (2004)
C _d	Calculation coefficient		-	-	1.32	0.20	-	Sadiq et al. (2004)
B _d	Width of ditch		mm	-	625	125	-	AWWA C600, (2005)
E_P	Modulus of elasticity of pipe		MPa	-	105000	15000	-	BS78-2 (1965)
K _d	Defection coefficient		-	-	0.108	0.0216	-	Sadiq et al. (2004)
I _c	Impact factor		-	-	1.5	0.375	-	Sadiq et al. (2004)
C_t	Surface load Coefficient		-	-	0.12	0.024	-	Sadiq et al. (2004)
F	Wheel load		Ν	30000	412000	20000	100000	Sadiq et al. (2004)
Α	Pipe effective length		mm	-	6500	200	-	AWWA C600, (2005)
γ	Soil Unit weight		N/mm ³	-	18.85×10 ⁻⁶	18.85×10 ⁻⁷	-	Sadiq et al. (2004)
k	Multiplying constant	Internal	-	-	0.92	0.18	-	Marshall (2001)
		External	-	-	2.54	0.5	-	Marshall (2001)
п	Exponential constant	Internal	-	-	0.4	0.08	-	Marshall (2001)
		External	-	-	0.32	0.06	-	Marshall (2001)

Table 6.2 Values of basic random variables used in the case study

Strength limit state. Rajani et al. (2000) developed a formula for total stresses in a buried pipe including both hoop and axial stresses (see Figure 6.14):

$$\sigma_{\rm h} = \sigma_{\rm F} + \sigma_{\rm S} + \sigma_{\rm L} + \sigma_{\rm V} \tag{6.16}$$

where σ_h is the total hoop or circumferential stress in the pipe, σ_F , σ_S , σ_L and σ_V are the hoop stresses due to internal fluid pressure, soil pressure, frost pressure and traffic stresses respectively.



Figure 6.14 Stresses and cracks on a pipe wall (σ_h : hoop stress and σ_a : axial stress)

Similarly the total axial or longitudinal stress in the pipe can be expressed as:

 $\sigma_a = \sigma_{Te} + \sigma_P + (\sigma_S + \sigma_L + \sigma_V) \nu_p$ (6.17) where σ_a is the total axial stress in the pipe, σ_{T_e} is the stress related to temperature difference, σ_P is the axial stress due to internal fluid pressure, ν_p is Poisson's ratio of pipe material. Details of the equations and references used for determining the above stresses are presented in Table 6.3.

In practice, a pipe is usually under both axial and hoop stresses (σ_a and σ_h). If the yield strength of the pipe material is σ_y , the two limit state functions for strength can be established as follows

Hoop stress limit state:
$$G_1(\sigma_y, \sigma_h, t) = \sigma_y - \sigma_h(t)$$
 (6.18)

Axial stress limit state:
$$G_2(\sigma_y, \sigma_a, t) = \sigma_y - \sigma_a(t)$$
 (6.19)

Toughness limit state. For localised stress concentration caused by defects, e.g., corrosion pits, the term stress intensity factor, K_I , is used (as it was mentioned in section 6.2.2(b)) to more accurately predict the stress state ("stress intensity") near the tip of a crack (caused by applied or residual stresses).

Stress Type	Model*	Reference		
σ_F , hoop stress due to internal fluid pressure	pD 2d	Rajani et al. (2000)		
σ_S , soil pressure	$\frac{3K_m \gamma B_d^2 C_d E_P d D}{E_P d^3 + 3K_d p D^3}$	Ahammed & Melchers (1994)		
σ_V , Traffic stress	$\frac{3 \text{ K}_{\text{m}} \text{ I}_{\text{c}} \text{ C}_{\text{t}} \text{ F} \text{ E}_{\text{P}} \text{ d} \text{ D}}{\text{A}(\text{E}_{\text{P}} \text{ d}^3 + 3\text{K}_{\text{d}} \text{ p} \text{ D}^3)}$	Ahammed & Melchers (1994)		
σ_{T_e} , Thermal stress	$- E_P \alpha_P \Delta T_e$	Rajani et al. (2000)		
σ_P , axial stress due to internal fluid pressure	$\frac{p}{2} \Bigl(\frac{D}{d} - 1 \Bigr) \nu_p$	Rajani et al. (2000)		

Table 6.3 Models for stresses on buried pipes considered in this study

*Notations are in Table 6.2

Instead of using Equations 6.6 and 6.10 which have been developed just for hoop stresses by Raju and Newman (1982), the formulations presented by Laham (1999) are used for calculation of stress intensity factors for crack pits in a pipe under different stresses.

According to Laham (1999), the stress intensity factor for a crack pit in a pipe under hoop stress is as follows:

$$K_{I-h} = \sqrt{\pi a} \sum_{i=0}^{3} \sigma_i f_i \left(\frac{a}{a}, \frac{2c}{a}, \frac{R}{a}\right)$$
(6.20)

and the stress intensity factor for a crack pit in a pipe under axial stress:

$$K_{I-a} = \sqrt{\pi a} \left(\sum_{i=0}^{3} \sigma_i f_i \left(\frac{a}{d}, \frac{2c}{a}, \frac{R}{d} \right) + \sigma_{bg} f_{bg} \left(\frac{a}{d}, \frac{2c}{a}, \frac{R}{d} \right) \right)$$
(6.21)

where

 K_{I-h} = Stress intensity factor for longitudinal crack in mode I, caused by hoop stress K_{I-a} = Stress intensity factor for circumferential crack in mode I, caused by axial stress a = Depth of the crack, i.e., corrosion pit

 $\sigma_i = \text{Stress normal to the crack plane}$

 f_i and f_{bg} = Geometry functions, depend on a, c (half-length of crack) and R (inner radius of pipe)

 σ_{bg} = the global bending stress, i.e. the maximum outer fibre bending stress

For internal and/or external crack pits, the difference in formulations of stress intensity factor (Equations 6.20 and 6.21) lies in geometry functions (i.e., f_i and f_{bg}), which have been presented in different tables by Laham (1999). Due to the propagation of corrosion, *a* changes with time so the stress intensity factors are time variant.

If K_C is the critical stress intensity factor, known as fracture toughness, beyond which the pipe cannot sustain propagation of the crack pit, the two limit state functions for fracture toughness can be established as follows:

Axial fracture limit state:
$$G_3(K_C, K_{I-h}, t) = K_C - K_{I-h}(t)$$
 (6.22)

Hoop fracture limit state: $G_4(K_C, K_{I-a}, t) = K_C - K_{I-a}(t)$ (6.23)

6.3.3 Calculation of the probability of failure

In the case of a pipe with multiple modes of failure, the probability of pipe failure can be determined using the methods of system reliability. Since the occurrence of either failure mode will constitute the failure of the pipe, a series system is appropriate for the assessment of pipe failures. The description of series, parallel and complex systems have already been explained in section 3.3.4.

Equation 5.7 is used for calculation of the probability of series system failure. To estimate $P_{f_i}(t)$, the probability of failure due to the *i*th failure mode, GDD model is considered in this study. The method of first passage probability is not considered for the analysis because the results of using this method for individual failure analysis in section 6.2.3 (a), showed that the dependency to auto-correlation coefficient affects the probability of failure significantly.

Therefore this method will not be used for multi failure analysis and only GDD model will be used for calculation of the probability of cast iron pipe system failure.

Using Gamma distributed degradation (GDD) Model

Considering the methodology presented in section 4.4.2, Equation 4.14 is reproduced for the all 4 limit state functions (i.e., $G_1(\sigma_y, \sigma_h, t)$, $G_2(\sigma_y, \sigma_a, t)$, $G_3(K_C, K_{I-h}, t)$ and $G_4(K_C, K_{I-a}, t)$). The results are used as $P_{f_i}(t)$ for i=1,...4 in Equation (5.7) and consequently the probability of the pipe system failure ($P_{fs}(t)$) is calculated.

The results of the probability of the pipe system failure are presented in Figures 6.15 and 6.16 for internal and external corrosion respectively.



Figure 6.15 Probability of the pipe system failure (internal corrosion)



Figure 6.16 Probability of the pipe system failure (external corrosion)

6.3.4 Verification of the results of the probability of pipe system failure

The Monte Carlo simulation method (see section 4.5 for the detail) is performed for verification of the results from GDD model in the previous section. Figures 6.17 and 6.18 show the comparison of the results of the probability of system failure from GDD model and Monte Carlo simulation for internal and external corrosion respectively.

The comparison shows that the probabilities of system failure predicted by GDD model can be verified by the results of Monte Carlo simulation method, particularly for small probabilities which are of most practical interest.



Figure 6.17 Verification of the results from GDD model by Monte Carlo simulation method (internal corrosion)



Figure 6.18 Verification of the results from GDD model by Monte Carlo simulation method (external corrosion)

6.3.5 Sensitivity analysis

It is known that the failure of a pipe can be affected by different factors, such as, pipe geometry, corrosion coefficients, soil properties and traffic loads. In view of large number of factors that affect the corrosion process and failure modes, it is of practical significance to identify those factors that affect the failure most so that more research can focus on those factors. The effect of each variable on the pipe failure can be estimated by reliability based sensitivity analysis as it was outlined in section 3.3.5.

To evaluate the sensitivity of the probability of failure to different random variables, sensitivity indexes are computed for the all 15 random variables. Figures 6.19 and 6.20 show relative contribution (α^2) and sensitivity ratios (SR) for 25-year time steps, respectively.

It is obvious from the results that the sensitivity indexes of internal pressure(P), modulus of elasticity(E_P), deflection coefficient(K_d), impact factor(I_c), surface load coefficient(C_t), wheel load(F) and pipe effective length (A) are very low for all values of time.

Among all variables, the relative contributions and sensitivity ratios of the corrosion parameters (k and n) are highly remarkable. This indicates that corrosion is very important factor for the design of underground pipelines with long lives.

Figure 6.19 also shows that the relative contribution of some other variables (e.g. wall thickness (*d*), bending moment coefficient (K_m), calculation coefficient (C_d) and width of ditch (B_d)) is large at early ages, but it gradually decreases within time. This suggests the relative unimportance of these variables particularly for old pipes.



Figure 6.19 Relative contribution of random variables in pipe failure at different times for external corrosion



Figure 6.20 Sensitivity ratio of random variables subjected to external corrosion for different elapsed times

Further sensitivity studies were carried out to investigate the effect on probability of failure of the level of variability (i.e. coefficient of variation) of each of corrosion model coefficient (i.e. k and n) as major random variables. The coefficient of variation for each of these parameters was varied from 0 to 0.5 in steps of 0.1. The coefficient of variation of all other variables was kept constant at the values given in Table 6.3. Figures 6.21 and 6.22 illustrate

the results for external corrosion for four different points of time (t). Generally the probability of failure increases while the coefficient of variation increases.



Figure 6.21 Probability of the pipe system failure due to external corrosion for different coefficients of variation of k at different times



Figure 6.22 Probability of the pipe system failure due to external corrosion for different coefficients of variation of n at different times

6.4 Summary

The two developed time dependent reliability analysis methods in chapter 4 (first passage probability method and gamma distributed degradation model), were applied on two different case studies of reliability analysis of cast iron pipes in the UK. The two following scenarios were considered:

Individual failure mode:

The concept of stress intensity in fracture mechanics has been employed to establish the limit state function for determining the probability of cast iron pipe failure. A widely used model for corrosion pit was adopted based on data mining and mathematical regressions. Between the two methods for calculation of the probability of the pipe failure, first passage probability method did not attain reliable results. The results from this method were dependent to the exact value of auto-correlation coefficient which is not normally available. The results from the other method (i.e., GDD model) showed a good agreement with the results from Monte Carlo simulation method. From the results, it has been found that the probability of the pipe collapse increases with the increase of the diameter of the pipes for both external and internal corrosion and that for a given diameter, the probability of pipe failure for pipes with external corrosion is much higher than that for pipes with internal corrosion. It has also been found that the tougher the pipe, the smaller the probability of its failure. It can be concluded that GDD model is a very useful tool to predict the probability of cast iron pipe failure and its remaining service life.

Sensitivity analysis of the two random variables (i.e., corrosion parameters, k and n) showed that at early ages, the multiplying constant (k) has higher contribution in failure function and as time passes, the contribution of exponential constant (n) become higher.

Generally the probability of failure is more sensitive to the variation of the coefficient of variation of exponential constant (n). It is also observed that the variability of the parameters (k and n) for low values of t, has more significant effect on the probability of failure. In other word, the sensitivity of corrosion parameters is more dependent on the actual value of coefficient of variation in early ages.

133

Multi failure mode:

A system reliability method for reliability analysis and service life prediction for a corrosion affected cast iron pipe in the UK was proposed. A merit of this method is that it considers multi failure modes as a system, including both loss of strength and toughness of the pipe in assessing pipe failures. For calculation of the probability of the pipeline failure, GDD model was used and the results were verified by Monte Carlo simulation method. A sensitivity analysis was undertaken to identify the most significant factors among 15 random variables that affect the pipe behaviour and failure. Among all variables, sensitivity indexes of the corrosion parameters (k and n) was highly remarkable. This indicates that corrosion is very important factor for the design of underground pipelines with longer lives. High values of contribution for these two variables means that the sensitivity of these variables is more dependent on the actual value of their coefficient of variation. In such cases, more concern should be taken in order to determine relevant parameter values.

In this chapter, the applicability of the new approach to reliability analysis of cast iron water pipes in the UK was evaluated. The uniqueness of the approach, to the inclusion of key random variables impacting upon on the reliability of cast iron water obtained pipes, was demonstrated.

A comparison of the result of the first passage probability method and GDD model was also made. The results showed a preference for GDD over the other method, as it is less time consuming and more tolerant of data availability. In the next chapter, these results are discussed in detail.

134

7 DISCUSSION AND ANALYSIS OF THE RESULTS

The main contribution of this research is the development and successful testing of the two time dependent analytical reliability analysis methods. The suitability of these models was investigated for two different UK case studies of concrete sewers and cast iron water pipes. Chapters 5 and 6, discuss the application of these models to pipelines in the UK. A modified Monte Carlo simulation was used to verify the results obtained.

In this chapter a critical analysis of the findings of this research is given. The two models are compared, and the advantages and disadvantages of both are discussed.

The results of reliability and sensitivity analysis for the UK case studies are given in sections 7.2 and 7.3. The extensive analyses undertaken provide new insight in to pipeline assessment, and this has significant beneficial implications for pipeline managers and maintenance engineers.

7.1 Comparison of the two time developed methods

The results of applying first passage probability method to concrete sewers showed a reasonably good agreement with the results from the Monte Carlo simulation method.

A new approach based on gamma process concept for reliability analysis of corrosion affected pipes also introduced in this research. The proposed method showed that a gamma process can properly model the monotonic behavior of the ageing and deterioration process of pipes. The method was called gamma distributed degradation (GDD) model and it was applied for reliability analysis of concrete sewers and cast iron water pipes.

Both developed methods (i.e., first passage probability and GDD) were applied for two scenarios for reliability analysis of concrete sewers and cast iron water pipes. The two scenarios were: individual failure assessment and multi failure mode assessment.

In this section, the two methods are compared and the strengths and weak points of each

method, based on the results of their application, are discussed.

7.1.1 First passage probability theory

To predict the service life of buried pipes, the first passage probability theory was developed. Failure due to corrosion was formulated and a stochastic model was proposed to consider uncertainties of basic random variables. The first passage probability method was employed to quantify the probability of failure, so that the time for the pipe to be unusable due to excessive corrosion can be determined.

Although the result of using first passage probability method was verified by Monte Carlo simulation method for the case of concrete sewer pipes, there are some weak points in the procedure which are discussed here.

a) Estimation of the coefficient of variation

To begin with first passage probability procedure, it is necessary to estimate the coefficient of variation of the deteriorating parameter (i.e. λ_d in Equation 5.5g and/or λ_{K_I} in Equation 6.14g). As it was mentioned in section 5.1.1 this estimation can be carried out by using a Monte Carlo simulation. It means that using first passage probability method (as an analytical method) still needs application of a numerical simulation (e.g. Monte Carlo) to estimate required parameters (i.e. λ_d and λ_{K_I}). This estimation is time consuming and will involve errors in calculations. In this case, the GDD model is less time consuming and a more reliable method for time dependent reliability analysis.

b) The effect of auto-correlation coefficient

Comparing results illustrated in Figures 5.1 and 5.12 with the results from Figures 6.4 and 6.5 shows that while in one case (concrete sewer case study) the effect of auto-correlation on probability of failure is negligible, in another case (cast iron pipe case study), significant effect for different auto-correlation coefficient can be concluded.

The effect of auto-correlation coefficient is not always negligible and the range of the results

for different auto-correlation coefficients in some cases is remarkable. For instance, in the case study for cast iron pipe (section 6.2.3a, Figure 6.4), for t=100 years, the probability of failure can change dramatically from 0.030 to 0.267 for $\rho_a = 0.1$ to 0.9. This is of practical significance because auto-correlation coefficient is not readily available and therefore any wrong assumption can lead to unreliable results.

This finding shows that using first passage probability theory for time dependent reliability analysis in some cases is very dependent to assumed amount of auto-correlation coefficient.

7.1.2 Gamma distributed degradation (GDD) model

Corrosion in the two types of the buried pipes (i.e., concrete sewers and cast iron water pipes) was modeled using the gamma process concept. The proposed method (named as gamma distributed degradation model, GDD) showed that a gamma process can model the monotonic behavior of the ageing and deterioration process of pipes. Service life of a corrosion affected pipe can be predicted with high accuracy for different level of probability of failure by using gamma distributed degradation model.

Compared with first passage probability method, gamma distributed degradation model is a straight forward method with independency to estimate coefficient of variation and auto-correlation coefficient.

For degradation processes such as corrosion where gradual damage monotonically accumulates over time, the gamma process is suitable for modeling. However, the availability of monitoring data should be considered. Initially, the gamma distributed degradation model was developed assuming that corrosion measurement data are available (section 4.4.2). in that case, estimation of gamma process parameters (i.e. shape and scale parameters (α and λ)) is only possible if at least two corrosion measurements (depth of corrosion) are available. To present more practical procedure, the proposed gamma distributed degradation model was

137

developed in section 4.4.3 to be independent from data of corrosion measurements, because in most cases, such data is not available in the water or wastewater industry. It can be concluded that the GDD method can be used as a rational tool for a comprehensive assessment of corrosion affected concrete sewers.

7.2 Reliability analysis and service life prediction of concrete sewers

The two developed methods for time dependent reliability analysis (i.e., first passage probability and GDD model) in Chapter 4, were applied for the concrete sewer pipeline in the UK. The assessment was carried out by considering two scenarios: individual failure mode and multi failure mode.

7.2.1 Results of individual failure mode analysis

In individual failure analysis scenario, concrete cover loss was assumed as the definition for failure. Analysis and assessment for repair and rehabilitation planning of the pipe can be carried out by using the results obtained from application of the developed methods for reliability analysis of concrete sewers. From the results obtained in section 5.3, the time for the pipe to be unserviceable, i.e., T_L , can be determined for a given acceptable probability of failure, P_a . For example, using the graph for GDD model in Figure 5.3, it can be obtained that $T_L = 63$ years for $P_a = 0.1$. If there is no intervention during the service period of (0, 63) years for the pipe, such as maintenance and repairs, T_L represents the time for the failure of the pipe, based on the reliability analysis. The information of T_L (i.e., time for interventions) is of practical importance to structural engineers and infrastructure managers with regard to planning for repairs and/or rehabilitation of the sewer. An optimum funding allocation for the pipeline system can be concluded by conducting a cost analysis for the repair and replacement of those corroded sewers with a higher risk of failure.

7.2.2 Results of multi failure mode analysis

Four failure modes including flexural, shear, cracking and cover loss were assumed as possible failure modes for multi failure mode analysis in section 5.4. As a comprehensive failure assessment, the time for the sewer to fail, i.e., T_L , due to concrete corrosion can be determined for a given acceptable probability of failure P_a . For example, using the graph for GDD model in Figure 5.14, it can be ascertained that $T_L = 58$ years for $P_a = 0.1$. If there is no intervention during the service period of (0, 58) years for the concerning pipe, such as maintenance and repairs, T_L represents the time for interventions or the end of service for the pipe; based on the performance criteria of the four assumed failure modes. The ascertainment of T_L (i.e., time for interventions) is of significant practical importance to structural engineers and asset managers of the concrete sewer in decision-making with regard to its repairs and/or rehabilitation, which are usually dependent on the budget situation of the day. Therefore, when to intervene is the first question of decision-makers.



Figure 7.1 Probability of system failure for different concrete cover, a_o , GDD model

To evaluate how the thickness of concrete cover may affect the service life of the pipe, a

parametric study was carried out. Figure 7.1 shows the probability of system failure over time for different thresholds (thicknesses of concrete cover, a_o) by using the GDD model. Although it is obvious that a concrete sewer with thicker concrete cover will last longer, the difference between the graphs in Figure 7.1 is a good quantitative indication to design engineers about the effectiveness of the thickness of concrete cover on service life of concrete sewers.

For instance, with an acceptable probability of failure of 10 percent ($P_a=0.1$), the service life for a pipe with 25mm concrete cover is 42 years. Increasing the concrete cover to 45mm will improve the service life to 82 years. This increment in service life can give a rational guidance for designing new pipes with longer service lives, considering more capital investment for production and using pipes with thicker concrete cover. Therefore a sewer designer can economically estimate the design and choose the optimum concrete cover with respect to higher service life in the conceptual design stage.

7.2.3 Sensitivity analysis and effectiveness of variables

A comprehensive sensitivity analysis was carried out in chapter 5 on reliability of concrete sewers. It can be identified from the results in Figure 5.4 that the relative contribution of some of the variables is considerably lower than other variables. These variables include stream velocity(u), the acid reaction factor (k) and the pH-dependent factor (j). This indicates that the inclusion of these as random variables has little influence on the probability of failure of the pipeline system. Therefore, in any future analysis, it would not be inaccurate to treat these variables as deterministic variables with constant quantities.

Among the variables, the relative contributions of the dissolved sulfide concentration ([DS]), the ratio of surface width of the stream to the perimeter of the exposed wall (b/\dot{P}) and alkalinity (A) are highly remarkable. Figures 5.5 to 5.7 presented a parametric sensitivity analysis of these three major variables. The graphs in these figures can be used for a quantitative sensitivity analysis. For instance, Figure 5.5 shows that considering 10 percent of acceptable probability of failure ($P_a = 0.1$), the service life increases from 50 to 105 years while sulphide concentration decreases from 2 mg/l to 1 mg/l. This is understandable since sulphide concentration is the main source of concrete corrosion in sewers.

Considerable effect of b/\dot{P} on service life of the concrete sewer is concluded from Figure 5.6. This can be due to the presence of more sulphide for a higher amount of relative depth. To clarify, the more sewerage in the pipe, the more available sulphide and consequently more corrosion will take place.

Sensitivity of the pipe failure to other major variable, i.e., alkalinity, (*A*), was shown in Figure 5.7. Considering acceptable probability of failure (P_a) equal to 0.1, Figure 5.7 shows, an increase of alkalinity (A) from 0.14 to 0.22 can lead to the increase of service life from approximately 60 to 95 years.

Alkalinity of concrete (*A*), which is expressed as the proportion of equivalent Calcium Carbonate (ASCE 60 2007), is the neutralizing capacity of concrete material. It needs to be noted that the results in Figure 5.7 can give a rational indication about the effect of concrete quality on the durability of concrete sewers. These results can be used for designing new concrete sewers with regard to concrete technology and the mix design of a durable concrete. For instance, the use of calcareous aggregate (limestone and dolomite) increases the alkalinity of concrete (Stutterheim and Van Aardt, (1953)) and thus prolongs the service life of concrete sewers. Results in Figure 5.7 can quantitatively show that how for each type of aggregate with specified alkalinity, the service life varies.

Further results of sensitivity analysis on major random variables illustrated in Figure 5.8 to 5.10 showed that the reliability index decreases as time and the coefficient of variation of random variables increases. It is also observed that the variability of these three random variables for low values of t has a more significant effect on the reliability index. To clarify,

the sensitivity of these variables is more dependent on the actual value of coefficient of variation for lower values of time.

The effectiveness of random variables on multi failure mode of the concrete sewer was examined in section 5.4.5 and illustrated in Figure 5.15 and 5.16. Similar to individual failure mode analysis, the significance of the three major variables (i.e, [DS], b/\dot{P} and A) on the system failure of the concrete sewer can be concluded for multi failure mode analysis.

Former studies also confirm this conclusion. For instance, research by Nielsen et al. (2009), Yongsiri et al. (2005) and De Belie et al. (2003) confirms the significance and effectiveness of the type of aggregates (i.e., alkalinity, A) and b/P ratio on the deterioration rate and service life of concrete sewers.

Among all variables, change in the velocity of the stream (*u*) has the least effect on the service life of the pipeline. For instance, for a given acceptable probability of failure of $P_a = 0.1$, service life of the pipeline only changes from 28 to 37 years. In contrast, for the variable [DS] (Dissolved sulphide concentration), the service life of the pipeline varies from 27 to 78 years, which is a much wider domain compared with the velocity of the stream (u).

The significant effect of dissolved sulphide concentration, [DS], on service life of concrete sewers is consistent with practical experience and laboratory observations (De Belie et al. (2004), Ma Guadalupe et al. (2007), Antony et al. (2010) and Wells et al. (2009)).

Consequently, in order to achieve a more accurate reliability analysis and service life prediction of sewers, infrastructure managers must note the importance of monitoring data surrounding this particular parameter

7.3 Reliability analysis and service life prediction of cast iron water pipes

In chapter 6, the two developed time dependent reliability analysis methods in this study, were also applied on two different case studies of reliability analysis of cast iron water pipes in the UK, considering an individual failure and multi failure mode.

7.3.1 Results of individual failure mode analysis

From Figure 6.6 it can be concluded that the probability of pipe failure increases as the diameter of the pipe increases. This makes sense since the internal pressure is larger in pipes with larger diameter and hence larger stresses in the pipe. Figure 6.7 shows the same trend for internal corrosion as well, i.e., that probability of pipe failure increases with the increase of diameter of the pipes for both external and internal corrosion.

For the purpose of repair and rehabilitation planning, the time for the pipe to be unserviceable (i.e., T_L in Equation 4.4) needs to be determined for a given assessment criterion and acceptable probability of failure, P_a . For example it can be obtained for GDD model and external corrosion from Figure 6.6 that $T_L = 67$ years for $P_a = 0.1$ for a cast iron pipe with diameter D= 305mm. If there is no intervention during the service period of (0, 67) years for the pipe, such as maintenance and repairs, T_L represents the service life of the pipe, based on the criteria considered in the reliability analysis.

For the purpose of illustration, when P_a is taken to be 0.1, the service lives of selected pipes from GDD model are shown in Table 7.1. From the results, it can be seen that small pipes outlast large pipes considerably. The results in Table 7.1 appear to be in agreement with practical observations that cast iron pipes are as old as 150 years and there are fewer failures for small pipes than those of large ones (O'Day et al. (1986)).

Diameter (mm)	For case 1	For case 2	For overall
254	85	590	83
234	85	590	85
305	67	402	65
406	51	340	49
		_ •	

Table 7.1 Service life (in years) for selected pipes

Figure 7.2 compares the probability of failure for pipes with external corrosion to those with internal ones in case of individual failure mode assessment. The Figure shows the results obtained from using GDD model for cast iron pipe with diameter D=305mm. It is very clear that the likelihood of collapse for pipes with external corrosion is much higher than that for pipes with internal corrosion, in particular at initial stage. There are possibly two reasons for this. One is that the external corrosion grows faster than internal one as shown in Equations 6.2 and 6.3. The other more important reason is that mechanically the corrosion defect on external surface would make it easier for the wall of the pressurised pipe to crack inwardly than that with internal corrosion defect to crack outwardly.



Figure 7.2 Probability of pipe failure for different cases, using GDD Model (D = 305mm), individual failure mode analysis

The result of a parametric sensitivity analysis on the effect of the pipe material property (i.e., toughness) on the probability of failure was illustrated in Figure 6.10. The figure shows that the tougher the pipe, the smaller the probability of its failure.

7.3.2 Results of multi failure mode analysis

More description and analysis of the results of the applied method (GDD model) are
presented in this section. The service life of the pipe (i.e., T_L in Equation 4.4) can be determined for a given assessment criterion and acceptable probability of failure P_a . For example, using the criteria of system failure, it can be obtained from Figure 6.19 that $T_L = 33$ years for $P_a = 10$ from GDD model. If there is no intervention during the service period of (0, 33) years for the pipe, such as maintenance and repairs, based on the criteria considered in system reliability analysis, the pipe will not be serviceable after year t > 33years. The information of T_L (i.e., time for intervention or service life) is of practical importance to structural engineers and asset managers with regard to planning for repairs and/or rehabilitation of the pipe networks. It can also help to achieve an optimum funding allocation for the repair and replacement of corroded pipes with higher risk of failure.

To compare the effect of internal and external corrosion on the safety of pipes, the probability of pipe failure due to external and internal corrosion from GDD model are shown in Figure 7.3. Considering Equation 6.1 as corrosion model, the difference for external and internal corrosion is in their multiplying and exponential coefficients, i.e. k and n. The results in Figure 7.3 shows that, compared with internal corrosion, external corrosion severely increases the probability of pipe failures and reduces its service life. This result is consistent with practical experience and observations (Figure 3.4, Chapter 3). As an example, while for an acceptable probability of failure of 10 percent ($P_a = 0.1$), the service life of the pipe subjected to internal corrosion is 148 years, for external corrosion, it reduces to 33 years (more than three times less).



Figure 7.3 Probability of pipe system failure with different corrosion, multi failure mode analysis

To compare the result of considering individual failure mode with the result of multi failure modes analysis, the result of using GDD model for the system limit state and for each individual limit state are illustrated in Figure 7.4.





It can be seen from this Figure that, the probability of the pipe failure is greater when multi failure modes are considered as a system than that when only each failure mode is considered individually. It is also seen that the likelihood of the pipe failure due to loss of toughness is much larger than that due to loss of strength, with the toughness caused by hoop fracture the largest. It is clear form Figure 7.4 that considering each failure mode individually in reliability analysis of pipes is not conservative which may pose undue risks to the public and subsequent disastrous consequences. The result of Figure 7.4 can also vindicate the significance of this study since it provides a more realistic and accurate method for the prediction of pipe failures.

7.3.3 Sensitivity analysis and effectiveness of variables

Sensitivity analysis on individual limit state analysis in section 6.2.6 showed that among the two random variable, the multiplying constant (k) has higher contribution in failure function at early ages and as time passes, the contribution of exponential constant (n) become higher (Figure 6.11). Further results from sensitivity analysis illustrated in Figures 6.12 and 6.13 showed that the variability of the parameters (k and n) for low values of t, has more significant effect on the probability of failure. It is to say, the sensitivity of corrosion parameters is more dependent on the actual value of coefficient of variation in early ages. In such cases, more concern should be taken in order to determine relevant parameter values.

A comprehensive sensitivity analysis on 15 random variables affecting multi failure mode of cast iron water pipes was carried out in section 6.3.6. The results presented in Figures 6.19 and 6.20 showed that the sensitivity indexes of some of the variables are very low for all values of time. These variables include internal pressure(P), modulus of elasticity(E_P), deflection coefficient(K_d), impact factor(I_c), surface load coefficient(C_t), wheel load(F) and pipe effective length (A). This indicates that the inclusion of these as variables has little influence on the probability of failure of the pipeline. Therefore, in any future analysis, it

would not be inaccurate to treat these variables as deterministic variables with constant quantities.

Among all variables, the relative contribution of the corrosion parameters (k and n) is highly remarkable. Sensitivity ratios are also considerably high for these two variables. This indicates that corrosion is very important factor for the design of underground pipelines with long lives. High values of contribution for these two variables means that the sensitivity of these variables is more dependent on the actual value of their coefficient of variation. In such cases, more concern should be taken in order to determine relevant parameter values.

From Figures 6.19 and 6.20 it is also observed that the sensitivity indexes of some other variables (e.g. wall thickness (d), bending moment coefficient (K_m), calculation coefficient (C_d) and width of ditch (B_d)) is large at early ages. However, their contribution and effectiveness decrease gradually within time and after a long elapsed time, t, the contribution of these variables to the pipe failure is very low. This suggests the relative unimportance of these variables particularly for old pipes.

Further results of sensitivity analysis on the effect of the level of variability (i.e. coefficient of variation) of each of corrosion model coefficient (i.e. k and n) on the probability of failure were shown in Figure 6.21 and 6.22. Overall, the probability of failure increases while the coefficient of variation increases. Results in these figures again confirm that the uncertainty characteristics of k and n have considerable influence on the probability of pipe failure. Therefore, as it was suggested above, extreme care should be taken in determining the values of these variables for reliability studies of cast iron pipes subject to corrosion.

The results from the application of the two new models for reliability assessment, is described in chapters 5 and 6. The following key aspects were presented in this chapter:

• Strength and weakness of first passage probability method and GDD model

- Results of reliability analysis and service life prediction of concrete sewers for individual failure assessment and for multi failure mode assessment
- Contribution and effectiveness of different variables on reliability of concrete sewers
- Results of reliability analysis and service life prediction of cast iron water pipes for individual failure assessment and for multi failure mode assessment
- Contribution and effectiveness of different variables on reliability of cast iron water pipes

In the next chapter the general conclusions and recommendations drawn from this study are presented. Suggestions for further work are given.

8 CONCLUSION AND RECOMMENDATIONS

A comprehensive discussion and analysis of the results obtained from the application of two new methods for service life prediction were given in the previous chapter. The main conclusions and recommendations for improved reliability analysis and service life prediction of buried pipes are summarised in this chapter.

It is concluded that the new insight gained through this work, will provide for enhanced reliability analysis of corrosion affected pipes, and may be applicable for other structures in general. The key outcome of this research is a more accurate and realistic basis for reliability analysis and service life prediction of pipelines.

Recommendations are given that address the need for further improvement in the field of reliability analysis of corrosion affected buried pipes.

8.1 Conclusion

This research aimed to develop and apply reliability analysis methods for the assessment and service life prediction of corrosion affected buried pipes (concrete sewers and cast iron water pipes). In this section the outcomes of the research corresponding to each of the objectives are presented briefly.

a) Understanding and investigating the design procedure, adopting models for corrosion and examining and understanding of reliability theory for pipes

• Understanding of the design procedure of buried pipes and the behaviour of pipes under various load were obtained by a comprehensive literature review in Chapter 3. The primary principle for the design of a pipeline is to ensure that both serviceability and ultimate limit states are not reached. Flexural and shear failures are two main ultimate limit states that should be considered in

design and assessment. Serviceability limit states may be measured by cracking or leakage for rigid pipes.

- In this research, structural deterioration (i.e., corrosion) in buried pipes as predominant cause of failure was studied. Variables which affect the corrosion of concrete sewers and cast iron water mains were investigated and corrosion models for these types of buried pipes were presented and adopted.
- Principles of structural reliability analysis, system reliability and sensitivity analysis were studied and a comprehensive literature review on reliability analysis of pipelines was carried out. To deal with uncertainties and scarcity of monitoring data, using time dependent methods for reliability analysis and service life prediction of corrosion affected buried pipes was intended.
- The literature review in this study also revealed that there is a lack of research on multi failure mode analysis of concrete sewers and cast iron pipes. Therefore the emphasis in the current research was on multi failure mode reliability analysis of the pipes.

b) Developing methods for reliability analysis of concrete sewers and cast iron pipes

• First passage probability theory and gamma distributed degradation (GDD) model were introduced and developed to be used for reliability analysis and service life prediction of concrete sewers and cast iron water pipes. To predict the probability of failure and service life, failure due to corrosion should be formulated in a form of limit state function and a stochastic model for corrosion should be presented to consider uncertainties of basic random variables. The first passage probability method and GDD model are employed to quantify the probability of failure, so that the time for the pipe to be

unusable due to excessive corrosion can be determined.

- A weak point for first passage probability method compare with GDD model is dependency to estimation of coefficient of variation of the deterioration process. This estimation is time consuming and will potentially involve errors in calculations.
- Since, in practice, for reliability analysis of corrosion affected pipes, data such as corrosion depth is not available; therefore, the GDD model was extended and developed for such cases.
- For a comprehensive reliability analysis, the effect of variables on the reliability of a pipeline can be analysed by doing an in-depth sensitivity analysis. Sensitivity indexes were introduced and adopted in this study as a tool to identify those variables that affect the pipe failure most; so that more research can focus on those variables.

c) Application of methods for reliability analysis of concrete sewers and cast iron pipes

- First passage probability and GDD model were applied for two scenarios of individual failure assessment and multi failure mode assessment for concrete sewers and cast iron water pipes in the UK. The results of applying first passage probability method showed that the method in some cases is very dependent to assumed amount of auto-correlation coefficient.
- Compared with first passage probability method, gamma distributed degradation model is a straight forward method with independency to estimate coefficient of variation and auto-correlation coefficient. Service life of a

corrosion affected pipe can be predicted with high accuracy for different level of probability of failure by using gamma distributed degradation model.

- The results of this study also indicate that considering each failure mode individually in reliability analysis of pipes is not conservative, which may pose undue risks to the public and subsequent disastrous consequences. In contrast, multi failure mode analysis can provide a more realistic and accurate method for the prediction of pipe failures. Accurate prediction of the service life of existing pipes has the potential to achieve risk-cost optimized strategy for the management of pipe asset.
- The results of a parametric sensitivity analysis can be used to assess how change in design parameters such as concrete cover of concrete sewers or toughness of cast iron pipes can affect the service life of the pipeline. Reliability indexes (relative contribution (α_x^2) and sensitivity ratios) were also estimated for both the concrete sewer case study and the cast iron water pipes case studies in the UK.
- For concrete sewers, the relative contributions of dissolved sulfide concentration, the ratio of surface width of the stream to the perimeter of the exposed wall and alkalinity ([DS], b/P and A), were highly remarkable. It was also observed that the variability of these three random variables for high values of time, has more significant effect on the reliability index. Thus, the sensitivity of these variables is more dependent on the actual value of the coefficient of variation for higher values of time.
- For a cast iron water pipe, the probability of the pipe failure increases with the

increase of the diameter of the pipes for both external and internal corrosion and that for a given diameter, the probability of pipe failure for pipes with external corrosion is much higher than that for pipes with internal corrosion. It has also been found that the tougher the pipe, the smaller the probability of its failure. Among all variables, the relative contribution of the corrosion parameters (k and n) was highly remarkable. High values of contribution for these two variables means that the sensitivity of these variables is more dependent on the actual value of their coefficient of variation. Therefore, more concern should be taken in order to determine the values for the corrosion parameters (k and n).

The achievements of this research and its contribution to knowledge are summarised in Figure 8.1 (the original contribution of the research has been defined in bold text).

The results of this research have been disseminated in the form of 6 journal papers and 4 conference papers (as listed in Appendix 2), illustrating the importance of this work to practitioners and regulators of corrosion affected buried infrastructure, or to corrosion affected structures as a whole.



Figure 8.1Contribution of this research in reliability analysis and service life prediction of pipelines

(The original contribution of the research is defined in bold text)

8.2 Recommendations for further research

With the trend of deregulation of water and wastewater industry in developed countries, the onus would now fall on the industry to prove the safety and reliability of existing pipe networks. The proposed methods in this research would improve the ability of the water and wastewater industry in predicting and preventing catastrophic failures of the water and sewer pipes.

Between the two developed methods in this research, the first passage probability method was challenging. In addition to the mathematical complexity of the first passage probability method, lack of data about auto correlation coefficient makes this procedure an unreliable method for practical reliability analysis problems. Instead, GDD model showed a good capability for modeling the monotonic progression of the deterioration processes in buried pipes (i.e. corrosion).

For the first passage probability method to be of practical use, having information about the auto-correlation coefficient is necessary. This can not be achieved without a considerable amount of monitoring data about the corrosion depth increment within the time. Therefore it is suggested to propose further research for monitoring corrosion depth measurements for a period of time (for instance 10 years) from concrete sewers and/or cast iron water pipes. The result of this research could give a sufficient estimate of auto-correlation coefficient. The results can also be used for calibration and verification of the other time dependent reliability analysis methods (such as gamma distributed degradation model).

For further works in the field of failure assessment of buried pipes, it is suggested to model the pipeline and existing loads in a finite element software (such as Abaqus or ANSYS). Then the model can be coupled to the stochastic reliability analysis codes produced in this research. This can give a more accurate and extensive stress analysis in 3D planes compare to the equations used in Tables 5.2 and/or 6.3.

In this research, the progression of corrosion depth in concrete sewers was assumed linear with respect to time (Equation 3.12). The corrosion rate remains unlikely to be linear as environmental condition (Wastewater properties, pH and temperature) and concrete properties continue to evolve. The prediction of service life will be much more accurate if the non-linearity of corrosion process become clear. To obtain the realistic relationship between the corrosion depth and time in a concrete sewer, an extensive lab experiment is suggested. In

156

the suggested experiment, concrete pipes are exposed to an accelerated corrosive environment. The corrosion depth in the concrete pipe sample is measured periodically and the monitoring data can be used for exploring the relationship between corrosion depth and time.

REFERENCES

Abdel-Hameed, M. (1975). A gamma wear process. IEEE Transactions on Reliability 24(2), 152–153.

ACI Committee 365, (2000) Service-Life Prediction—State-of-the-Art Report'

Ahammed, M. & Melchers, R.E. (1994), 'Reliability of underground pipelines subject to corrosion', Journal of Transportation Engineering, Vol. 120, No. 6, Nov/Dec

Ahammed, M. & Melchers, R.E, (1995) 'Probabilistic analysis of pipelines subjected to pitting corrosion leaks', Engineering Structures, Vol. 17, No. 2, 1995

Ahammed, M. and Melchers, R. E. (1997), "Probabilistic analysis of undergroud pipelines subject to combined stress and corrosion." Engineering Structures, 19(12), 988-994.

American Concrete Pipe Association, ACPA, (1993), Concrete pipe technology handbook, Irving, Texas

American Concrete Pipe Association, ACPA, (1996), Design data #40- Standard installations and bedding factors for the indirect design method, Irving, Texas

American Concrete Pipe Association, ACPA, (2007), Concrete pipe – design manual, Library of Congress catalog number 78-58624

Amirat, A., Mohamed-Chateauneuf, A., and Chaoui, K., (2006) Reliability assessment of underground pipelines under the combined effect of active corrosion and residual stress, International Journal of Pressure Vessels and Piping 83, 107-117

Ana, E., Bauwens, W., Pessemier, M., Thoeye, C., Smolders, S., Boonen, I. and De Gueldre, G., (2008), Investigating the effects of specific sewer attributes on sewer ageing-a Belgian case study, 11th International conference on urban drainage, Edinburgh, Scotland, UK

Antony, P. J., Keller, J. and Bond, P.L., (2010), Examination of Concrete Corrosion Using a Laboratory Experimental Set up Simulating Sewer Conditions, 6th International Conference on Sewer Processes and Networks, IWA, Gold Coast, Australia

ASCE Manuals and Reports of Engineering Practice No. 60, (2007), 'Gravity Sanitary Sewers', 2nd edition, American Society of Civil Engineers, New York, USA

ASCE Manuals and Reports of Engineering Practice - No. 69. (1989), Sulphide in Wastewater Collection and Treatment Systems, American Society of Civil Engineers

ASCE 15-98, (2000), "Standard practice for direct design of buried precast concrete pipe using standard installations (SIDD)", Reston, Va.

ASTM C76, (2010) Standard Specification for Reinforced Concrete Culvert, Storm Drain, and Sewer Pipe,

ASTM-E1820-01, (2001), Standard Test Method for Measurement of Fracture Toughness, ASTM International

ASTM E632-82(1996) Standard Practice for Developing Accelerated Tests to Aid Prediction of the Service Life of Building Components and Materials

Atkinson, K., Whiter, J. T., Mulheron, M. J. and Smith, P. A. (2002), "Failure of small diameter cast iron pipes" Urban Water, 4, 263-271.

BD 82/00, (2000), Design manual for roads and bridges, Volume 2, section 2, part 10, Design of buried rigid pipes, August

Benmansour, A., and Mrabet, Z., (2002), Reliability of Buried pipes, Asranet (Integrating Advanced Structural Analysis with Structural Reliability Analysis) international colloquium 8-10 July, Glasgow, Scotland

Bogdanoff, J.L. and Kozin, F., (1985), Probabilistic Models of Cumulative Damage, (John Wiley & Sons: New York).

Brian R. Kyle, (1997), "Asset service life prediction implementing reliability-based techniques", CSCE Annual Conference, MAJOR PUBLIC WORKS Key Technologies for the 21st Century, Sherbrooke, Quebec, 27-30 May

BS EN 1295-1 (1997), Structural Design of Buried Pipelines under Various Conditions of Loading, Part 1: General Requirements, BSI.

Camarinopoulos, L., Chatzoulis, A., Frontistou-Yannas, S. and Kallidromitis, V. (1999), Assessment of the time-dependent structural reliability of buried water mains, Reliability Engineering and System Safety, 65, 41-53.

Chacker, V. and Palmer, J. D., Eds. (1989), Effect of Soil Characteristic on Corrosion. ASTM special technical publication, Ann Arbor

Cheung M.S. & Kyle B.R. (1996). Service life prediction of concrete structures by reliability analysis. Journal of construction and building materials, 10(1) 45-55

Cinlar, E., Bazant, Z.P. and Osman, E., (1977), Stochastic process for extrapolating concrete creep. Journal of the Engineering Mechanics Division 103 (EM6), 1069-1088.

Clair A. M. St., and Sinha, S., (2012), State-of-the-technology review on water pipe condition, deterioration and failure rate prediction models, *Urban Water Journal* Vol. 9, No. 2, 85–112

Clifton, J. R., (1991), "Predicting Remaining Service Life of Concrete," NISTIR 4712, National Institute of Standards and Technology, Gaithersburg, Md

Clifton, J. R., and Knab, L. I., (1989), Service Life of Concrete, NISTIR 89-4086, National Institute of Standards and Technology, Gaithersburg, Md

Davis, P., De Silva, D., Gould, S. and Burn, S., (2005), Condition assessment and failure prediction for asbestos cement sewer mains, Pipes Wagga Wagga Conference, Charles Sturt University, Wagga Wagga, New South Wales, Australia.

Davis, P., De Silva, D., Marlow, D., Moglia, M., Gould, S. and Burn, S., (2008), Failure prediction and optimal scheduling of replacements in asbestos cement water pipes, Journal of Water Supply: Research and Technology—AQUA, 57.4

Davis, P. and Marlow, D., (2008)Asset Management: Quantifying Economic Lifetime of Large-Diameter Pipelines, Journal - American Water Works Association, pp 110-119

Dawotola, A.W., Trafalis, T.B., Mustaffa, Z., van Gelder, P.H.A.J.M, Vrijling, J.K., (2012), Risk based maintenance of a cros-country petroleum pipeline system, Journal of Pipeline Systems Engineering and Practice. Submitted September 2, 2011; accepted July 13, 2012

De Belie, N., Monteny, J., Beeldens, J. A., Vincke, E., Van Gemert, D. and Verstraete, W., (2004), Experimental research and prediction of the effect of chemical and biogenic sulfuric acid on different types of commercially produced concrete sewer pipes, journal of Cement and Concrete Research 34, 2223–2236

Dehghan, A, McManus, K. J. and Gad, E. F., (2008), Probabilistic failure prediction for deteriorating pipelines: nonparametric approach, Journal of Performance of Constructed Facilities, Vol. 22, No. 1

Dimitri V. Val and Leonid Chernin, (2009), Serviceability Reliability of Reinforced Concrete Beams with Corroded Reinforcement, Journal of Structural Engineering, Vol. 135, No. 8, August 1

Ditlevsen, O., and Madsen, H. O., (1996), Structural Reliability Methods, John Wiley and Sons

Doleac, M.L., Lackey, S.L., and Bratton, G.N. (1980), "Prediction of time-to failure for buried cast iron pipe", Proceedings of the Annual Conference of the American Water Works Association, Denver, Colo. pp. 31–38

Dufresne, F., Gerber, H.U. and Shiu, E.S.W., (1991), Risk theory with the gamma process. ASTIN Bulletin, 21(2), 177 – 192.

Ece Erdogmus, Brian N. Skourup; and Maher Tadros, (2010), Recommendations for Design of Reinforced Concrete Pipe, Journal of Pipeline Systems, ASCE, Engineering and Practice, Vol. 1, No. 1, February 1

Ehlen M. A., Thomas M. D. A., Bentz E. C., (2009), *LIFE-365 life time prediction model Version 2.0*, Concrete International, May

Ellingwood, B.R. & Mori, Y., (1993). Probabilistic methods for condition assessment and life prediction of concrete structures in nuclear power plants. Nuclear Engineering and Design, 142:155–166.

EPA 540-R-02-002, (2001), *Risk Assessment Guidance for Superfund: Volume III - Part A, Process for Conducting Probabilistic Risk Assessment*, Office of Emergency and Remedial Response U.S. Environmental Protection Agency Washington, DC 20460

EPA/600/R-12/017, (2012), Condition assessment technologies for water transmission and distribution systems, United States, Environmental Protection Agency

Ferguson TS, Klass MJ. (1972), A representation of independent increment processes without Gaussian components. Ann Math Stat; 43(5):1634–43

Frederiksen, Jens Mejer, Geiker, Mette Rica, (2008) 'Chloride ingress prediction: Part 1: Analytical model for time dependent diffusion coefficient and surface concentration', Concrete modelling – CONMOD'08, Proceedings of the International RILEM Symposium, , Rilem publications, France

Freudenthal, A. M., (1956), Safety and the probability of structural failure, Transactions, 121, 1337-1397 ASCE

Grigg, N. S. (2003), Infrastructure management systems, Water, wastewater, and stormwater infrastructure management, Lewis Publishers, Boca Raton, FL, 1–17.

Hau, Y., Clarke, B., Howes, C., Cunningham, R. & Mathews, M., (2005), 'Defects in sewer pipe joints and water tests', Proc. Institution of Civil Engineers, Water Management 158, Issue WM3, September, pp. 119-125.

Hertzberg, R. W., (1996), Deformation and Fracture Mechanics of Engineering Materials. Chichester, Wiley

Hoffmans, G.J.C.M. and Pilarczyk K.W., (1995). Local scour downstream of hydraulic structures. Journal of Hydraulic Engineering 121(4), 326–340.

Hvitved-Jacobsen, T., (2002) Sewer Processes – Microbial and Chemical Process Engineering of Sewer Networks. CRC Press, Boca Raton, Florida, USA

Institution of Civil Engineers, (2009), ICE State of The Nation Report Defending Critical Infrastructure, State of the Nation reports

Jensen, H. S., (2009), 'Hydrogen sulphide induced concrete corrosion of sewer networks', PhD dissertation, Aalborg University

Jiang, Y., and Sinha, K. C., (1989), Bridge service life prediction model using the Marcov chain, Transport Research record, 1223, 24-30

Kaempfer, W. and Berndt, M., (1999) Estimation of service life of concrete pipes in sewer networks, Durability of Building Materials and Components 8. Edited by M.A. Lacasse and D.J. Vanier. Institute for Research in Construction, Ottawa ON, K1A 0R6, Canada, pp. 36-45.

Kienow, K.E. & Kienow, K.K. (2001), Inspection and Evaluation of Concrete Sewers 12 Inches to 27 Inches in Diameter, Inspection and Evaluation Report, Blair, Church & Flynn Consulting Engineers, for the City of Fresno, California

Kienow, K.E. & Kienow, K.K. (2004), Risk management ... predicting your next concrete pipe sewer failure before it happens. ASCE International Conference on Pipelines, San Diego, California, USA

Kim, J., Bae, C., Woo, H., Kim, J., and Hong, S., (2007). Assessment of residual tensile strength on cast iron pipes. In: Proceedings of the Pipelines: Advances and Experiences with Trenchless Pipeline Projects, 1–7.

Kirmeyer, G.J., Richards, W., and Smith C.D. (1994), 'an assessment of water distribution systems and associated research needs', American Water Works Association Research Foundation (90658), Denver, Colorado.

Kong J. S., and Frangopol D. M., (2005), Sensitivity Analysis in Reliability-Based Lifetime Performance Prediction Using Simulation, Journal of Materials in Civil Engineering, Vol. 17, No. 3, June 1

Kucera, V. and Mattsson, E. (1987), "Atmospheric Corrosion", in Corrosion Mechanics. Mansfeld, F (ed), Marcel Dekker Inc, New York.

Lacasse M. A, Christer Sjöström, (2003), Methods for service life prediction of building materials and components – recent activities of the CIB W80 / RILEM 175-SLM

Laham, S Al (1999), "Stress Intensity Factor and Limit Load Handbook", Structural Integrity Branch British Energy Generation Ltd, EPD/GEN/REP/0316/98, ISSUE 2

Lee, L. S., Estrada, H., Baumert, M., (2010), 'Time-Dependent Reliability Analysis of FRP Rehabilitated Pipes, Journal of composites for construction, ASCE, Vol. 14, No.3, June 1

Li C.Q. & Melchers R.E. (1993), Out-crossing from convex polyhedrons for non-stationary Gaussian processes. Journal of Engineering Mechanics, ASCE, 119 (11) 2354-2361

Li, S.X., Yu, S.R., Zeng, H.L., Li, J.H. & Liang, R. (2009), 'Predicting corrosion remaining life of underground pipelines with a mechanically-based probabilistic model', Journal of Petroleum Science and Engineering, 65[3-4], 162-166.

Madsen, H.O., S. Krenk & N.C. Lind, (1986), Methods of Structural Safety. Prentice-Hall

Ma. Guadalupe D., Gutierrez-Padilla., Bielefeldt. A., Hernandez, M. and Silverstein, J., (2007), Monitoring of microbially induced concrete corrosion in pipelines, NACE, Corrosion 2007, Paper No.07514.

Mahmoodian, M. and Alani, A., (2013a), Modelling deterioration in concrete pipes as a stochastic gamma process for time dependent reliability analysis, ASCE, Journal of pipeline systems engineering and practice.

http://ascelibrary.org/doi/abs/10.1061/%28ASCE%29PS.1949-1204.0000145

Mahmoodian, M. and Alani, A., (2013b), Multi failure mode assessment of buried concrete pipes subjected to time dependent deterioration using system reliability analysis, accepted for publication, Journal of failure analysis and prevention

Mahmoodian, M. and Li, C. Q. (2011a), Structural System Reliability Analysis of Cast Iron Water Mains, 2nd Iranian Conference on Reliability Engineering, 24-26 October, Tehran, Iran.

Mahmoodian, M. and Li, C. Q. (2011b), Service life prediction of underground concrete pipes subjected to corrosion, 4th International Conference on Concrete Repair, 26-28 September, Dresden, Germany.

Makar, J. M., Desnoyers, R., McDonald, S. E., (2001), Failure modes and mechanisms in gray cast iron pipe, NRCC-44218, Underground Infrastructure Research: Municipal, Industrial and Environmental Applications, Proceedings, Kitchener, Ontario, 1-10.

Mann, E. and Frey, J. (2011) Optimized pipe renewal programs ensure cost-effective Asset management, Pipelines 2011: pp. 44-54

Marshall, P. (2001), The Residual Structural Properties of Cast Iron Pipes - Structural and Design Criteria for Linings for Water Mains, UK Water Industry Research.

Melchers, R E, (1999), Structural Reliability Analysis and Prediction, 2nd Edition, John Wiley and Sons, Chichester.

Melchers, R E, (2005a), Statistical Characterization of Pitting Corrosion, Part 1: Data Analysis, Corrosion, Vol. 61, No. 7

Melchers, R E, (2005b), Statistical Characterization of Pitting Corrosion,Part 2: Probabilistic Modeling for Maximum Pit Depth, Corrosion, Vol. 61, No. 8

Melchers, R E, (2008), Extreme value statistics and long-term marine pitting corrosion of steel, Probabilistic Engineering Mechanics 23 (2008) 482–488

Meyer, W.J., (1980), Case study of prediction of sulphide generation and corrosion in sewers', Journal Water Pollution Control Federation, 52[11], 2666-2674

Michel, A., Geiker, M. R., Stang, H., Olesen, J. F., (2010), 'Numerical modeling of reinforcement corrosion in concrete structures', 8th fib International PhD Symposium in Civil Engineering, Kgs. Lyngby, Denmark

Michel, A., Solgaard, A. O. S., Geiker, M. R., Stang, H., Olesen, J. F., (2010), 'Modelling', CONMOD'10, Symposium on concrete modelling, Lausanne, Switzerland

Misiunas, D. (2005), Failure Monitoring and Asset Condition Assessment in Water Supply Systems, PhD dissertation, Department of Industrial Electrical Engineering and Automation Lund University Moglia, M., Davis, P., and Burn, S., (2008), Strong exploration of a cast iron pipe failure model, *Journal of Reliability Engineering and System Safety* 93, 863–874

Moser, A. P. and Folkman, S., (2008), Buried piep design, Thgird edition, Mc Graw Hill.

Neethling, R.A. and Mah, M.K. Stenstrom, (1989), 'Causes and control of concrete pipe corrosion', Annual Report Submitted to the County Sanitation, Districts of Los Angeles County, Civil Engineering Department, School of Public Health, University of California, Los Angeles, CA 90024, June

NG, S. K., and F. Moses, (1966), Prediction of bridge service life using time-dependent reliability analysis, bridge management 3: inspection, maintenance, assessment and repair, E&FNSpon

Nicolai, R.P., Budai, G., Dekker, R. and Vreijling, M., (2004), Modeling the deterioration of the coating on steel structures: a comparison of methods. In proceedings of the IEEE Conference on Systems, Man and Cybernetics, IEEE, Danvers, pp. 4177-4182.

Nielsen, A.H., Jensen, H.S., Hvitved-Jacobsen, T. & Vollertsen, J., (2009), 'New findings in hydrogen sulphide related corrosion of concrete sewers', Proc. Int. Conf. on Pipeline: Infrastructure's Hidden Assets, ASCE, San Diego, USA, 15-19 August, vol. 1, pp. 344-353.

NRC (1995). Water mains break data on different pipe materials for 1992 and 1993, Technical Report A-7019.1, National Research Council Canada

Nygaard, P. V., Geiker, M. R., Elsener, B., (2009), 'Corrosion rate of steel in concrete -Evaluation of confinement techniques for on-site corrosion rate', Journal of Materials and Structures vol: 42, issue: 8, Springer Netherlands

O'Day, D.K., Weiss, R., Chiavari, S., and Blair, D. (1986), Water main evaluation for rehabilitation/replacement, American Water Works Association Research Foundation (90509), Denver, Colo.

OFWAT., (2002), 'Maintaining Water and Sewerage Systems in England and Wales, Our Proposed Approach for the 2004 Periodic Review', London

Okabe, S. M. Odagiri, T. Ito and H. Satoh, (2007), Succession of sulphur-oxidizing bacteria in the microbial community on corroding concrete in sewer systems. Applied and Environmental Microbiology 73(3), 971-980

Papoulis, A. and Pillai S.U., (2002), Probability, random variables, and stochastic processes. Fourth edition, McGraw-Hill, New York, 852p

Pelletier, G., Milhot, A. and Villeneuve, J.-P. (2003). Modeling water pipe breaks - three case studies, Journal of Water Resources Planning and Management 129(2): 115–123.

Pomeroy, R.D. (1976), The problem of hydrogen sulphide in sewers. Clay Pipe Development Association.

Rajani, B. and Kleiner, Y. (2001), Comprehensive review of structural deterioration of water mains: physically based models, National Research Council Canada, NRCC-43722

Rajani, B. and Kleiner, Y. (2004), Non-destructive inspection techniques to determine structural distress indicators in water mains, Evaluation and Control of Water Loss in Urban Water Networks, Valencia, Spain, June 21-25, 2004, pp. 1–20

Rajani, B. and Makar, J. (2000), "A methodology to estimate remaining service life of grey cast iron water mains." Canadian Journal of Civil Engineering, 27(6), 1259-1272.

Rajani, B., Makar, J., McDonald, S., Zhan, C., Kuraoka, S., Jen, CK., and Viens, M. (2000), Investigation of grey cast iron water mains to develop a methodology for estimating service life. American Water Works Association Research Foundation, Denver, Colo.

Rajani, B. and Tesfamariam, S. (2007), Estimating time to failure of Cast Iron water mains, *Water Management*, 160(WM2), 83-88

Rubinstein, R. Y. and Kroese, D. P. (2008), Simulation and the Monte Carlo Method, second edition, John Wiley and Sons.

Raju, I. S. and Newman, J. G. (1982), "Stress-intensity factors for internal and external surface cracks in cylindrical vessels." Journal of Pressure Vessel Technology, 104, 293-298.

Randall-Smith, M., Russell, A., and Oliphant, R. (1992), Guidance manual for the structural condition assessment of the trunk mains, Water Research Centre, Swindon, United Kingdom

Read, G. & Vickridge, I., (1997), 'Sewerage, Rehabilitation and New Construction, Repair and Renovation', Arnold, London.

Rubinstein, R. Y. and Kroese, D. P. (2008), Simulation and the Monte Carlo Method, second edition, John Wiley and Sons.

Sadiq, R., Rajani, B. and Kleiner, Y. (2004), "Probabilistic risk analysis of corrosion associated failures in cast iron water mains", Reliability Engineering & System Safety, 86(1), 1-10.

Salman, B. and Salem, O., (2012), Modeling Failure of Wastewater Collection Lines Using Various Section-Level Regression Models, ASCE Journal of Infrastructure Systems, Vol. 18, No. 2, June 1

Saltelli, A., Tarantola, S., Campolongo, F., and Ratto, M. (2004). Sensitivity analysis in practice: A guide to assessing scientific models, Wiley, New York.

Sharma, K. J., Yuan, Z., de Haas, D., Hamilton G., Corrie, S. and Keller, J., (2008), Dynamics and dynamic modelling of H_2S production in sewer systems, Water Research 42, 2527 - 2538

Sheikh AK, Boah JK and Hansen DA (1990) Statistical modelling of pitting corrosion and pipeline reliability. Corrosion-NACE. 46(3) 190–197

Shreir, L. L., Jarman, R. A. and Burstein, G. T. (1994), Corrosion, Butterworth Heinemann.

Singpurwalla N., (1997), Gamma processes and their generalizations: an overview. In: Cooke R, Mendel M, Vrijling H, editors. Engineering probabilistic design and maintenance for flood protection. Dordrecht: Kluwer Academic Publishers; p. 67–75.

Sivakumar Babu, G. L. and Rajaparthy S. Rao, (2005), reliability measures for buried flexible pipes, Canadian Geotechnical journal, 42: 541-549

Stutterheim and Van Aardt, (1953), 'Corrosion of concrete sewers and some possible remedies', The South African Industrial Chemist. Vol. 7, No. 10,

Timoshenko, S. (1940), "Strength of material", second edition, D. Van Nostrand Company, INC.

The Urban Waste Water Treatment Directive, 91/271/EEC, (2012), Waste water treatment in the United Kingdom – 2012 Implementation of the European Union Urban Waste Water Treatment Directive – 91/271/EEC, Department for Environment, Food and Rural Affairs, UK

Thoft-Christensen, P. and M. J. Baker. (1982), Structural Reliability Theory and Its Applications, Springer-Verlag Berlin, Heidelberg.

UKWIR (2002), The residual structural properties of cast iron pipes structural and design criteria for linings for water mains, 01/WM/02/14

USEPA (1985) Odor and corrosion control in sanitary sewerage systems and treatment plants, USEPA 625/1-85/018, Washington D. C. USA

van der Weide H., (1997), Gamma processes. In: Cooke R, Mendel M, Vrijling H, editors. Engineering probabilistic design and maintenance for flood protection. Dordrecht: Kluwer Academic Publishers; p. 77–83.

van Noortwijk, J.M. and Klatter, H.E., (1999). Optimal inspection decisions for the block mats of the Eastern-Scheldt barrier. Reliability Engineering and System Safety 65(3), 203–211.

van Noortwijk, J.M. and Pandey M.D., (2003), A stochastic deterioration process for timedependent reliability analysis, Proceedings of the Eleventh IFIP WG 7.5 Working Conference on Reliability and Optimization of Structural Systems, 2-5 November 2003, Banff, Canada, pages 259-265

van Noortwijk, J.M. and Frangopol, D.M., (2004), Two probabilistic life-cycle maintenance models for deteriorating civil infrastructures. Probabilistic Engineering Mechanics, 19(4), 345 – 359. p. 77–83.

van Noortwijk, J.M., van der Weide, J.A.M., Kallen, M.J. and Pandey, M.D., (2007), Gamma process and peaks-over-threshold distributions for time-dependent reliability, Reliability Engineering and System Safety, 92, pp 1651-1658

Vincke, E., (2002), Biogenic sulfuric acid corrosion of concrete: microbial interaction, simulation and prevention. Ph.D. Thesis, Faculty of Bio-engineering Science, University Ghent, Ghent, Belgium, pp. 7–9.

Vollertsen, J., Nielsen, A. H., Jensen, H. S., Andersen, T. W., Jacobsen, T. H., (2008), Corrosion of concrete sewers-The kinetics of hydrogen sulphide oxidation, Journal of Science of the total environment, 394, pp 162-170

Water UK, (2007), Working on behalf of the Water Industry for a sustainable future , Water UK, <u>http://www.water.org.uk/home/resources-and-links/waterfacts/resources</u>

Weimer, D. (2001). Water loss management and techniques, Technical report, DVGW, The German technical and Scientific Association for Gas and Water

Wells, T., Melchers R. E. and Bond, P. (2009), Factors involved in the long term corrosion of concrete sewers, Australasian Corrosion Association Proceedings of Corrosion and Prevention, Corrosion & Prevention - 2009, Coffs Harbour, Australia

Whittle, A. J., Tennakoon, J., (2005), Predicting the residual life of PVC sewer pipes, Plastics, Rubber and Composites, Volume 34, Number 7, pp. 311-317, September

Yamini, H., (2009), Probability of failure analysis and condition assessment of cast iron pipes due to internal and external corrosion in water distribution systems, PhD dissertation, University of British Colombia

Yongsiri, C., Vollertsen, J. and Jacobsen, T.H., (2005), Influence of wastewater constituents on hydrogen sulfide emission in sewer networks, Journal of Environmental Engineering, ASCE, Vol. 131, No. 12

Yves, L. G. and Patrick, E. (2000), Using maintenance records to forecast failures in water networks, Urban Water, 2, 173-181

Zhang, L., P. De Schryver, B. De. Gusseme, W. De Muynck, N. Boon and W. Verstraete (2008), Chemical and biological technologies for hydrogen sulphide emission control in sewer systems: A review. Water Research 42(1-2), 1-12.

Zhou W., (2011), Reliability Evaluation of Corroding Pipelines Considering Multiple Failure Modes and Time-dependent Internal Pressure, Journal of Infrastructure Systems, 17(4), 216– 224

APPENDIX 1- Codes and programming

A1.1Multi Failure Mode Reliability Analysis for Concrete Sewer

```
% system reliability analysis for 4 limit state functions (bending, shear,
% crack, cover loss)
%bP(ii) is b/P' and bPSD is its standard deviation,
        %for h/D=0.2==>>b/P'=0.36 and bPSD=0.072
        %for h/D=0.4==>>b/P'=0.55 and bPSD=0.11
        %for h/D=0.6==>>b/P'=0.71 and bPSD=0.14
        % revised 24 Feb 2012
clear
bP(1)=0.36;bP(2)=0.55;bP(3)=0.99;bPSD(1)=0.072;bPSD(2)=0.11;bPSD(3)=0.14;
%d is distance from compression face to centroid of tension
reinforcement(mm)
d=60;
% h overall thickness of member (wall thickness), (mm)
h=102;
db=12;
         % diameter of rebar in inner cage, mm
%Ms= Service load bending moment acting on length b, (Nmm/m)
%Ns= Axial thrust acting on length b, service load condition (+ when
compressive, - when tensile), (N/m)
Ms=6992511/2;
Ns=16043;
Mu=0.95*(6992511/2);% bending moment in job site
Msite=0.95*(6992511/2);% maximum bending moment in job site
Nu=0.9*16043;
Vsite=14219.5; % from file loads.xls, calculated from page 30 of ASCE 15-98
fy=270; %yeild strength of reinforcement (MPa)
tb=25;Sl=100; % tb: clear cover over reinforcement (mm), Sl: spacing of
circumferential reinforcement (mm)
B1=(25.4*tb*S1/4)^{(1/3)};
Phif=0.95;% Phif is strength reduction factor for flexure, acording to
ASCE15-98 page 8 is 0.95
C1=1; % C1 is crack control coefficient for type of reinforcement, page 12,
ASCE15-98 (2000)
b=1000;
fc=27.6;
r=395; %radius to the centerline of pipe wall, mm
As=360; % As is the area of tension reinforcemnet in unit length, (mm2)
a=(fy*As+Nu)/(0.85*b*fc);
Phiv=0.9; Fvp=1; Fcr=1;
୧୧୧୧୧
୧୫୫୫୫୫୫୫୫୫୫୫୫୫୫୫
Ro=As/(b*d);
if (Ro>0.02)
    Ro=0.02;
end
****
Fd=0.8+(41/d);
if (Fd>1.3)
    Fd=1.3;
end
ଚ୍ଚଚ୍ଚର୍ଚ୍ଚର୍ଚ୍ଚର୍ଚ୍ଚର୍ଚ୍ଚର୍ଚ୍ଚ
Fc=1+(d/(2*r));
 for ii=3:3;
N=1000;
lqcl1=0;
```

```
for t=1:200;
    lgcl1=0;
               lgcl2=0;
                          lgcl3=0; lgcl4=0;
    for i=1:N;
       for j=1:6;
           u(j)=rand;
           x(j)=norminv(u(j));
       end
       %calculation of basic random variables:
     k(i)=0.9+x(1)*0.16;
     A(i)=0.22+x(2)*0.07;
     j(i)=0.30+x(3)*0.04;
     DS(i) = 2 + x(4) * 0.5;
     v(i)=0.7+x(5)*0.12;
     bp(i)=bP(ii)+x(6)*bPSD(ii)*0;
       %calculation of Phi (hydorgen sulfide flux to pipe surface)
       %the pipe slope= 0.0015
       Phi(i)=0.7*((0.0015*v(i))^(3/8))*j(i)*DS(i)*bp(i);
       % calculation of rate of corrosion c(i):
       c(t,i)=11.5*k(i)*Phi(i)*(1/A(i));
       Lossh(t,i)=h-c(t,i)*t;
        if (Lossh(t,i)>0)
    $$$$$$$$$$$$$$$$$$$$$$$$$$$$BENDING$$$$$$$$$$$$$$$$$$$$$$$$$$
       Mu(i) = As*fy*(d-(a/2)) + Nu*((Lossh(t,i)-a)/2);
       if Mu(i)<0</pre>
           Mu(i)=0;
       end
       $$$$$$$$$$$$$$$$$$$$$$$$$$$SHEAR$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$
      Fn(t,i)=1-(Nu/(3.5*b*Lossh(t,i)*t));
Vb(t,i)=0.083*b*Phiv*d*Fvp*(fc^0.5)*(1.1+(63*As)/(b*d))*((Fd*Fn(t,i))/Fc);
      if Vb(t,i)<0</pre>
          Vb(t,i)=0;
      end
      e(i) = (Ms/Ns)+d-(Lossh(t,i)/2);
% maximum for jjj(i) is 0.9
        jjj(i)=0.74+((2.54*e(i))/d);
        if (jjj(i)>0.9)
            jjj(i)=0.9;
        end
       iii(i)=1/(1-((jjj(i)*d)/e(i)));
       AA(i)=B1/(5250*Phif*d*As);
       BB(i) = (Ms + (Ns*(d-(Lossh(t,i))/2)))/(iii(i)*jjj(i));
       CC(i)=0.083*C1*b*((Lossh(t,i))^2)*(fc^{0.5});
       F(i) = AA(i) * (BB(i) - CC(i));
       if F(i) < 0
           F(i) = 0;
       end
      % cover(i) = Lossh(t,i)-d-(db/2);
      %if cover(i)<0</pre>
       %
           cover(i)=0;
     % end
     cover(i) = c(t,i)*t;
```

```
2
      lgcl1=(Vb(t,i)<=Vsite)+lgcl1;</pre>
       lgcl2=(Mu(i)<=Msite)+lgcl2;</pre>
       lgcl3=(F(i)>=Fcr)+lgcl3;
       lgcl4=(cover(i)>=tb)+lgcl4;
     else
                          lgcl2=1+lgcl2; lgcl3=1+lgcl3; lgcl4=1+lgcl4;
          lgcl1=1+lgcl1;
      end
 end
%calculation and plotting of probability of failure in time t
 Pf1(t)=lgcl1/N;
                  Pf2(t)=lgcl2/N;
                                    Pf3(t)=lgcl3/N;
                                                        Pf4(t)=lgcl4/N;
 Pf(t)=1-(1-Pf1(t))*(1-Pf2(t))*(1-Pf3(t)*Pf4(t)); % considering crack and
cover are paralell together and totally series with bending and shear
   %Pf(t)=1-(1-Pf4(t))*(1-Pf2(t))*(1-Pf3(t))*(1-Pf1(t));
 end
hold on
end
T=1:200;
 plot(T,Pf(T))
 hold on
```

A1.2Multi Failure Mode Reliability Analysis for Cast Iron Water Pipe

```
% using Monte Carlo simulation to calculate probability of system failure
% limit state functions are strength<stress and KI>Kc means the stresses
should be more than strength and stress intensity factor should be more
than toughness to failure occur.
%when a cast iron pipe subjected to corrosion (external corrosion)
% written 29 June 2012
ffrost=0; % 0 or 1
%thickness is random therefore:
N = 1000;
for t=1:60;
    lgcl1=0;
    lgcl2=0;
    lgc13=0;
    lgcl4=0;
    for i=1:N;
% ***************(1), calculation of corrosion depth****************************
            u=rand;
            x1=norminv(u);
                Kconst(i) = 2.54 + x1 * 0.5;
            u=rand;
            x2=norminv(u);
        nconst(i) = 0.32 + x2 * 0.06;
        d(i)=Kconst(i)*t^nconst(i); % condition 0<d(i)<wthickness, after
calculation of wthickness it is checked
        if d(i) < 0
            d(i) = 0;
        end
```

```
% ****************(2-1), calculation of hoop stress*****************************
            u=rand;
            x1=norminv(u);
%% to consider max and min of p=internal pressure, max= 1.1, min =0.3
\% therefore x1 should be greater than -2and less than 2
        while (x1>2)|(x1<-2)|
            u=rand;
            x1=norminv(u);
        end
                pinternal(i)=0.7+x1*0.2;
%%to consider max and min of D=internal diametere, max= 260, min=240
% therefore x2 should be greater than -0.98 and less than 0.42
            u=rand;
            x2=norminv(u);
        while (x2>0.29) | (x2<-0.87)
            u=rand;
            x2=norminv(u);
        end
        D(i) = 305 + x2 + 17.14;
u=rand;
x3=norminv(u);
wthickness(i)=17.52+x3*0.7;
if d(i)>wthickness(i)
    d(i)=wthickness(i);
end
u=rand;
x4=norminv(u);
Km(i)=0.235+x4*0.04; %Bending moment coefficient
u=rand;
x5=norminv(u);
Ep(i)=105000+x5*15000; %modulus of elasticity of pipes
u=rand;
x6=norminv(u);
Kd(i)=0.108+x6*0.02; %Deflection coefficient
u=rand;
x7=norminv(u);
Ic(i)=1.25+x7*0.20;
                      % Impact factor
u=rand;
x8=norminv(u);
Ct(i)=0.12+x8*0.025;
                                % surface load coefficient
%% to consider max and min of F= wheel load of trafic, max= 100000,
min=30000
% therefore x9 should be greater than -1 and less than 2.5
 u=rand;
 x9=norminv(u);
 while (x9>2.5)|(x9<-1)|
  u=rand;
  x9=norminv(u);
 end
 F(i) = 50000 + x9 * 20000;
u=rand;
x10=norminv(u);
A(i) = 6500 + x10 + 200;
                          % pipe effective length
u=rand;
x11=norminv(u);
Gama(i)=(18.2/1000000)+x11*(18.2/10000000);
                                                      % unit weight of
soil
u=rand;
x12=norminv(u);
                               % calculation coefficient
Cd(i)=1.32+x12*0.25;
```

```
u=rand;
x13=norminv(u);
Bd(i)=625+x13*125;
                           % width of ditch
       newthickness(i)=wthickness(i)-d(i);
       if d(i)==wthickness(i)
       SigmaF(i)=0;
       else
       SigmaF(i)=(pinternal(i)*D(i))/(2*newthickness(i));
       end
SigmaV(i)=(3*Km(i)*Ic(i)*Ct(i)*F(i)*Ep(i)*newthickness(i)*D(i))/(A(i)*(Ep(i
)*(newthickness(i)^3)+3*Kd(i)*pinternal(i)*D(i)^3));
SigmaS(i)=(3*Km(i)*Gama(i)*Bd(i)*Bd(i)*Cd(i)*Ep(i)*newthickness(i)*D(i))/(E
p(i)*newthickness(i)^3+3*Kd(i)*pinternal(i)*D(i)^3);
       Sigmahoop(i)=SigmaF(i)+SigmaV(i)+(1+ffrost)*SigmaS(i);
 Alfap=11/1000000; %thermal coefficent of the pipe
 DeltaT=-10; %min=-10, max=0, Twater-Tground
 vp=0.21; %poisson ratio of pipe material
 SigmaT(i)=-Ep(i)*Alfap*DeltaT;
 if d(i)==wthickness(i)
    SigmaFprim(i)=0;
else
  SigmaFprim(i)=0.5*pinternal(i)*((D(i)/newthickness(i))-1)*vp;
 end
SigmaAxial(i)=SigmaT(i)+SigmaFprim(i)+(SigmaV(i)+(1+ffrost)*SigmaS(i))*vp;
% ****************(2-3), calculation of maximum Stress***************************
          % Stress(i)=max(SigmaAxial(i),Sigmahoop(i));
         % Stress(i)=(SigmaAxial(i)^2+Sigmahoop(i)^2)^0.5;
% ***************(3), calculation of stength************************
 % yeild strength of cast iron is 137 Mpa acording to ASTM A-48
Strength(i)=135;
% ************ calculation of stress intensity factor KI-hoop***********
at(i)=d(i)/wthickness(i); % is the ratio of a/t , corrosion depth over wall
thickness
u=rand;
ca(i)=4*u+1; % is the ratio of c/a , corrosion length over half
corrosion depth
Rt(i)=(D(i)/2)/wthickness(i); % is the ratio of R/t , internal radious
over wall thickness
% fi from page AI.21 Laham 1998
fi(i)=0.076*at(i)^2+0.0125*at(i)+0.6554;
KIaxial(i)=(1/31.62)*1.772*Sigmahoop(i)*fi(i)*d(i)^0.5; % 31.62 is
multiplied to change the dimension to MPa.m^0.5
% ************ calculation of stress intensity factor KI-axial************
at(i)=d(i)/wthickness(i); % is the ratio of a/t , corrosion depth over wall
thickness
u=rand;
ca(i) = 4*u+1;
                % is the ratio of c/a , corrosion length over half
corrosion depth
```

```
Rt(i)=(D(i)/2)/wthickness(i); % is the ratio of R/t , internal radious
over wall thickness
% fi and fbg from page AI.33 Laham 1998
fi(i)=0.05*at(i)+.6564;
fbg(i)=-0.2188*at(i)^3+.268*at(i)^2-0.073*at(i)+0.6589;
KIhoop(i)=(1/31.62)*1.772*d(i)^0.5*((Sigmahoop(i)*fi(i))+(SigmaAxial(i)*fbg
(i)));
        % 31.62 is multiplied to change the dimension to MPa.m^0.5
          % fracture toughness, Kq= 0.086*d^2-2.7d+29.3 MPa.m^0.5
Kq=15;
    % ************(4), Checking limit state function*********************************
lgcl1=(Sigmahoop(i)>Strength(i))+lgcl1;
lgcl2=(SigmaAxial(i)>Strength(i))+lgcl2;
lgcl3=(KIaxial(i)>Kq)+lgcl3;
lgcl4=(KIhoop(i)>Kq)+lgcl4;
    end
      Pf1(t)=lgcl1/N; % for stress-strength limit state-hoop
                       % for stress-strength limit state-Axial
      Pf2(t)=lgcl2/N;
                      % for stress intensity factor - toughness limit
     Pf3(t)=lqcl3/N;
state-hoop
                            % for stress intensity factor - toughness limit
          Pf4(t)=lqcl4/N;
state-Axial
Pf(t)=1-(1-Pf1(t))*(1-Pf2(t))*(1-Pf3(t))*(1-Pf4(t));
end
for t=1:60
 plot(t,Pf(t),'p')
  plot(t,Pf1(t),'r')
     plot(t,Pf2(t),'g')
       plot(t, Pf3(t), 'y')
       plot(t,Pf4(t),'b')
%plot(t,Proabability(t))
 hold on
end
```

APPENDIX 2-List of Publications

a) Journal papers:

- 1- Mahmoodian, M. and Alani, A., (2013), Modelling deterioration in concrete pipes as a stochastic gamma process for time dependent reliability analysis, ASCE, Journal of pipeline systems engineering and practice. <u>http://ascelibrary.org/doi/abs/10.1061/%28ASCE%29PS.1949-1204.0000145</u>
- 2- Mahmoodian, M. and Alani, A., (2013), Multi failure mode assessment of buried concrete pipes subjected to time dependent deterioration using system reliability analysis, accepted for publication, *Journal of failure analysis and prevention*
- 3- Alani, A., Faramarzi, A., Mahmoodian, M., and Tee K. F., (2013), Prediction of sulphide build-up in filled sewer pipes, Under review *Journal of Environmental Technology*
- 4- Alani, A., and Mahmoodian, M., (2013), Sensitivity analysis for reliability assessment of concrete pipes subjected to sulphide corrosion, Under review, *Journal of Urban Water*
- 5- Mahmoodian, M. and Alani, A., (2013), A gamma distributed degradation rate (GDDR) model for time dependent structural reliability analysis of concrete pipes subject to sulphide corrosion, under review, *International Journal of Reliability and Safety*
- 6- Mahmoodian, M., Alani, A. and Tee K. F., (2012), Stochastic failure analysis of the gusset plates in the Mississippi River Bridge, *International Journal of Forensic Engineering*, Vol.1, No.2, pp.153 166

b) Conference papers:

- 1- Mahmoodian, M. and Alani, A., (2013), Time dependent reliability analysis of underground concrete pipes subjected to sulphide corrosion, 11th International Conference on Structural Safety & Reliability, June 2013, USA
- 2- Mahmoodian, M. and Li, C. Q. (2011), Structural System Reliability Analysis of Cast Iron Water Mains, 2nd Iranian Conference on Reliability Engineering, 24-26 October, Tehran, Iran.
- 3- Mahmoodian, M. and Li, C. Q. (2011), Service life prediction of underground concrete pipes subjected to corrosion, 4th International Conference on Concrete Repair, 26-28 September, Dresden, Germany.
- 4- Tee, K. F., Li, C. Q. and Mahmoodian, M. (2011), Prediction of Time-variant Probability of Failure for Concrete Sewer Pipes, International conference on durability of building material and components, 12-15 April, Porto, Portugal.

APPENDIX 3- Published Works